EDWARDS AQUIFER APPLICATION ABOVEGROUND STORAGE TANK PLAN

Capitol Aggregates Rio Medina Operation Medina County, Texas

Submitted to



11551 Nacogdoches Road San Antonio, Texas 78217

Submitted by

Geosyntec <a>

consultants

engineers | scientists | innovators Engineering Firm Registration No. 1182 8217 Shoal Creek Blvd, Suite 200 Austin, Texas 78757

Submitted June 2023

Texas Commission on Environmental Quality Edwards Aquifer Application Cover Page

Our Review of Your Application

The Edwards Aquifer Program staff conducts an administrative and technical review of all applications. The turnaround time for administrative review can be up to 30 days as outlined in 30 TAC 213.4(e). Generally administrative completeness is determined during the intake meeting or within a few days of receipt. The turnaround time for technical review of an administratively complete Edwards Aquifer application is 90 days as outlined in 30 TAC 213.4(e). Please know that the review and approval time is directly impacted by the quality and completeness of the initial application that is received. In order to conduct a timely review, it is imperative that the information provided in an Edwards Aquifer application include final plans, be accurate, complete, and in compliance with <u>30 TAC 213</u>.

Administrative Review

1. <u>Edwards Aquifer applications</u> must be deemed administratively complete before a technical review can begin. To be considered administratively complete, the application must contain completed forms and attachments, provide the requested information, and meet all the site plan requirements. The submitted application and plan sheets should be final plans. Please submit one full-size set of plan sheets with the original application, and half-size sets with the additional copies.

To ensure that all applicable documents are included in the application, the program has developed tools to guide you and web pages to provide all forms, checklists, and guidance. Please visit the below website for assistance: <u>http://www.tceq.texas.gov/field/eapp</u>.

- 2. This Edwards Aquifer Application Cover Page form (certified by the applicant or agent) must be included in the application and brought to the administrative review meeting.
- 3. Administrative reviews are scheduled with program staff who will conduct the review. Applicants or their authorized agent should call the appropriate regional office, according to the county in which the project is located, to schedule a review. The average meeting time is one hour.
- 4. In the meeting, the application is examined for administrative completeness. Deficiencies will be noted by staff and emailed or faxed to the applicant and authorized agent at the end of the meeting, or shortly after. Administrative deficiencies will cause the application to be deemed incomplete and returned.

An appointment should be made to resubmit the application. The application is re-examined to ensure all deficiencies are resolved. The application will only be deemed administratively complete when all administrative deficiencies are addressed.

- 5. If an application is received by mail, courier service, or otherwise submitted without a review meeting, the administrative review will be conducted within 30 days. The applicant and agent will be contacted with the results of the administrative review. If the application is found to be administratively incomplete, it can be retrieved from the regional office or returned by regular mail. If returned by mail, the regional office may require arrangements for return shipping.
- 6. If the geologic assessment was completed before October 1, 2004 and the site contains "possibly sensitive" features, the assessment must be updated in accordance with the *Instructions to Geologists* (TCEQ-0585 Instructions).

Technical Review

- 1. When an application is deemed administratively complete, the technical review period begins. The regional office will distribute copies of the application to the identified affected city, county, and groundwater conservation district whose jurisdiction includes the subject site. These entities and the public have 30 days to provide comments on the application to the regional office. All comments received are reviewed by TCEQ.
- 2. A site assessment is usually conducted as part of the technical review, to evaluate the geologic assessment and observe existing site conditions. The site must be accessible to our staff. The site boundaries should be

clearly marked, features identified in the geologic assessment should be flagged, roadways marked and the alignment of the Sewage Collection System and manholes should be staked at the time the application is submitted. If the site is not marked the application may be returned.

- 3. We evaluate the application for technical completeness and contact the applicant and agent via Notice of Deficiency (NOD) to request additional information and identify technical deficiencies. There are two deficiency response periods available to the applicant. There are 14 days to resolve deficiencies noted in the first NOD. If a second NOD is issued, there is an additional 14 days to resolve deficiencies. If the response to the second notice is not received, is incomplete or inadequate, or provides new information that is incomplete or inadequate, the application must be withdrawn or will be denied. Please note that because the technical review is underway, whether the application is withdrawn or denied **the application fee will be forfeited**.
- 4. The program has 90 calendar days to complete the technical review of the application. If the application is technically adequate, such that it complies with the Edwards Aquifer rules, and is protective of the Edwards Aquifer during and after construction, an approval letter will be issued. Construction or other regulated activity may not begin until an approval is issued.

Mid-Review Modifications

It is important to have final site plans prior to beginning the permitting process with TCEQ to avoid delays.

Occasionally, circumstances arise where you may have significant design and/or site plan changes after your Edwards Aquifer application has been deemed administratively complete by TCEQ. This is considered a "Mid-Review Modification". Mid-Review Modifications may require redistribution of an application that includes the proposed modifications for public comment.

If you are proposing a Mid-Review Modification, two options are available:

- If the technical review has begun your application can be denied/withdrawn, your fees will be forfeited, and the plan will have to be resubmitted.
- TCEQ can continue the technical review of the application as it was submitted, and a modification application can be submitted at a later time.

If the application is denied/withdrawn, the resubmitted application will be subject to the administrative and technical review processes and will be treated as a new application. The application will be redistributed to the affected jurisdictions.

Please contact the regional office if you have questions. If your project is located in Williamson, Travis, or Hays County, contact TCEQ's Austin Regional Office at 512-339-2929. If your project is in Comal, Bexar, Medina, Uvalde, or Kinney County, contact TCEQ's San Antonio Regional Office at 210-490-3096

Please fill out all required fields below and submit with your application.

1. Regulated Entity Name: Capitol Aggregates Rio Medina Operation				2. Regulated Entity No.: RN108959032					
3. Customer Name: Capitol Aggregates Inc.			4. Customer No.: CN604033142						
5. Project Type: (Please circle/check one)	New		Modification		Extension		Exception		
6. Plan Type: (Please circle/check one)	WPAP	CZP	SCS	UST	AST	EXP	EXT	Technical Clarification	Optional Enhanced Measures
7. Land Use: (Please circle/check one)	Resider	ntial	Non-residential			8. Sit	e (acres):	503	
9. Application Fee:	\$6,500		10. Permanent Bl			BMP(s	s):	Secondary cont	ainment structure

11. SCS (Linear Ft.):	0	12. AST/UST (No. Tanks):	10 regulated ASTs
13. County:	Medina	14. Watershed:	Tributary to Elm Creek within the Frio River Basin

Application Distribution

Instructions: Use the table below to determine the number of applications required. One original and one copy of the application, plus additional copies (as needed) for each affected incorporated city, county, and groundwater conservation district are required. Linear projects or large projects, which cross into multiple jurisdictions, can require additional copies. Refer to the "Texas Groundwater Conservation Districts within the EAPP Boundaries" map found at:

http://www.tceq.texas.gov/assets/public/compliance/field_ops/eapp/EAPP%20GWCD%20map.pdf

For more detailed boundaries, please contact the conservation	on district directly.
---------------------------------------------------------------	-----------------------

Austin Region					
County:	Hays	Travis	Williamson		
Original (1 req.)					
Region (1 req.)	—	—			
County(ies)					
Groundwater Conservation District(s)	Edwards Aquifer Authority Barton Springs/ Edwards Aquifer Hays Trinity Plum Crook	Barton Springs/ Edwards Aquifer	NA		
City(ies) Jurisdiction	Austin Buda Dripping Springs Kyle Mountain City San Marcos Wimberley Woodcreek	Austin Bee Cave Pflugerville Rollingwood Round Rock Sunset Valley West Lake Hills	Austin Cedar Park Florence Georgetown Jerrell Leander Liberty Hill Pflugerville Round Rock		

San Antonio Region					
County:	Bexar	Comal	Kinney	Medina	Uvalde
Original (1 req.)				_X_	
Region (1 req.)				_X_	
County(ies)				_X_	
Groundwater Conservation District(s)	Edwards Aquifer Authority Trinity-Glen Rose	Edwards Aquifer Authority Comal Trinity	Kinney	_x_EAA _x_Medina	EAA Uvalde

City(ies) Jurisdiction	Castle Hills Fair Oaks Ranch Helotes Hill Country Village Hollywood Park San Antonio (SAWS) Shavano Park	Bulverde Fair Oaks Ranch Garden Ridge New Braunfels Schertz	NA	San Antonio ETJ (SAWS)	NA
---------------------------	----------------------------------------------------------------------------------------------------------------------------	-------------------------------------------------------------------------	----	------------------------------	----

I certify that to the best of my knowledge, that the application is complete and accurate. This application is hereby submitted to TCEQ for administrative review and technical review.

J. Brandon Klenzendorf, P.E.

6.14.2023	
Date	
	6 · 14 · 2023 Date

FOR TCEQ INTERNAL USE ONLY				
Date(s)Reviewed:		Date Adm	ninistratively Complete:	
Received From:		Correct N	Tumber of Copies:	
Received By:		Distribut	ion Date:	
EAPP File Number:		Complex:		
Admin. Review(s) (No.):		No. AR Rounds:		
Delinquent Fees (Y/N):		Review T	iew Time Spent:	
Lat./Long. Verified:		SOS Cust	omer Verification:	
Agent Authorization Complete/Notarized (Y/N):		Fee	Payable to TCEQ (Y/N):	
Core Data Form Complete (Y/N):		Check:	Signed (Y/N):	
Core Data Form Incomplete Nos.:			Less than 90 days old (Y/N):	

General Information Form

Texas Commission on Environmental Quality

For Regulated Activities on the Edwards Aquifer Recharge and Transition Zones and Relating to 30 TAC §213.4(b) & §213.5(b)(2)(A), (B) Effective June 1, 1999

To ensure that the application is administratively complete, confirm that all fields in the form are complete, verify that all requested information is provided, consistently reference the same site and contact person in all forms in the application, and ensure forms are signed by the appropriate party.

Note: Including all the information requested in the form and attachments contributes to more streamlined technical reviews.

Signature

To the best of my knowledge, the responses to this form accurately reflect all information requested concerning the proposed regulated activities and methods to protect the Edwards Aquifer. This **General Information Form** is hereby submitted for TCEQ review. The application was prepared by:

Print Name of Customer/Agent: J. Brandon Klenzendorf, P.E.

Date: 6.14.2023

Signature of Customer/Agent:

- Brandon Kl

Project Information



- 1. Regulated Entity Name: Capitol Aggregates Rio Medina Operation
- 2. County: Medina
- 3. Stream Basin: Elm Creek tributary in the Frio River Basin
- 4. Groundwater Conservation District (If applicable): <u>Edwards Aquifer Authority and Medina</u> <u>County GCD</u>
- 5. Edwards Aquifer Zone:



6. Plan Type:

WPAP
SCS

☐ Modification
▲ AST

TCEQ-0587 (Rev. 02-11-15)

1 of 4

🗌 UST

Exception Request

7. Customer (Applicant):

Contact Person: <u>Andrew Frye</u> Entity: <u>Capitol Aggregates, Inc.</u> Mailing Address: <u>11551 Nacogdoches Road</u> City, State: <u>San Antonio, Texas</u> Zip: <u>78217</u> Telephone: <u>210-871-7214</u> FAX: <u>N/A</u> Email Address: <u>Andrew.Fyre@CapitolAggregates.com</u>

8. Agent/Representative (If any):

Contact Person: <u>Brandon Klenzendorf</u>	
Entity: <u>Geosyntec Consultants</u>	
Mailing Address: <u>8217 Shoal Creek Boulevard, S</u>	<u>Suite 200</u>
City, State: <u>Austin, Texas</u>	Zip: <u>78757</u>
Telephone: <u>512-354-3281</u>	FAX: <u>N/A</u>
Email Address: <u>bklenzendorf@geosyntec.com</u>	

9. Project Location:

The project site is located inside the city limits of _____.

The project site is located outside the city limits but inside the ETJ (extra-territorial jurisdiction) of _____.

- \square The project site is not located within any city's limits or ETJ.
- 10. The location of the project site is described below. The description provides sufficient detail and clarity so that the TCEQ's Regional staff can easily locate the project and site boundaries for a field investigation.

The project site is located west of County Road 265 and southwest of Mico in Medina County, Texas (Attachment A). The southern boundary of the project site is located approximately 2.5 miles north of County Road 354. The western boundary of the project site is approximately 0.75 miles west of County Road 265. The northern boundary of the site is located approximately 1.5 miles south of County Road 368. The street address of the site is 1576 County Road 265, Hondo, Texas 78861.

- 11. Attachment A Road Map. A road map showing directions to and the location of the project site is attached. The project location and site boundaries are clearly shown on the map.
- 12. Attachment B USGS / Edwards Recharge Zone Map. A copy of the official 7 ½ minute USGS Quadrangle Map (Scale: 1" = 2000') of the Edwards Recharge Zone is attached. The map(s) clearly show:
 - Project site boundaries.
 - USGS Quadrangle Name(s).
 - Boundaries of the Recharge Zone (and Transition Zone, if applicable).
 - Drainage path from the project site to the boundary of the Recharge Zone.

13. The TCEQ must be able to inspect the project site or the application will be returned. Sufficient survey staking is provided on the project to allow TCEQ regional staff to locate the boundaries and alignment of the regulated activities and the geologic or manmade features noted in the Geologic Assessment.

Survey staking will be completed by this date: N/A

- 14. Attachment C Project Description. Attached at the end of this form is a detailed narrative description of the proposed project. The project description is consistent throughout the application and contains, at a minimum, the following details:
 - Area of the site
 - Offsite areas
 - Impervious cover
 - \boxtimes Permanent BMP(s)
 - Proposed site use
 - Site history
 - Previous development
 - Area(s) to be demolished
- 15. Existing project site conditions are noted below:
 - Existing commercial site
 - Existing industrial site
 - Existing residential site
 - \boxtimes Existing paved and/or unpaved roads
 - \boxtimes Undeveloped (Cleared)
 - Undeveloped (Undisturbed/Uncleared)
 -] Other: _____

Prohibited Activities

- 16. I am aware that the following activities are prohibited on the Recharge Zone and are not proposed for this project:
 - (1) Waste disposal wells regulated under 30 TAC Chapter 331 of this title (relating to Underground Injection Control);
 - (2) New feedlot/concentrated animal feeding operations, as defined in 30 TAC §213.3;
 - (3) Land disposal of Class I wastes, as defined in 30 TAC §335.1;
 - (4) The use of sewage holding tanks as parts of organized collection systems; and
 - (5) New municipal solid waste landfill facilities required to meet and comply with Type I standards which are defined in §330.41(b), (c), and (d) of this title (relating to Types of Municipal Solid Waste Facilities).
 - (6) New municipal and industrial wastewater discharges into or adjacent to water in the state that would create additional pollutant loading.

- 17. I am aware that the following activities are prohibited on the Transition Zone and are not proposed for this project:
 - (1) Waste disposal wells regulated under 30 TAC Chapter 331 (relating to Underground Injection Control);
 - (2) Land disposal of Class I wastes, as defined in 30 TAC §335.1; and
 - (3) New municipal solid waste landfill facilities required to meet and comply with Type I standards which are defined in §330.41 (b), (c), and (d) of this title.

Administrative Information

18. The fee for the plan(s) is based on:

- For a Water Pollution Abatement Plan or Modification, the total acreage of the site where regulated activities will occur.
- For an Organized Sewage Collection System Plan or Modification, the total linear footage of all collection system lines.
- For a UST Facility Plan or Modification or an AST Facility Plan or Modification, the total number of tanks or piping systems.
- A request for an exception to any substantive portion of the regulations related to the protection of water quality.
- A request for an extension to a previously approved plan.
- 19. Application fees are due and payable at the time the application is filed. If the correct fee is not submitted, the TCEQ is not required to consider the application until the correct fee is submitted. Both the fee and the Edwards Aquifer Fee Form have been sent to the Commission's:

TCEQ cashier

 Austin Regional Office (for projects in Hays, Travis, and Williamson Counties)
 San Antonio Regional Office (for projects in Bexar, Comal, Kinney, Medina, and Uvalde Counties)

- 20. Submit one (1) original and one (1) copy of the application, plus additional copies as needed for each affected incorporated city, groundwater conservation district, and county in which the project will be located. The TCEQ will distribute the additional copies to these jurisdictions. The copies must be submitted to the appropriate regional office.
- 21. \square No person shall commence any regulated activity until the Edwards Aquifer Protection Plan(s) for the activity has been filed with and approved by the Executive Director.

<u>Attachment A – Road Map</u>

See attached Road Map with directions to the Rio Medina Operation Site which includes the Tank Storage Area Project Site and boundaries clearly shown.

Attachment B – USGS/Edwards Recharge Zone Map

See attached USGS/Edwards Recharge Zone Map with the official 7 $\frac{1}{2}$ minute USGS Quadrangle Map (scale: 1" = 2000') of the Edwards Recharge Zone. The map clearly shows the Rio Medina Operation Site and the Tank Storage Area Project Site boundaries, USGS Quadrangle names, boundaries of the Recharge Zone and Transition Zone, and drainage path from the project site to the boundary of the Recharge Zone.



CADD\PROJECTS\R\R!D MEDINA QUARRY DEVP\WATER POLLUTION ABA (TXW0492.02)\FIGURES\TXW049202F01



Attachment C – Project Description

Capitol Aggregates, Inc. (Capitol Aggregates) is submitting this Aboveground Storage Tank (AST) Plan for the Capitol Aggregates Rio Medina Operation at 1576 County Road 265, Hondo, Texas. The overall Rio Medina Operation Site includes areas in the Edwards Aquifer Recharge Zone and has an approved Water Pollution Abatement Plan (WPAP) submitted by Geosyntec Consultants on behalf of Capitol Aggregates. The WPAP was approved on 22 June 2016 (Edwards Aquifer Protection Program ID No. 13000073). The Capitol Aggregates Rio Medina Operation Site consists of 503 acres, all within the Recharge Zone. Permanent pollution abatement measures consist of natural vegetated buffers, earthen berms, rock berms, and an extended detention basin in series with a grassy swale. The Rio Medina Operation Site also has an approved WPAP modification submitted by Geosyntec Consultants on behalf of Century Asphalt (who leases a portion of the Site). The WPAP modification was approved on 24 August 2018 (Additional ID No. 13000715). Additional permanent pollution abatement measures associated with the leased area consist of earthen berms and a wet basin. The AST Plan application will not impact the leased area associated with the WPAP modification.

The proposed Tank Storage Area Project Site is located entirely over the Edwards Aquifer Recharge Zone and consists of operation of ten permanent, regulated ASTs (identified as AST 1 through AST 10) within an impervious concrete dike secondary containment structure (with a roof system), designed to have an available storage volume sufficient to contain one and one-half times the cumulative storage capacity of the regulated tanks plus the volume displaced by unregulated tanks that will be stored within the secondary containment. The containment structure at the Tank Storage Area also includes operation of two permanent unregulated ASTs (identified as AST 11 and AST 12) with unregulated materials (i.e., Diesel Exhaust Fluid Additive and Antifreeze, respectively, which are not classified as hazardous substances under Federal Hazardous Waste Regulations 40 CFR 261) plus eighteen regulated 55-gallon portable drums containing petroleum hydrocarbon material. The cumulative storage volume of the portable drums was accounted for in the required containment volume calculation for one and one-half times the cumulative storage capacity and the displacement volume from unregulated AST 11 and AST 12 were accounted for in the available storage volume provided by the containment structure.

Due to scheduling and worker availability challenges, supply chain concerns, and material availability, Capitol Aggregates may initiate construction activities associated with the concrete pad and structure, roofing system, and electrical systems prior to approval from TCEQ on the AST application. A TCEQ Investigator recently conducted a Compliance Investigation at the Site and identified a portable 500-gallon fuel tank as an alleged violation. This AST Plan application includes this 500-gallon fuel tank and additional future tank installations. Capitol

Aggregates will not initiate additional tank installation or transfer of tank contents until approval is received.

Area of the Site

The total area of the Capitol Aggregates Rio Medina Operation Site consists of 503 acres. This AST Plan only applies to the regulated activities associated with the on-site storage tanks. The limits of the new Tank Storage Area Project Site consist of an area of less than 0.05 acres.

Off-Site Areas

Off-site areas surrounding the Tank Storage Area Project Site consist of the existing quarry operations to the west, storage containers and electrical transformer/equipment to the south and east, and undeveloped areas to the north. Off-site areas will not be required for operations associated with this project.

Impervious Cover

The original approved WPAP lists the impervious cover over the Recharge Zone as 4.9% (24.5 acres of impervious cover for the Recharge Zone area of 503 acres). The approved WPAP modification increased the impervious cover to 33.5 acres or 6.7%. Impervious cover associated with the Tank Storage Area Project Site is included as part of the existing fueling and maintenance area and will not result in any additional impervious cover.

Permanent BMPs

The Tank Storage Area Project Site is located within secondary containment within the drainage area associated with the extended detention basin in series with a grassy swale which are designed to mitigate the impervious cover for the existing fueling and maintenance area. Therefore, no additional permanent BMPs are proposed for this project.

Proposed Site Use

The proposed project associated with this AST Plan application consist of operation of ASTs within the secondary containment structure for use as a fuel island. The Rio Medina Operation Site and proposed site use will not be modified as a result of this project.

Site History and Previous Development

The Rio Medina Operation Site was approved for construction as a quarry following approval of the original WPAP on 22 June 2016. The site was primarily undeveloped prior to quarry operations in 2016. Historical land use appears to be primarily cattle ranching, with the majority of the site being undeveloped with natural vegetation aside from a residential house and

associated outbuildings that were located within the existing quarry footprint. A WPAP Modification was submitted and subsequently approved on 24 August 2018 for development of the hot mix asphalt plant area.

Areas to be Demolished

No areas are proposed to be demolished for this project.

Geologic Assessment Form (TCEQ-0585) Supplemental Information

Based on discussions with TCEQ staff, form TCEQ-0585, Geologic Assessment, is not required for this AST Plan application because the original Geologic Assessment approved with the original WPAP provides the necessary information. The original submittal for the Geologic Assessment form TCEQ-0585 (signed by Mark E. Stoker, P.G., for Geosyntec Consultants on 18 March 2016) is provided for reference. Modifications to the Geologic Assessment form beyond the original Geologic Assessment are not proposed as part of the AST Plan application.

No sensitive geologic features were identified near the Tank Storage Area Project Site associated with this AST Plan application. Acreage has not been added or removed from the Plan. Therefore, a reevaluation of the Geologic Assessment is not required.

<u>Attachment A – Geologic Assessment Table</u>

The Geologic Assessment Table from the original approved WPAP will not be modified as a result of this AST Plan application. The original Geologic Assessment from the WPAP modification is provided for reference.

<u> Attachment B – Stratigraphic Column</u>

The original Geologic Assessment approved with the WPAP provides information on the stratigraphic column. Modified information is not required for this AST Plan application.

<u>Attachment C – Site Geology</u>

The original Geologic Assessment approved with the WPAP provides information on the site geology. Modified information is not required for this AST Plan application.

Attachment D – Site Geologic Map(s)

The original Geologic Assessment approved with the WPAP provides the Site Geologic Maps. Modified information is not required for this AST Plan application.

Geologic Assessment

Texas Commission on Environmental Quality

For Regulated Activities on The Edwards Aquifer Recharge/transition Zones and Relating to 30 TAC §213.5(b)(3), Effective June 1, 1999

To ensure that the application is administratively complete, confirm that all fields in the form are complete, verify that all requested information is provided, consistently reference the same site and contact person in all forms in the application, and ensure forms are signed by the appropriate party.

Note: Including all the information requested in the form and attachments contributes to more streamlined technical reviews.

Signature

To the best of my knowledge, the responses to this form accurately reflect all information requested concerning the proposed regulated activities and methods to protect the Edwards Aquifer. My signature certifies that I am qualified as a geologist as defined by 30 TAC Chapter 213.

Print Name of Geologist: Mark Stoker, P.G.

Telephone: 210-510-4176

Date: <u>4/18/2016</u>

Fax: <u>N/A</u>

Representing: <u>Geosyntec Consultants, Inc., Texas Geoscience Firm Registration No. 50256</u> (Name of Company and TBPG or TBPE registration number)

Signature of Geologist:

Regulated Entity Name: Capitol Aggregates Rio Medina Operations

Project Information

- 1. Date(s) Geologic Assessment was performed: April 6-10, 2015; April 11, 2016
- 2. Type of Project:

🖂 WPAP SCS

3. Location of Project:



Contributing Zone within the Transition Zone

AST UST MARKE STOKER GEOLOGY 1118 COENSED 1 of 3

- 4. X Attachment A Geologic Assessment Table. Completed Geologic Assessment Table (Form TCEQ-0585-Table) is attached.
- 5. Soil cover on the project site is summarized in the table below and uses the SCS Hydrologic Soil Groups* (Urban Hydrology for Small Watersheds, Technical Release No. 55, Appendix A, Soil Conservation Service, 1986). If there is more than one soil type on the project site, show each soil type on the site Geologic Map or a separate soils map.

Table 1 - Soil Units, Infiltration	
Characteristics and Thickness	

Soil Name	Group*	Thickness(feet)
Dina Association	C	1.5 to 3.5
Pratley Clay	С	2.0 to 3.5
Speck Association	D	1.0 to 1.5
Eckrant-Rock outcrop Association - undulating	D	0.5 to 1.5
	_	

Soil Name	Group*	Thickness(feet)
Eckrant-Rock outcrop Association - hilly	D	0.5 to 1.5

* Soil Group Definitions (Abbreviated)

- A. Soils having a high infiltration rate when thoroughly wetted.
- B. Soils having a moderate infiltration rate when thoroughly wetted.
- C. Soils having a slow infiltration rate when thoroughly wetted.
- D. Soils having a very slow infiltration rate when thoroughly wetted.
- 6. Attachment B Stratigraphic Column. A stratigraphic column showing formations, members, and thicknesses is attached. The outcropping unit, if present, should be at the top of the stratigraphic column. Otherwise, the uppermost unit should be at the top of the stratigraphic column.
- 7. X Attachment C Site Geology. A narrative description of the site specific geology including any features identified in the Geologic Assessment Table, a discussion of the potential for fluid movement to the Edwards Aquifer, stratigraphy, structure(s), and karst characteristics is attached.
- 8. Attachment D Site Geologic Map(s). The Site Geologic Map must be the same scale as the applicant's Site Plan. The minimum scale is 1": 400'

Applicant's Site Plan Scale: 1'' = 400' at half size, 11''x17''Site Geologic Map Scale: 1'' = 400'Site Soils Map Scale (if more than 1 soil type): 1'' = 400'

9. Method of collecting positional data:

Global Positioning System (GPS) technology.

Other method(s). Please describe method of data collection:

- 10. The project site and boundaries are clearly shown and labeled on the Site Geologic Map.
- 11. Surface geologic units are shown and labeled on the Site Geologic Map.
- 12. Geologic or manmade features were discovered on the project site during the field investigation. They are shown and labeled on the Site Geologic Map and are described in the attached Geologic Assessment Table.

Geologic or manmade features were not discovered on the project site during the field investigation.

- 13. The Recharge Zone boundary is shown and labeled, if appropriate.
- 14. All known wells (test holes, water, oil, unplugged, capped and/or abandoned, etc.): If applicable, the information must agree with Item No. 20 of the WPAP Application Section.

There are <u>8</u> (#) wells present on the project site and the locations are shown and labeled. (Check all of the following that apply.)

 \boxtimes The wells are not in use and have been properly abandoned.

The wells are not in use and will be properly abandoned.

The wells are in use and comply with 16 TAC Chapter 76.

There are no wells or test holes of any kind known to exist on the project site.

Administrative Information

15. Submit one (1) original and one (1) copy of the application, plus additional copies as needed for each affected incorporated city, groundwater conservation district, and county in which the project will be located. The TCEQ will distribute the additional copies to these jurisdictions. The copies must be submitted to the appropriate regional office.

GEOLOGIC ASSESSMENT REPORT

Capitol Aggregates Rio Medina Operation Medina County, Texas

Submitted to



11551 Nacogdoches Road San Antonio, Texas 78217

Submitted by



consultants

engineers | scientists | innovators Texas Geoscience Firm Registration No. 50256 8217 Shoal Creek Blvd, Suite 200 Austin, Texas 78757

> Submitted January 2016 Revised March 2016, April 2016

TABLE OF CONTENTS

LIST OF FIGURES

- Figure 1 Site Location and Edwards Aquifer Recharge Zone Map
- Figure 2Site Geologic Map
- Figure 2A South Site Geologic Map
- Figure 2B North Site Geologic Map
- Figure 3 Site Soils Map
- Figure 3A South Site Soils Map
- Figure 3B North Site Soils Map

LIST OF ATTACHMENTS

Geologic Assessment Table
Stratigraphic Column
Site Geology
Site Geologic Maps
Photo Log
Soil Boring Logs
Well Registration and Well Construction Logs

LIST OF ACRONYMS AND ABBREVIATIONS

BMPs	Best Management Practices
EA	Edwards Aquifer
EAA	Edwards Aquifer Authority
Feature ID	Feature Identification Number
ft	feet
ft-msl	feet mean sea level
GA	Geologic Assessment
GA Instructions	Instructions to Geologists for Geologic Assessments on the Edwards
	Aquifer Recharge/Transition Zones (TCEQ-0585 Instructions)
Geosyntec	Geosyntec Consultants, Inc.
GPS	Global Positioning System
HSG	Hydrologic Soil Group
Kdvr	Devils River Trend
NRCS	Natural Resources Conservation Service
SB	Soil Boring
Site	Capitol Aggregates Rio Medina Operation
TCEQ	Texas Commission on Environmental Quality
USDA	United States Department of Agriculture
USGS	United States Geological Survey
WPAP	Water Pollution Abatement Plan

SECTION 1

INTRODUCTION

As requested by Capitol Aggregates, Inc., Geosyntec Consultants, Inc. (Geosyntec) has prepared this Geologic Assessment (GA) for a 503-acre tract of ranch land located west of County Road 265 in Medina County, Texas and referred to as the Capitol Aggregates Rio Medina Operation (Site). The Site is located in the Edwards Aquifer (EA) recharge zone (Figure 1) and has been proposed as the location for a limestone quarry. This GA has been prepared in support of this proposed development.

A Water Pollution Abatement Plan (WPAP) is required by the Texas Commission on Environmental Quality (TCEQ) for any regulated activity (construction/development) proposed over the EA recharge zone. This GA was performed as a component of the WPAP. The primary objective of this GA is to identify potential migration pathways for site-related constituents to the EA. In addition, this GA provides sufficient geologic information to develop and implement Best Management Practices (BMPs) during Site development.

Information presented in this GA is based on a review of relevant literature, including aerial photographs, a field survey of geologic features, and a review of logs for six soil borings (B1 to B6) conducted at the Site in July 2014 (see Figure 2). The remainder of this GA is organized into the following sections:

- Section 2 presents the results of a review of historical aerials for the Site and geologic studies relevant to the development of the Site;
- Section 3describes the field survey conducted to identify the presence of geologic and manmade features at the Site;
- Section 4 presents the soil associations found at the site and describes their physical and hydraulic characteristics;
- Section 5 describes the site geology;
- Section 6 describes the Site topography and features observed during the field survey and Geosyntec's investigation of potential karst features;
- Section 7 summarizes the key findings of the GA; and
- Section 8 lists references cited in this GA.

Figures and attachments are provided at the end of the text. The attachments are listed below:

- Attachment A Geologic Assessment Table
- Attachment B Stratigraphic Column
- Attachment C Site Geology
- Attachment D Site Geologic Maps
- Attachment E Photo Log
- Attachment F Soil Boring Logs
- Attachment G Well Registration and Well Construction Logs

Attachments A to D are required by TCEQ Form 0585.

SECTION 2

LITERATURE REVIEW

The objective of the literature review was to identify site-specific information potentially relevant to Site development and the GA. The review included historical aerials from Google Earth, United States Geological Survey (USGS) data, and other relevant geologic studies. As specified by TCEQ's 2004 Instructions for Geologists for Geologic Assessments on the Edwards Aquifer Recharge/Transition Zones [GA Instructions (TCEQ, 2004)], the review focused on literature or other sources that identify any of the following features:

- visible karst features (e.g., sink holes, caves, etc.);
- manmade recharge features (e.g., ponds, ditches, etc.);
- creek beds and drainage features;
- former land use practices and modifications; and
- topography.

Technical literature relevant to the Site and GA include the following:

- "Geologic Framework and Hydrogeologic Characteristics of the Outcrops of the Edwards and Trinity Aquifers, Medina Lake Area, Texas" (Small and Lambert, 2008);
- "Hydrogeology, Hydrologic Budget, and Water Chemistry of the Medina Lake Area, Texas" (Lambert et al., 2000);
- "Geology and Groundwater Resources of Medina County, Texas" (Holt, 1959) [not cited];
- "Geologic Map of the Edwards Aquifer Recharge Zone, South-Central Texas" (Blome et al., 2005); and
- historical aerial images from Google Earth (accessed 2015).

This review was conducted prior to the field survey so that the information could be used to help develop and focus the field survey approach.

SECTION 3

FIELD SURVEY

The objective of the field survey was to identify the presence of geologic and manmade features at the Site. The field survey was conducted using the tributary to Elm Creek and other drainage pathways identified in the review of aerial photographs as a means to hypothetically subdivide the Site (Figure 2 at a scale of 1 inch (in.) = 500 feet (ft) and Figures 2a and 2b at a scale of 1 in. = 400 ft). On the morning of April 6, 2015, Mr. Joe Tellez, a Field Geologist with Geosyntec, and Mr. Mark Stoker, P.G., Geosyntec's Geologist-of-Record for the GA, conducted a general vehicular survey of the Site. The land owners accompanied Geosyntec personnel to key Site features (e.g., water wells, former soil boring locations, creek beds, ponds, etc.) and identified various roads, trails, and landmarks. Transects were subsequently walked at right angles to the trails in 50 ft or less intervals. Drainages such as the tributary to Elm Creek were surveyed by walking along the creek bed. Geologic features were assigned a Feature Identification Number (Feature ID), listed in Attachment A in accordance with TCEQ's GA Instructions (TCEQ, 2004) and marked with flags or tape. Most of the Site has been cleared of Ashe Juniper (*Juniperus ashei*) and has been used (historically and currently) for cattle grazing.

The following is a daily summary of field activities:

April 6, 2015:

Mr. Tellez and Mr. Stoker conducted a survey along the bed of the tributary to Elm Creek. Geologic features were identified in the creek bed, photographed, and assigned the following Feature IDs: S-2-O (Limestone Cliff Ledges), S-3-O (Gray Bedrock Ledge), S-4-O (White Rock Exposure), S-5-O (Gray Bedrock), and S-7-O (Jointed Bedrock). A survey of the two hilltops at the northern third of the Site was conducted, as well as a survey of the area surrounding Soil Boring 2 (B2). The North Manmade Pond and the Middle Manmade Pond were also inspected and identified as geologic features (S-8-MB and S-9-MB, respectively) and photographed.

April 7, 2015:

Mr. Tellez surveyed drainages south of the northeastern hilltop (elevation 1180 ft mean sea level (ft-msl)) and identified the Diversion Lake Fault (S-1-F) trace. In addition, the ridge at the east side of the tributary to Elm Creek above feature S-2-O (Limestone Cliff Ledges) and the south central portion of the Site (in the area of the South Water Well [S-14-MB] and Soil Boring B1) were assessed for potential recharge features. Inspection of the upstream portion of the tributary to Elm Creek and the west bank identified two limestone boulders (Attachment E, Photo 14) along

Diversion Lake Fault (S-1-F). A change was observed in the character of the ground surface indicating an apparent trace of the fault. Exposed limestone and limestone rubble was present north of the boulders, while soil with few rock fragments and vegetation extended south of the boulders. An exposed limestone ledge (S-6-O) was identified to the north on the up-thrown side of the fault trace.

April 8, 2015:

Mr. Tellez surveyed the central, southwestern, and southeastern areas of the Site to identify potential recharge features. The Middle Topographic Depression (S-11-CD), South Water Well (S-14-MB), South Manmade Pond (S-10-MB), and South Topographic Depression (S-12-CD) were identified and inspected. In addition, the hilltop between the Middle Manmade Pond (S-9-MB) and tributary to Elm Creek was inspected for general geologic conditions and evidence of potential recharge features. The hilltop was covered with soils of the Eckrant-Rock outcrop association, undulating [United States Department of Agriculture (USDA), 1977] series. Some limestone ledges were exposed through the soil on the slope west of the hilltop.

April 9, 2015:

Features identified during the reconnaissance and marked with flags or tape were revisited by Ms. April Brandon of Geosyntec to obtain Global Positioning System (GPS) coordinates (WGS 1984) using a handheld Trimble® GeoExplorer. Data were collected with a horizontal accuracy ranging from 0.1 to 0.7 ft. A supplemental assessment was performed by Mr. Tellez at the drainages south of the northeastern hilltop (elevation 1180 ft-msl) to verify the absence of karst features along the Diversion Lake Fault trace.

April 10, 2015:

Mr. Tellez conducted a Site visit to assess the Site conditions after an overnight rainfall. Continued rainfall limited access to the entire Site, but observations of drainage and infiltration were made at the Middle Manmade Pond (S-9-MB) and Middle Topographic Depression (S-11-CD). These observations are discussed further in Section 6.1.

April 11, 2016:

Mr. Stoker and Ms. Brandon remobilized to the Site to conduct additional surveys in the limited areas previously surveyed in transects greater than 50 ft. This included sections in the northeastern and southern portions of the property. Personnel documented general geologic features and observed for evidence of potential recharge features. Surveys were performed in transects of 50 ft or less.

Exposed limestone and limestone rubble was present in the northeastern portion of the Site, in addition to steep cliff areas. The steep cliff areas were re-evaluated for karst features in areas subject to infiltration of stormwater. No additional karst features were identified.

Limestone ledges exposed through the topsoil were observed in the southern portion of the Site. Surveys were more intensely focused on areas of dense vegetation and uneven terrain. No additional karst features were identified.

SECTION 4

SOIL SURVEY

The USDA Natural Resources Conservation Service (NRCS) soil classification map for the Site is provided in Figure 3 at a scale of 1 in. = 500 ft and in Figures 3A and 3B at a scale of 1 in. = 400 ft. The Site encompasses three soil associations: the Dina, Speck, and Eckrant [referred to as the Tarrant series in USDA, (1977)]. The Pratley Clay series occurs in the far northern areas of the site and is sometimes included as part of the Dina association. General information presented herein on the soils was derived from USDA (1977, 2015). The Hydrologic Soil Group (HSG) for each soil type is included to provide an indication of the minimum rate of infiltration for the bare soil after prolonged wetting (USDA, 1986).

Dina Association

The Dina association (DNC) at the Site consists of gravelly silt loams on uplands. Slopes are 1 to 5 percent. This association is about 70 percent Dina soils with a similar soil that is 15 to 35 percent gravel occurring in the lower part of the soil profile. The remaining 30 percent consists mainly of Quihi, Speck, and Tarrant (Eckrant) soils with occasional rock outcrops. Surface runoff is medium, and the hazard of erosion is moderate. Coarse fragments on the surface help reduce erosion and evaporation. Dina association soils are classified as HSG C (USDA, 1986). Dina soils are not well suited to cultivation. Nearly all the acreage of this association is used for range and wildlife habitat.

Pratley Clay Series

The Pratley clay series (PrB) consists of moderately deep, well-drained, nearly level to gently sloping soils on uplands. These soils formed in clayey material over limestone. In a representative profile, the surface layer is very dark grayish-brown clay about 10 inches (in.) thick. The next layer is 23 in. thick, with the upper 10 in. consisting of brown clay and a few fine limestone fragments and the lower 13 in. consisting of reddish-brown clay and a few fine limestone and caliche fragments. The underlying material, to a depth of 48 in., is caliche that is indurated and laminar in the upper 1 in. and weakly cemented below that depth. Permeability is moderately slow, and the available water capacity is medium. Pratley clay soils are classified as HSG C (USDA, 1986). Pratley clay soils are used mainly for range and wildlife habitat, but some areas are farmed. The hazard of erosion is moderate.

Speck Association

The Speck association (SPD) consists of shallow soils on uplands, typically between steeper soils on hills and soils on drainage ways. It is mainly found in rounded to irregularly shaped areas. Slopes range from 1 to 8 percent, but are mainly 1 to 3 percent. This association consists of about

50 percent of a Speck clay loam, 10 percent of a Speck soil that is very gravelly, 10 percent of a similar soil that is less than 10 in. deep to limestone, 5 percent of Topia soils, 5 percent of Dina soils, 5 percent of Kavett soils, 5 percent of Tarrant soils, and 10 percent of other soils. Surface runoff is medium, and the hazard of erosion is moderate. Speck association soils are identified as HSG D based on the predominance of clay loam (USDA, 1986). These soils are used mostly for range and wildlife habitat. A few small areas are used for crops.

Eckrant-Rock Outcrop Association

The Eckrant-Rock outcrop association, undulating and hilly (TAD and TAF, respectively), consists of very shallow to shallow, well drained, undulating to steep clayey soils over indurated fractured limestone on limestone ridges. Slopes range from 1 to 30 percent. Surface runoff is rapid, and the hazard of erosion is moderate to high. Eckrant-Rock outcrop soils are classified as HSG D (USDA, 1986). These soils are used for range and wildlife habitat.

SECTION 5

SITE GEOLOGY

As required by TCEQ Form 0585, the Geologic Assessment Table (TCEQ-0585-Table) is provided in Attachment A of this GA, a Stratigraphic Column for the region is presented and described in Attachment B, a narrative description of the Site geology is presented in Attachment C, and Site Geologic Maps are referenced in Attachment D and included as Figures 2, 2A, and 2B of this GA.

SECTION 6

SITE DESCRIPTION

6.1 <u>Site Topography</u>

6.1.1 Overview

The northern area of the Site consists of large hills, hillsides, and dry creek bed drainage systems. The tributary to Elm Creek drains the Site, dividing the northern portion of the site into two hilltops to the east (elevation 1180 ft-msl) and west (elevation 1170 ft-msl) of the tributary (See Figure 2B). An east-west drainage sloping westward towards the tributary to Elm Creek separates the hill at the northeastern corner of the site from a broad hilltop (max elevation 1110 ft-msl) which makes up the relatively flat southern third of the site (Figure 2A). The southern slope of this hilltop forms the southern boundary of the Site (Figures 2A). At the time of the field survey, the tributary to Elm Creek and other drainage pathways appeared dry. A period of rainfall occurred on the night of April 9, 2015 and continued into the morning of 10 April 2015; however, because of the hazardous condition of trails at the Site, it was not possible to observe water flow at the tributary to Elm Creek and other drainage pathways.

6.1.2 Natural Topographical Features

Two natural topographic depressions were found on the Site:

- 1) Middle Topographic Depression (S-11-CD on Figure 2A) located in a wooded area of the broad hilltop (Attachment E, Photos 17 and 18); and
- 2) South Topographic Depression (S-12-CD on Figure 2A) located near the southeast corner of the Site (Attachment E, Photos 21 and 22) and initially identified during the literature review (USGS, 1964).

Neither topographic depression was identified as a recharge feature since both locations had evidence that water ponded in the depression and had slow infiltration (e.g., change in vegetation and soil accumulation via runoff). These features are both located on soil classified as Speck association, undulating (USDA, 1977) and have very slow infiltration rates (Figure 3A).

6.1.3 Manmade Topographical Features

Three manmade topographic depressions were found on the Site:

1) North Manmade Pond (S-8-MB on Figure 2B) was dry at the time of the field survey (Attachment E, Photo 7);

TXW0492/2.1_f-0585_GA Report_160418_clean.docx

- 2) Middle Manmade Pond (S-9-MB on Figure 2A) was filled with heavy dark clayey soil and had small pools of water at the time of the field survey (Attachment E, Photo 8); and
- 3) South Manmade Pond (S-10-MB on Figure 2A) was filled with heavy dark clayey soil and had small pools of water at the time of the field survey (Attachment E, Photo 20).

The ponds are located on soil types classified as very low infiltration rates (USDA, 1977). Two of the three ponds had standing water during the field survey.

6.2 <u>Site Features</u>

A brief description of a subset of Site features identified during the literature search, aerial photo review, and field survey are detailed in Attachment A and presented in Figures 2A and 2B. A brief description of each feature is provided below along with example photos.

S-1-F – Fault Trace (Figure 2B)



A 2000 study (Small and Clark, 2000) identified a fault which extends southwest to northeast across the Site. During the field survey the apparent trace of the fault was located and identified (S-1-F). The photos of the fault include limestone boulders appearing to identify the fault trace to the southwest and a limestone ledge marking the fault trace.

S-2-O, S-3-O, S-6-O – Limestone Cliff Ledges (Figure 2B)



Limestone cliff ledges are located along the dry creek bed. Numerous small overhanging ledges occur in this area and extended one to five feet into the limestone formation along the dry creek.

S-4-O, S-5-O, S-7-O – White, Gray and Jointed Bedrock (Figure 2B)





Jointed bedrock is located in the dry creek bed in the northern portion of the Site. The joints and fractures appeared to be within an individual bed with material (organic soil) plugging the expression of the fractures and joints. The fractures may propagate deeply due to prolonged exposure and weathering.

S-8-*MB* – *North Manmade Pond* (Figure 2B)



A pond (dry at the time of the field survey) was located in the northern hilly portion of the Site. Based on the presence of an earthen berm and absence of sinkhole relief and features, the pond does not appear to be naturally occurring. Soils were observed in the bottom of the pond.

S-9-MB – Middle Manmade Pond (Figure 2A)



A pond is located in the middle portion of the Site. Based on the presence of an earthen berm and absence of sinkhole relief and features, the pond does not appear to be naturally occurring. Soils were observed in the bottom of the pond.

S-10-MB – South Manmade Pond (Figure 2A)



A pond is located in the southern portion of the Site. Based on the presence of an earthen berm and absence of sinkhole relief and features, the pond does not appear to be naturally occurring. Soils were observed in the bottom of the pond.

TXW0492/2.1_f-0585_GA Report_160418_clean.docx

Geosyntec Consultants Submitted January 2016 Revised April 2016 Page 13 of 18

6.3 Investigation of Potential Karst Features

6.3.1 Surface Observations

The most prominent exposures of the Devils River Limestone were in the bed of the tributary to Elm Creek and the smaller drainages feeding into the tributary. In addition, resistive ledges of Devils River Limestone (Feature S-2-O) were partially exposed on the slopes at the Site (Attachment E, Photo 1). The reconnaissance of the creek bed provided the best view of the Devils River Limestone, although in many areas of the bedrock is covered by alluvium (Attachment E, Photo 4). No karst features were seen in the bedrock of the creek bed but there were areas of fractured bed rock (Feature S-5-O) (Attachment E, Photo 3) and jointed strata (S-7-O) (Attachment E, Photo 2) particularly in the area of the Diversion Lake Fault.

6.3.2 Subsurface Data

Geosyntec's Field Geologist examined the log for each of the six borings (B1 to B6) conducted at the Site in July 2014 for evidence of karst features. The boring logs are provided in Attachment F and boring locations are shown on Figure 3, 3A, and 3B. The following details characteristics observed within the first 30 ft.

All borings logged 0.5 ft to 20 ft of topsoil, except Boring B2 which recorded no soil cover. The log of Boring B5 (at a ground elevation of 1110.78 ft-msl) and Boring B6 (at a ground elevation of 1171.8 ft-msl, the highest at the Site) were located north of the Diversion Lake Fault and describe primarily hard, dense limestone within the first 20 ft of the surface, with chalk and clay seams. The Boring B5 log identified solution features and leached pockets within 10 ft of the surface, but these are underlain by 20 ft of dense limestone. In both Borings B5 and B6 logs the shallowest porous limestone is logged at 30 ft.

Boring B3 was the only boring with a boring log that indicated groundwater was encountered; groundwater was reported at a depth of 185 ft (corresponding to a water elevation of 839.24 ftmsl). Boring B3 was also the lowest in ground elevation at 1024.24 ft-msl, closest to the tributary to Elm Creek, and very close or on the trace of the Diversion Lake Fault. Boring B3 logged hard dense limestone with some solution features, vertical fractures (both open and healed) within 10 ft of the surface. Two vuggy zones were reported at depths of 23 to 25 ft and 28 to 29.5 ft.

Borings B1 and B2 are at nearly the same ground elevation (1100.19 ft and 1101.4 ft, respectively) and south of the Diversion Lake Fault on the down thrown side. Boring B1 logged hard dense limestone with vuggy zones, iron staining, and some solution features within the first 20 ft the vuggy limestone continued from 20 to 30 ft with the addition of red clay and calcite pockets. Boring B2 encountered similar strata, logging hard dense limestone with some solution cavities to 17 ft. Hard vuggy limestone (sometimes crystalline) was described from 17 to 30 ft where a hard dense limestone bed was encountered.

TXW0492/2.1_f-0585_GA Report_160418_clean.docx
Boring B4, the southern most boring, logged a hard dense fractured limestone with vuggy zones to 25 ft. A hard dense brecciated limestone (fractures were both open and healed) was reported from 25 to approximately 50 ft.

All borings had been properly grouted to surface (Attachment E, Photos 6 and 12). Soil boring logs are included in Attachment F.

In addition to the six borings, two active water wells were present at the Site. These are given the identifiers North Water Well (S-13-MB) and South Water Well (S-14-MB) and are shown on Figure 2A. The condition of the well heads and associated equipment and sheds of these wells were inspected and documented (Attachment E, Photos 19 and 31). The South Water Well is registered with the Edwards Aquifer Authority (EAA), and the well drilling log is provided in Attachment G. The North Water Well is currently unregistered. In general, the well heads appeared in good working condition with no evidence of direct conduits from the surface.

SECTION 7

CONCLUSIONS

Geosyntec conducted this GA by completing a literature review, aerial photo review, field survey, and review of subsurface conditions as logged in six borings conducted at the Site in 2014. Key findings from the GA are listed below:

- Surface Geology: The Site is located on the Devils River Limestone formation.
- Surface Soils: Site soils are predominately HSG C and D (USDA, 1986), and thus have a very slow to slow infiltration rate when thoroughly wetted; and
- Topographic Features:
 - $\circ\,$ Three manmade ponds located on soil types classified as having very low infiltration rates,
 - Two active water well,
 - $\circ\,$ Two natural topographical depressions, neither of which was identified as a recharge feature, and
 - The trace of a fault (Diversion Lake Fault) that crosses the Site, but had no observed karst features other than fractures;
- Fractured, jointed and vuggy limestone was observed at the surface and in soil boring logs;
- The six soil borings conducted at the Site indicated solution cavities, fractures and leached zones to depths of over 150 ft below ground surface.

Based on soils at the Site classified as having either low to moderate (HSG C and D) infiltration rates (USDA, 1986), examined structural features (e.g., the Diversion Lake Fault) not having large solution cavities or solution-enlarged fractures, the lack of any cave or natural sinkhole features, and the Devils River Limestone formation being a more dense member of the Edwards formation, none of the identified features were given a rating of sensitive.

With the development of the Site as a quarry, it is expected that recharge to the Edwards Aquifer will increase over current Site conditions. However, the quantity of recharge is anticipated to be relatively small compared to that of nearby Medina Lake.

SECTION 8

REFERENCES

- Blome, C.D., Faith, J.R., Collins, E.W., Pedraza, D.E., Murray, K.E. (2004). "Geologic Map Compilation of the Upper Seco Creek Area, Medina and Uvalde Counties, south-central Texas. USGS Open-File Report 2004-1430. Available at: http://pubs.usgs.gov/ of/2004/1430/pdf/OF2004_1430.pdf.
- Blome, C.D., Faith, J.R., Pedraza, D.E., Ozuna, G.B., Cole, J.C., Clark, A.K., Small, T.A., and Morris, R.R. (2005). "Geologic Map of the Edwards Aquifer Recharge Zone, South-Central Texas," United States Geological Survey (USGS). Available at: http://pubs.usgs.gov/sim/2005/2873/pdf/2873_Print.pdf.
- Google Earth (2015). Aerial photographs for 1995, 2005, 2008, 2009, 2011, 2012, 2013, 2014, Latitude 29.489203°, Longitude -98.992232°. Accessed 1 April.
- Holt, Jr., C.L.R. (1959). "Geology and Ground-Water Resources of Medina County, Texas," USGS Water-Supply Paper 1422. Available at: http://pubs.usgs.gov/wsp/1422/report.pdf.
- Lambert, R.B., Grimm, K.C., and Lee, R.W. (2000). "Hydrogeology, Hydrologic Budget, and Water Chemistry of the Medina Lake Area, Texas," USGS Water-Resources Investigations Report 00-4148. Available at: http://pubs.usgs.gov/wri/wri004148/pdf/ wri00-4148.pdf.
- Slattery, R.N. and Miller, L.D. (2004). "A Water-Budget Analysis of Medina and Diversion Lakes and the Medina/Diversion Lake System, with Estimated Recharge to Edwards Aquifer, San Antonio Area, Texas," USGS Scientific Investigations Report 2004-5209. Available at: http://pubs.usgs.gov/sir/2004/5209/pdf/sir2004-5209.pdf.
- Small, T.A. and Clark, A.K. (2000). "Geologic Framework and Hydrogeologic Characteristics of the Edwards Aquifer Outcrop, Medina County, Texas," USGS Water-Resources Investigations Report 00-4195. Available at: http://pubs.usgs.gov/wri/wri00-4195 /pdf/wri00-4195.pdf.
- Small, T.A. and Lambert, R.B. (2008). "Geologic Framework and Hydrogeologic Characteristics of the Outcrops of the Edwards and Trinity Aquifers, Medina Lake Area, Texas," USGS Water-Resources Investigations Report 97-4290. Available at: http://pubs.usgs.gov/ wri/wri97-4290/pdf/wri97-4290.pdf.
- TCEQ (2004). "Instructions to Geologists for Geologic Assessments on the Edwards Aquifer Recharge/Transition Zones. TCEQ-0585-Instructions. October 1.

TXW0492/2.1_f-0585_GA Report_160418_clean.docx

- TCEQ (2005). "Edwards Aquifer Protection Program, Chapter 213 Rules Recharge Zone, Transition Zone, Contributing Zone, and Contributing Zone within the Transition Zone," October 1.
- USDA (1977). "Soil Survey of Medina County, Texas," Soil Conservation Service in cooperation with Texas Agricultural Experiment Station, August.
- USDA (1986). "Urban Hydrology for Small Watersheds," Natural Resources Conservation Service. USDA, Technical Release 55, Second Edition, June.
- USDA (2015). Web Soil Survey by Soil Survey Staff, Natural Resources Conservation Service. USDA. Available at: http://websoilsurvey.nrcs.usda.gov/. Accessed 17 April.
- USGS (1964). "Timber Creek, Texas," 7.5 minute topographic map.
- USGS (1969). "Quihi, Texas," 7.5 minute topographic map.
- USGS (1982). "Medina Creek, Texas," 7.5 minute topographic map.
- USGS (1982). "Riomedina, Texas," 7.5 minute topographic map.





DDIPROJECTSIRIRIO MEDINA QUARRY DEVPIWATER POLLUTION ABA (TXW0492.02))FIGURESITXW04920103F04

		SCALE IN F	EET							
	A									
	FEATURE ID	FEATURE NA	ME							
2	S-1-F	FAULT TRACI	E							
1	S-1-F	FAULT TRACE								
	S-2-0	LIMESTONE CLIFF L	EDGES							
2	S-3-0	GRAY BEDROCK L	EDGE							
	S-4-0	WHITE ROCK EXPO	SURE							
1	S-5-0	GRAY BEDROC	ж							
1	S-6-0	ROCK LEDGE								
	S-7-0	JOINTED BEDROCK								
	S-8-MB	NORTH MANMADE POND								
	S-9-MB	MIDDLE MANMADE POND								
	S-10-MB	SOUTH MANMADE POND								
	S-11-CD	MIDDLE TOPOGRAPHIC DEPRESSION								
*	S-12-CD	SOUTH TOPOGRAPHIC DEPRESSION								
11	S-13-MB	NORTH WATER WELL								
1801	S-14-MB	SOUTH WATER WELL								
	S-15-MB	SOIL BORING B1								
	S-16-MB	SOIL BORING B2								
11	S-17-MB	SOIL BORING B3								
	S-18-MB	SOIL BORING B4								
	S-19-MB	SOIL BORING B5								
	S-20-MB	SOIL BORING B	6							
SITE IS IN THE EDWARDS AQUIFER RECHARGE ZONE										
	SITE GEOLOGIC MAP CAPITOL AGGREGATES RIO MEDINA OPERATION MEDINA COUNTY, TX									
	Geo	syntec >	TCEQ-0585							
		IRM REGISTRATION NO. 50256								
-	AUSTIN, IX	MARCH 2016	2							

#

B6 (S-20-MB)

N

500



LEGEND											
1100 EXISTING GROUND ELEVATION (FEET)											
		PROPERTY BOUNDARY									
•	1	WATER WELL									
-\$-		GEOLOGIC FEATURE									
-0-	;	SOIL BORING									
	1	MANMADE POND									
- x - x - x -	— X — 1	FENCE									
FEATURE ID		FEATURE NAME									
S-1-F		FAULT TRACE									
S-1-F		FAULT TRACE									
S-2-0	L	IMESTONE CLIFF LEDGES	6								
S-3-0		GRAY BEDROCK LEDGE									
S-4-0		WHITE ROCK EXPOSURE									
S-5-0 GRAY BEDROCK											
S-6-O ROCK LEDGE											
S-7-0 JOINTED BEDROCK											
S-8-MB		NORTH MANMADE POND									
S-9-MB											
S-10-MB											
S-11-CD	MIDDL										
S-12-CD	SOUT	NORTH WATER WELL									
S-13-MB		NORTH WATER WELL									
S-14-MB		SOUTH WATER WELL									
S-15-MB		SOIL BORING B1									
S-16-MB		SOIL BORING B2									
5-17-MB		SOIL BORING B4									
S-18-MB		SOIL BORING B4	SOIL BORING B4								
S-19-MB		SOIL BORING B5									
5-20-IVIB		SOIL BORING BO									
0 400' SCALE IN FEET											
SITE IS	IN THE RECH/	EDWARDS AQU ARGE ZONE	JIFER								
SOUTH SITE GEOLOGIC MAP CAPITOL AGGREGATES RIO MEDINA OPERATION MEDINA COUNTY, TX											
Ge	osyn	tec⊳	TCEQ-0585								
TX GEOSCIENC	CONSULI	ATTES DISTRATION NO. 50256	FIGURE								
AUSTIN, T	X	MARCH 2016	2A								



N								
LEGEND								
1100	EXISTING GROUND ELEVATION (FEET)							
	PROPERTY BOUNDARY							
•	WATER WELL							
-\$-	GEOLOGIC FEATURE							
ф (SOIL BORING							
	MANMADE POND							
- X - X - X - FENCE								
FEATURE ID	FEATURE NAME							
S-1-F	FAULT TRACE							
S-1-F	FAULT TRACE							
S-2-0	LIMESTONE CLIFF LEDGES							
S-3-0	GRAY BEDROCK LEDGE							
S-4-0	WHITE ROCK EXPOSURE							
S-5-O	GRAY BEDROCK							
S-6-0	ROCK LEDGE							
S-7-0	JOINTED BEDROCK							
S-8-MB	NORTH MANMADE POND							
S-9-MB	MIDDLE MANMADE POND							
S-10-MB	SOUTH MANMADE POND							
S-11-CD	MIDDLE TOPOGRAPHIC DEPRESSION							
S-12-CD	SOUTH TOPOGRAPHIC DEPRESSION							
S-13-MB	NORTH WATER WELL							
S-14-MB	SOUTH WATER WELL							

 S-15-MB
 SOIL BORING B1

 S-16-MB
 SOIL BORING B2

 S-17-MB
 SOIL BORING B3

 S-18-MB
 SOIL BORING B4

 S-19-MB
 SOIL BORING B5

 S-20-MB
 SOIL BORING B6

 0
 400'

SITE IS IN THE EDWARDS AQUIFER RECHARGE ZONE

SCALE IN FEET

NORTH SITE GEOLOGIC MAP CAPITOL AGGREGATES RIO MEDINA OPERATION MEDINA COUNTY,TX

Geosyn	TCEQ-0585			
CONSUL TX GEOSCIENCE FIRM REC	tants SISTRATION NO. 50256	FIGURE		
AUSTIN, TX	MARCH 2016	2B		



ADDIPROJECTSIRIRIO MEDINA QUARRY DEVPIWATER POLLUTION ABA (TXW0492.02))FIGURESITXW04920103F05A

	B6 (S-20-MB)								
\sum		MAP SYMBOL		SOIL NAME	HYDROLOGIC SOIL GROUP				
		DNC	DINA AS	SOCIATION, GENTLY UNDULATING	с				
		PrB	PRA PE	TLEY CLAY, 0 TO 3 RCENT SLOPES	С				
TAF		SPD	SPE	CK ASSOCIATION, UNDULATING	D				
		TAD	ECKRAN OUTC	IT (TARRANT) - ROCK ROP ASSOCIATION, UNDULATING	D				
		TAF	ECKRAN OUTC	IT (TARRANT) - ROCK ROP ASSOCIATION, HILLY	D				
		TE IS II		500' LE IN FEET EDWARDS A	QUIFER				
RECHARGE ZONE SITE SOILS MAP CAPITOL AGGREGATES RIO MEDINA OPERATION MEDINA COUNTY, TX									
Simil		Geo	syn	tec	TCEQ-0585				
TX GEOSCIENCE FIRM REGISTRATION NO. 50256 AUSTIN, TX MARCH 2016 3									



ICADDIPROJECTSIRIRIO MEDINA QUARRY DEVPIWATER POLLUTION ABA (TXW0492.02))FIGURESITXW04920103F05



LEGEND

	EXISTING GROUND ELEVATION (FEET)
	PROPERTY BOUNDARY
\$	WATER WELL
- \	GEOLOGIC FEATURE
+	SOIL BORING
- x - x - x - x -	FENCE

MAP SYMBOL	SOIL NAME	HYDROLOGIC SOIL GROUP
DNC	DINA ASSOCIATION, GENTLY UNDULATING	С
PrB	PRATLEY CLAY, 0 TO 3 PERCENT SLOPES	С
SPD	SPECK ASSOCIATION, UNDULATING	D
TAD	ECKRANT (TARRANT) - ROCK OUTCROP ASSOCIATION, UNDULATING	D
TAF	ECKRANT (TARRANT) - ROCK OUTCROP ASSOCIATION, HILLY	D



SITE IS IN THE EDWARDS AQUIFER RECHARGE ZONE

SOUTH SITE SOILS MAP CAPITOL AGGREGATES RIO MEDINA OPERATION MEDINA COUNTY, TX

Geosyn	tec⊳	TCEQ-0585			
CONSUL TX GEOSCIENCE FIRM REG	FIGURE				
AUSTIN, TX	MARCH 2016	3A			





MAP SYMBOL	SOIL NAME	HYDROLOGIC SOIL GROUP
DNC	DINA ASSOCIATION, GENTLY UNDULATING	С
PrB	PRATLEY CLAY, 0 TO 3 PERCENT SLOPES	С
SPD	SPECK ASSOCIATION, UNDULATING	D
TAD	ECKRANT (TARRANT) - ROCK OUTCROP ASSOCIATION, UNDULATING	D
TAF	ECKRANT (TARRANT) - ROCK OUTCROP ASSOCIATION, HILLY	D



SITE IS IN THE EDWARDS AQUIFER RECHARGE ZONE

NORTH SITE SOILS MAP CAPITOL AGGREGATES RIO MEDINA OPERATION MEDINA,TX

SILIS	Geosyn	TCEQ-0585			
9	CONSULT TX GEOSCIENCE FIRM REC	FIGURE			
	AUSTIN, TX	MARCH 2016	3B		

Capitol Aggregates Rio Medina Operation Medina County, Texas TCEQ-0585, Geologic Assessment

ATTACHMENTS

TXW0492/2.1_f-0585_GA Report_160418_clean.docx

Geosyntec Consultants Submitted January 2016 Revised March 2016

ATTACHMENT A Geologic Assessment Table

GEOLOGI	C ASSESSMEN	T TABLE			<u></u>		PR	OJE	CT NA	M	E:	Capito	Aggr	egates Rio	Medina C	perat	ion, M	Medina	Count	y, Texas
	LOCATION					FFATURE CHARACTERISTICS								EVAL	UATI	ÓN	PH	YSICA	L SETTING	
lA	в.	IC*	2A	2B	3		4		5	5A	6	7	8A	8B	9	1	ю		11	12
FEATUREID	LAITIUDE	LONGITUDE	FEATURE TYPE	POINTS	FORMATION	DIME	NSIONS	(FEET)	IREND (DEGREES)	DOM	DENS ITY (NOTFT)	APER IURE (FFE I)	DJUL	RELATIVE INFILTRATION RATE	TOTAL	S ENS	плих	CATCHM (AC	ENTAREA RES)	IOPOGRAPHY
						х	Y	Z		ю						-40	<u>.:40</u>	<l6< td=""><td><u>-16</u></td><td></td></l6<>	<u>-16</u>	
S-1-F	29,49104	-98.99411	F	20	Kdvr				N68E				0	5	25	X			X	Hillside
S-1-F	29.49101	-98.99419	F	20	Kdvr				N68E				0	5	25	X			X	Hillside
S-2-0	29.48989	-98.99404	0	5	Kdvr								0	5	10	x		x		Drainage
8-3-0	29.49081	-98,99352	0	5	Kdvr						1		0	5	10	x		X		Drainage
S-4-0	29.49060	-98.99344	0	5	Kdvr								0	5	10	X		X		Streambed
S-5-0	29.49043	-98.99338	0	5	Kdvr				NW-SE		1	0.1	0	5	10	X		X		Streambed
S-6-O	29.49097	-98.99431	0	5	Kdvr					Γ	[0	5	10	x		X		Drainage
\$-7-0	29.48952	-98.99505	0	5	Kdvr				NE-SW				0	5	10	X		X		Streambed
S-8-MB	29,49601	-98.99425	MB	30	Kdvr	197	132						0	8	38	X			X	NA
\$-9-MB	29.48642	-98.99211	MB	30	Kdvr	45	53						0	8	38	X			X	NA
S-10-MB	29.48454	-98.98943	MB	30	Kdvr	64	73						0	8	38	X			x	NA
\$-11-CD	29,48601	-98,99094	CD	5	Kdvr								0	5	10	X			X	NA
\$-12-CD	29.48164	-98,98684	CD	5	Kdvr								0	5	10	X			X	NA
\$-13-MB	29.48667	-98.98947	MB	30	Kdvr								0	5	35	X		X		NA
5-14-MB	29.48291	-98.99229	MB	30	Kdvr								0	5	35	X		X		NA
S-15-MB (B-1)	29.48360	-98,99006	MB	5	Kdvr						1		X	5	10	x		X		NA
S-16-MB (B-2)	29.48745	-98.98946	MB	5	Kdvr						1		X	5	10	X		x		NA
S-17-MB (B-3)	29.49188	-98.99337	MB	5	Kdvr								x	5	10	X		X		NA
S-18-MB (B-4)	29.48138	-98,99418	MB	5	Kdvr								x	5	10	X		Х		NA
S-19-MB (B-5)	29.49480	-98.98907	MB	5	Kdvr								X	5	10	X		x		NA
S-20-MB (B-6)	29.49859	-98.99849	MB	5	Kdvr								X	5	10	X		x		NA
* DATUM:	WG8 1984	Annora agus an ta					9-19-19-19-19-19-19-19-19-19-19-19-19-19													
2A TYPE		ТҮРЕ		28	POINTS						8	A INFIL	LING							
с	Cave				30		N	None,	exposed	bedr	ock									
SC	Solution cavity				20		с	Coarse	e - cobble	es, br	eakdow	ı, sand, g	ravel						68.98	
SF	Solution-enlarged fract	ure(s)		20 O Loose or soft mud or soil, organics, leaves, sticks, dark colors																
F	Fault			20 F Fines, compacted clay-rich sediment, soil profile, gray or red colors																
0	Other natural bedrock	features		5 V Vegetation. Give details in narrative description																
мв	Manmade feature in be	drock			30	30 FS Flowstone, cements, cave deposits														
SW	Swallow hole				30	30 X Other materials														
SH	Sinkhole				20	20														
CD Z	Non-karst closed depre	ession			5		CHE	1 1106 -		12 T	OPOG	SAPHY Floodel	in St	ambad						1 Formanno
<u>L</u>	Zone, clustered or align	ned features			30		Cim	, millo	p, rillsio	ie, D	ramage,	гюодріа	ini, sire	amoed	1					1 LMAF

Attachment A - Geologic Assessment Table

I have read, I understood, and I have followed the Texas Commission on Environmental Quality's Instructions to Geologists. The information presented here complies with that document and is a true representation of the conditions observed in the field. My signature certifies that I am qualified as a geologist as defined by 30 TAC Chapter 213.

Marsa Stortes

8/26/6 Date Sheet 1

Geosyntec Consultants Submitted January 2016 Revised March 2016

GEOLOGY

1118

. ----

1. Ale and the second s

TCEQ-0585-Table (Rev. 10-01-04) TXW0492/2,1_f-0585_GA Report_160117_clean.docx

Capitol Aggregates Rio Medina Operation Medina County, Texas TCEQ-0585, Geologic Assessment

ATTACHMENT B Stratigraphic Column

Stratigraphic Column

The Site is located over the Devils River Limestone formation (Kdvr). Outcrops of this formation are extensive across the hilly northern portion of the Site.

Formation	Thickness
*Kdvr - Devils River Limestone	300 to 390 ft

*Defined as part of the Edwards Aquifer in Title 30 of the Texas Administrative Code (TAC): 30 TAC § 213.3(8).

Stratigraphy of the Devils River Limestone

The Devils River Limestone is the only stratigraphic unit exposed at the site. In the northern area of the Site, alluvium soil covers areas along the streambed and in valleys, while the southern more flat area is predominantly covered by alluvium soil. The Devils River Limestone is equivalent to the Edwards Group exposed approximately one mile north of the Site.

A general description of the Devils River Limestone presented below was derived from Blome et al. (2005). Devils River Limestone is Lower Cretaceous age. The upper 250 ft consists of miliolids, shell-fragment wackestones and grainstones containing rudists and chert. The middle of the formation consists of recrystallized and brecciated mudstones that grade downward into alternating beds of vuggy spar and chert-bearing wackestone and grainstone. The lower 120 to 250 ft contains sparry limestone and nodular, burrowed mudstone to wackestone, with gastropods, miliolids, and *Exogyra texana*.

Geologic maps typically show a dashed boundary separating the San Marcos platform and the Devils River trend exposures because field studies of the San Marcos platform/Devils River trend transition area reveal that the facies boundary is distinct and follows topographic expression in some areas.

Formal subdivision of the Devils River Formation has been lacking to date. Geologic maps of the Texas Bureau of Economic Geology show the Devils River Formation subdivided into upper and lower units, whereas all USGS mapping to date shows the Devils River Formation as a single undifferentiated unit (Blome et al., 2004). Thickness ranges from 540 to 670 ft in Medina and Uvalde Counties.

Capitol Aggregates Rio Medina Operation Medina County, Texas TCEQ-0585, Geologic Assessment

ATTACHMENT C Site Geology

Structure

The Site is crossed by Diversion Lake Fault (oriented North 68 degrees E) and downthrown to the southeast. This fault was mapped by the USGS (Small and Lambert, 1997; Small and Clark, 2000; Blome et al., 2005). The trace was identified in the field on the west side of the tributary to Elm Creek (Feature S-1-F on Figure 2B and 3B). The trace of this fault is marked by an area of fractured bedrock but with little observed effect on the topography (Attachment E, Photos 14 and 15). The path of the fault trace was examined for karst features on both west and east sides of the tributary. A detailed reconnaissance of the east side of the tributary was performed since the fault trace aligns with a northeast-southwest trending drainage (Figure 3B); however, with the exception of fractures, no karst features such as solution cavities or other recharge features were observed in association with the fault.

The USGS (Small and Lambert, 1997; Blome et al., 2005; Small and Clark, 2000) mapped two inferred faults at the southeast corner of the Site. The fault traces could not be identified in the field because of the soil cover.

Site Specific Geology

At the Site, the uppermost part of the Devils River Limestone has been removed by erosion. Strata characteristic of the middle part, present at the surface, were observed to be brecciated, or have vuggy porosity and abundant chert. The determination of the extent of erosion was based on the use of an argillaceous limestone marker bed logged in four of six borings conducted at the Site in 2014. The argillaceous limestone was used to divide Devils River Limestone into informal and site specific upper and lower units.

Six borings were drilled at the Site in July 2014; the logs of these borings were reviewed to assist in the interpretation of the Site geology. The following stratigraphic description was based on the boring logs (for boring locations see Figures 2A and 2B).

The two hilltops at the northern part of the Site are on the up-thrown block of the Diversion Lake Fault (see Figure 3B). Surface observations and review of logs for Boring B5 and B6 (Attachment F) indicate only the lower unit of the Devils River Limestone is present north of the fault. The argillaceous limestone marker bed was not logged in the borings, indicating that the upper part of the Devils River Limestone and the argillaceous limestone bed have been removed by erosion.

South and southeast of the Diversion Lake Fault in the area of the broad hilltop, which makes up the southern third of the site, an argillaceous limestone bed was logged in Borings B1, B2, B3, and B4 and was used as a marker bed to correlate the logs. South of the fault the lower unit and a portion the upper unit of the Devils River Limestone is present.

Potential for Fluid Movement from the Site to the Edwards Aquifer

The Site is located in the Recharge Zone of the Edwards Aquifer. In this Zone, highly faulted and fractured Edwards limestones outcrop at the land surface allowing water to infiltrate into the aquifer. The Medina Lake, a 6,066-acre surface water reservoir located north of the Site, was built partially on the Recharge Zone and contributes significant amounts of recharge to the Edwards Aquifer (Slattery and Miller, 2004).

The Devils River Limestone at the Site is primarily overlain by soils with very low to low infiltration rates, although rock outcrops do occur. The soils limit infiltration and allow precipitation to pond in natural topographic depressions and manmade ponds. It is anticipated that most recharge at the Site currently occurs in bed of the tributary to Elm Creek, although it is usually very difficult to detect exactly where recharge occurs.

With the development of the Site as a quarry, it is expected that recharge to the Edwards Aquifer will increase over current Site conditions. However, the quantity of recharge is anticipated to be relatively small compared to that of nearby Medina Lake.

ATTACHMENT D Site Geologic Maps

Site Geologic Maps are included as Figures 2, 2A, and 2B of this GA.

Capitol Aggregates Rio Medina Operation Medina County, Texas TCEQ-0585, Geologic Assessment

ATTACHMENT E Photo Log





Photo 3: S-5-O (Gray Bedrock)



Photo 4: S-4-O (White Rock Exposure) in foreground and alluvium in background



Photo 5: S-3-O (Gray Bedrock Ledge)





Photo 7: North Manmade Pond (S-8-MB) – red line added along earthen berms





Photo 9: Ridge above Limestone Cliff Ledges (S-2-O) – view to west



Photo 10: View from ridge to west of tributary to Elm Creek



Photo 11: Near Soil Boring 1 (B1) – view to west





Photo 13: Gray fractured bedrock in tributary to Elm Creek – view to southeast (near of crossing of Fault Trace S-1-F and tributary to Elm Creek)





Photo 15: Limestone ledge indicates presence of fault trace (S-1-F) - view to northwest



Photo 16: Rock Ledge (S-6-O) on up-thrown side of Diversion Lake Fault.



Photo 17: Middle Topographic Depression (S-11-CD) – view to southeast



Photo 18: Middle Topographic Depression (S-11-CD) – note soil accumulation





Photo 21: South Topographic Depression (S-12-CD) - view to south along interior fence line



Photo 22: South Topographic Depression (S-12-CD) - view to west of depression



Photo 23: General view at southwest portion of Site – view toward southwest corner of property



Photo 24: Limestone ledge at west-facing slope of hilltop located due west of Middle Manmade Pond (S-9-MB)



Photo 26: Dry creek of tributary to Elm Creek (south of feature S-7-O) view to southeast (up-stream)



Photo 27: Exposed bedrock in drainages east of tributary to Elm Creek (east of Soil Boring B3)



Photo 28: Limestone ledge in drainages east of tributary to Elm Creek (east of Soil Boring B3)


Photo 29: Middle Manmade Pond (Feature S-9-MB) after rain event - note accumulation of runoff



Photo 30: Middle Topographic Depression (S-11-CD) after rain event note muddy tracks and four inch diameter pit in bedrock that has collected water



Photo 31: North Water Well (S-13-MB) shed and tank

ATTACHMENT F Soil Boring Logs

6	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. B-1										
TYPE	OF I	BOR	ang: Continvour Cove	LO	CATI	on:	Pane	Property, Medina Com	itu		
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION ELEVATION:	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	MOISTURE		UNIT DRY WT.		
		AUgered	Clay Topsoil Hardgrag dence limestone W/WV954 zonestoc		ar s		· .	burs			
- 5-	j≹a-		fron-staining + solution features	5.0'	4.0'	80%	59%	Marrie Minne			
-/0-				5.0'	4i1'	827	35%	(19')			
-/3-			in take have brown lawhad	5,0'	5.V	100%	5820				
-25			thand gray viggy Limestone w/dense seam(221.8-22.8 2 w/red chan pockets below 23.6	5,0'	5.0	10070	478	Learhed & Collapsed members			
-30			Mixed have gray dense Limester have slightly yuggy limestone and have yuggy Limestone we have calcity peckets,	5.0'	5.0	1002	54%	21'-100.5' (79,5')			
-35			staining	5.0'	4,3'	863	36%				
-40			Hand lightgray brece iated	5.0	48	9670	70%				
-45			Limes Tone 43.4 Hard Higray dense limistone	5.0'	5.0	10070	882				
-522			What ially healed open vertical fracture @ 45.0 - 46.6 2 w/occ. solution features below, 48	5.0'	5.0	10070	83%				
	4		(continued)						-		
	COM	PL	ETION DEPTH:		DEPTH TO WATER:						
DATE:							DATE:				
L	-		DAIE.	DATE:							

:

1

6	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. B-1									
TYPE	OFI	BOF	RING: Continuous Core	LO	CATI	ON: B	ane	Property Medina Count	4	
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	N INSTURE		UNIT DRY WT.	
-55	•		- 5/ightly vuggy @ 55-65.3'	5.0'	5,0	10020	62%		,	
-/00-	pl > .			5,0'	5.0	10020	76%			
			- w/white soft seam@ 44,9'-65 63	5.0'	5.0	100%	75%		-	
12			Hard light gray & tan 5 1, Vugyte Linestonew Jocc. solution Flat Une Hard It, gray deuse limestonew J 5/8 "chert sean @ 67.7" 70"	5.0'	4.8	94070	57)			
-75			Moderately hard brown than be low ? argillaceous Linestone, crumbly be 192 Hard It gray dense limestone w/soft pockets of tam silty leached materia	5.0'	4,5'	90%	527			
-80-			8p.'	5.0'	<i>\$</i> ,8	9270	48%			
-02	-		Hard It. gray dense limestone w/iron-stained open tractures tvogs	5.0'	5.0	10070	78%			
00			Hard red - brown + light gray very vuggy Limestone brecciated Hard light gray dense Limestone	5.0'	3.5	70%	35%			
			ton silty leached pockets	5.0	4.7'	93%	5-97			
-95				5.0	4,7'	932	52%			
-100		H	(continued)							
	сом	PL	ETION DEPTH:		DEPI	TH TO	WAT	ER:		
			DATE:				DA	TE:		

.:

.*

C	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. B-1										
TYPE	OFI	BOR	EING: Continuous Core	LO	CATI	DN: t	Bane	Property Medina Can	nty		
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	CONTENT %		UNIT DRY WT.		
-105	•		Moderately hard tan argillaceous Lines tone w/softseams, - porsos below 104	5.0	4.8'	95%	5370	Regional Dense Nomber 100,5-140.0			
-///)	# .		-1"chert se am @ 108.8 -w/occ, iron-stained open fracture	5.0	4.0	80%	21%	(39,5')			
-115-			- prc. 51. hander, less arg illaceous	5.0	4:2	85%	28%		-		
-120-			seams	5,0	4.8	95%	40%				
-125-				5,0	5.0'	100%	592				
-136-				5.0	4.8	95%	28%				
-135				5.0	4.9'	98%	32%				
-140-	•		140	5,0	5.12	10070	74)2				
-145			Moderately hard It tan twhite grainy Limestone -occtan clayey laminations,	5.0	5.0'	100?	72%	Grainstone 140'-150 (10'+)			
-158-			' 10145 -146.3	5.0	4,5	90%	32%				
	CON		FTION DEPTH. 150 D'		DED						
	CUN	PL.	DATE: 7/21/2014		DEPI	нто	WAT	TE: 7/21/2014			

: "

$\frac{1}{12} + \frac{1}{12} $	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. B-2									
$\frac{1}{15} = \frac{1}{1000} + \frac{1}{10000} + \frac{1}{10000$	LBS/CD IT									
$\frac{-10}{-15} - \frac{1}{12} chert seam@ 11.0' = 5,0' 5,0' 100% 24% T.2' 5,0' 4.8' 96% 92% 17.2 Hard Tight grey slightly vuggy = 20 \frac{17.2'}{2.0'} 5,0' 4.8' 96% 92% 17.2 (22)' + tard dense light grey crystallineLinestme 5,0' 5.0' 100% 93% = 3'' chert seam @ 25.0'' 25.2' = 5,0' 5.0' 100% 93% = 25,2' = 3'' chert seam @ 25.0'' 25.2' = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 100% 93% = 5,0' 5.0' 5.0' 100% 93% = 5,0' 5.0' 5.0' 5.0' 5.0' 5.0' 5.0' 5.0' 5.$										
-20- Hard 15ght grey slightly vuggy 5,0 4.8 460 9213 -20- Hard dense lightgray crystalline Linestone 5,0' 5.0' 100% 93% Wloce, soft seam -253 "chert seam (225.0" 25.2"										
Harlts 1/c Lanue Magay linderty 0										
-30- Hard tan dense limestore 31.2' Hard Highry tan vuqqy limestore 5.0' 4.5' 90% 52% (88.3'										
-35- Hard 1t. gray dense limestore 5,0' 4,9' 98% 77% wlowc. yvgs 40- Hard 1t. gray tawvegsy limestore 47.1										
-45 Hard It. grag deuse limestruite filled W/occ. calcite pocket + calcite filled C. fracture 2 practure 2 gray be/ow 47.5' 50' 4.3' 86% 49%										
COMPLETION DEPTH: DEPTH TO WATER:	-									

./

. .

E	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. 8-2									
TYPE	OF	BOR	ung: Continuous Core	LO	CATI	DN:	Ban	e Property, Medinalou	naty	
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	* INSTRUCTOR		UNIT DRY WT. LBS/CU FT	
-55	•		- cherty below 51.5 Hard lightgray slightly Vuggy Limestone	5,0'	5,0'	100%	100%	Elacted tallapud	·	
-4.0-			Hard It, gray + tam grainy Limes tone Wocc. solution feature + healed vertial fractures 60.0	5.0	5.0	10070	97%			
			Hard Is. gray tossiliforous limestone - 7" chert seam @ 61.6 w/acc. solution features 655	5,0'	5.0	100%	100%		-	
-70			Hard tow slavgillereevs Limest me wleck, solution flatures Hard + moderately hard tout white arg: laceous times for 70.8	5.0	4,5	9070	79%			
70			Moderately hard tam + lighttan argillaceous cinestone v/bcc. hard brown porous limestone seams	5.0'	2.1	40%	0%			
-15-			80,F	5,0'	<u>4.0</u>	80%	1670			
			Hard 17, gray + tan dense limestone W/000. cale 140- filled sean +C" chert scam@ 80, 7 Hard tan+ gray Leached 1: mestone 850	5.0'	5.0'	100%	70%			
-90-	-		Hard Hig rag + tan trid brown dense limbstrue w/open + healta inclined fractures + sec. solution features	5.0'	5.0'	100%	92%	Leached &		
-95				5.0	5.0	100%	38%	member.		
-1/27				5.0	4.8	96%	87%			
			(continued)						-	
	CON	IPL	ETION DEPTH:		DEPT	THTO	WATI	ER:		
			DATE:				DA	TE:		
Repairment			· · ·				DA	1.4. Auto		



6	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. B-Z										
TYPE	OFI	BOF	RING: Continuous Core	LO	CATI	ON: E	Bane	Property Milina Count	ÿ		
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	S MOISTURE		UNIT DRY WT. LBS/CU FT		
156			moduately hard 14. t. an grange timestone	5,0	5,0	10070	.65 ×		/ / 		
				5,0	5.0'	10070	72%	Grainstone member			
-1664				5,0	5,0'	100%	85%	137.0 1.5			
7/65			*	5,0'	4.8	96%	7720	244			
			Moderately hard + hard 1+, 97-99 + tan dense slightly appillacement	5,0'	4.5'	9070	83)0	Gramstone			
-12/3-			Fractures	5,0'	3.0	100%	6270	membre			
-185			Hard tant light quy dence Limestone wijtoh-stained Fractures + occ, solution features	5,0'	5.0'	100%	83%	Kirschberge			
-198-				5,0	5.0'	1003	8990	member of for Kainer for			
-195			Hard brown voggy imestance, reached wird any - filled fractures	5.0	4.4	88%	35%	Kirschburg Evaported			
-2.00			110 CION Scame (0193.0	5.0	5.0'	100%	82%				
	CON	ur L	DATE: 7-19-2014		DEP	н то	WAT	ER: .TE:			

!"

6	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. 8-3										
TYPE	OFI	BOF	RING: Continuous Core	LO	CATI	ON:	Ban	e Property, Medinala	unty		
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE	CORE RECOVERED	CORE RECOVERY %	Noiseure.		UNIT DRY WT.		
-5-	•		Hard light gray dense limestone Wlocc. Solution features + iron-stained vertical grenthealed	3.5	3.3	95%	4870				
-10-			fractures + occasional clay pockets - 1"chart pocket@8.0"	5,0	4,5	90%	663	Collapsed members	-		
-15-				5.0	5:0	10070	josto		- - -		
-20-			- "Intern Otto a"	5.0	4.D	80%	47%				
-25			-2,5 cheriseam @ 20,9 -vuqqyzone 23-25'	5,0	5.0	10070	71%				
-30-			- VUAqy zone 28'-29.2"	5.0	<i>4</i> .9	98%	73%				
-35-				5.0	4.0	80%	77%				
-40-	•		÷ 	5.0	4.2	85%	322				
-45-			Hardlight gray very vuggy limeston highly broken	e 5.0	1.3	35%	25%		-		
-50			Hard light gray dense limestone	5.0	3.5	70%	19%	· · · · · · · · · · · · · · · · · · ·			
	-		(continued)						-		
	COM	PL)	ETION DEPTH: DATE:		DEPI	THO	WATI	ER:			
Remainstration	DATE: DATE:										

.....

:

TYPE OF BORI	NG: Continuous Core	LO	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. 13-3										
		THE OF BURLING: COTTINUOUS CORE LOCATION: DUNE Troperty, Medinalaunty											
DEPTH FT SYMBOLS SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE	CORE RECOVERY %	Same and the second sec		UNIT DRY WT.						
	55,0	5,0	3.9	78%	78%		×						
H H	Hard tan + gray very vorgy 56,5 linestone buloic, clay in yugs tard It, grayfoss; liferous dense limestone vocc. tan are illocome lamination	5.0	4.5	90%	48%		-						
	stained Fracture 600 Hard H. groy dense limestone w/vertic inclined Wealed + open fractures w/clay in some fractures	el 5.0	5;0	10070	882	Leached +	-						
-70-	695	5,0	5.0	เงฮ์โอ	70%	collapumbers Members 2.5'-80.5'							
-75-	ense limestme + argillaceous inistone w/1 ron-stained vertical solution fracture@72,5-73.5	5.D	5.0	1000	95%	Leached +							
	-5" clay seam @ 77,6" hard+ deme below 78,5" 80.4	5,0	5.0	100%	77%	members 2,5-80,5	-						
	roderately hard tan argillaceous inestone wlocc, hard tossiliterous seams	5.0	5.0	100%	74%	Regional							
-90-	i j	5.0	5.0	100%	52%	member 80.5,05							
-95-		5,0	5:0	100%	6220								
-102		5,0	5.0	100%	47%	~							
	(continued)						-						
COMPLET	COMPLETION DEPTH: DEPTH TO WATER:												
	DATE:				DA	TE:							

.....

.*



.*

6	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. 8-3									
TYPE	OF	BOF	RING: Continuous Core	LO	CATI	ON:	Ban	e Property Medinale	unty	
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION elevation:	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	MAIABURE		UNIT DRY WT.	
	÷		155.0'	5.0	2,5	51%	19%			
755	ulta .		Hard tan dense + slightly vuggy Limestore w/red clay pockets	5.D	3,1	6270	31%	Kirschberg		
				5.0	3,2	6390	322	131,8-205	-	
-170-			-w/occ red clay laminations	5.0	5,0	100%	77%			
-176				5,0	4,2	85%	47%			
-108				5.0	5,0	10070	4570			
180				5.0	3.7'	73%	53%			
-100			÷.	5.0	2.5	100%	3270			
_195				5.0	2.6	52%	30%			
-1.02			Hard ton dense limestonew/occ. solution teatures topen vertical tractures + redelay lamination 3 + occ., chest pocket	5,0	3.0	60%	43%		-	
			(continued)							
	COM	PL	ETION DEPTH:		DEPI	THTO	WATI	ER:		
L			DATE:				DA	TE:		

;

2

E	A)	CAPITOL AGGI LOG OF BO	REG/	TE: G N	S, L' O.	Г D . В	3	
TYPE	OF	BOR	ING: Continuous Core	LO	CATI	ON:	Ban	e Property Medinals	unty
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	MOINTURE S		UNIT DRY WT.
-905			Hard tandeme / mestare co/occ so/otran features	5,0	4.2	85%	55%	Kirschberg Evaporite 131.8-205	×
	jų kari					•			
			*				•		-
						-			
					6.				
			2						
									-
	5								-
	COM	PL	DATE: 7-25-2014		DEPI	TH TO	WATI DA	ER: 185' TE: 7-25-2014	

÷

:

6	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. B-4									
TYPE	OFI	BOR	EING: Continuous Core	LO	CATI	on:	Ban	e Property Medinals	unty	
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	AMAIATURE		UNIT DRY WT.	
-5	<u>.</u>	Augered	Broken timestone 5.0		n. *		•	Cyclic typanine	×	
	ults .		Hard It gray dense broken limestone, vuggy@5.0-6.5; chertpacket@7.9'	5,0	4.8	97%	437.	Cyclic + marine	-	
			-brown leached moderately hand pocket @ 11.6'-11.8'	5.0	4.9	98%	75%	1.0-25	-	
-20			Mixed harddensegray + lightgray limestone + hard vuggy limestone	5.0	5,0	10070	53%	cyclic + marine members		
40			25,0	5.0	5.0	10070	6270	1.0-25.0		
-20-			Hard dense brecciated+reheated It.group Limestone w/fractured +broken zones	5.0	5.0	10070	72%	Leached + Collagred		
-36				5.0	5.0	100%	75%	members 25-108.2		
-40			-chert pocket@38,5'	5.0	5.0	10070	49%			
-46			-chertpocket@42.7'	5.0	4.3	873	47%			
-50				5.0	4.9	98%	55%			
	5		(continued)						-	
	сом	PLI	ETION DEPTH:		DEPI	т то	WATI	CR:		
	•		DATE:				DA	TE:		

.....

.

6	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. 8-4									
TYPI	COFI	BOR	RING: Continuous Core	LO	CATI	ON:	Ban	e Property, Medinala	unty	
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	MANATURE		UNIT DRY WT.	
-55	•••		Hard dense It. groy crystalline Limestone	5,0	5,0	100%	69%	Leached+ collapsed	N 1 1	
-60-	1		Hard tan + lightgray very dense fine-grained limestone w/breccinted and rehected zones + occ. vugs w/ inn stains	5.0	4.2	83%	42%			
-65-			*	5.0	4.8	97%	72%	Leachil	-	
-70-				5,0	5.0	1007.	75%	Callagued		
-75-			- gray below 72' -wlocc, calcite seams tocc.	5.0	3.0	602	86%			
-80-			clayey laminations	5.0	5.0	100%	85%			
-85-			85.0	5,0	3.8	75%	56%			
-90	•		Hard tan dense limestone w/occ. leached seams + inclined open fractures + cherty zones	5.0	5,D	1000	78%		-	
-95-				5.0	5.0	10070	892			
-100-			Hard tan calciteand light gray limestone	5.0	3.9	78%	662			
	:		(Continued)						-	
	COMPLETION DEPTH: DEPTH TO WATER:									
L			DATE:	-			DA	TE:		

.

:

E	CAPITOL AGGREGATES, LTD. LOG OF BORING NO. 8-4									
TYPE	OF	BOI	RING: Continuous Core	LC	CATI	ON:	Ban	e Property, Medinale	unty	
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE	CORE	CORE RECOVERY %	MOISTURE		UNIT DRY WT.	
-1.05	÷		-2'void 102'-104'	5.0	1.9	38%	32%	Leached + collapsed members		
	J.S.		Vertical Fracture 108,2 Mixed hand + moderately hard	5.0	4,3	86%	17%	25-108,2	-	
			tan dense limestone and argillaceous limestone whoisps of clayey material	5,0	5.0	100%	56%	dense member 108.2'-130'		
-120				5.0	4,9	98%	1670	(21.8)		
			Hard tan dense limestone, w/	5,0	5,0	10070	19 %	P. A.		
-139			Solution fratures + red clay highly broken, mixed WI moderately hardtan argillaceous lime stone wited clay below 125 130'	5,0	3,8	77%	25%	dense member		
-/35			Mixed Moderately hard gramy limestone + harddense It.gray Limestone, ironstainedw/red clay+ vertical open fractures	5.0	2,4	48%	10%	Grainstone member		
-146-	•		e V	5.0	1.7	33%	12%	(15)		
-)45			145'	5.0	3.8	7570	32%			
-1517			Hard light gray dense limestone, broken w/numerous voids	5.0	1.2	23%	0%	Kirschberg Evaporite 145-190	-	
	•		(continued)						-	
	СОМ	PLI	ETION DEPTH:		DEPT	HTO	WATE	R:		
L			DATE:				DAT	ГЕ:		

فري و

...

6	$\begin{array}{c} \textbf{CAPITOL AGGREGATES, LTD.} \\ \textbf{LOG OF BORING NO.} B-4 \end{array}$										
TYPE	OF	BOI	RING: Continuous Core	LO	CATI	on:	Ban	e Property Medinals	unty		
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE RECOVERED	CORE RECOVERY %	A MAIRTURE		UNIT DRY WT.		
			155'	5,0	1.0	2010	0%	Kirschberg Evaporite, 145-190	× 1111		
			Hard gray-brown dense limestone, 157 Moderately hard brown + tan leached porous crumbly limestone 160'	5,0	3,7	73%	33%				
			Hard tan dense limestane W/solution features + red clay + open fractures, below 164	5,0	3,7	73%	50%		-		
-110-			טרו	5,0	2,6	52%	13%				
-175			Hardlightgroydense Brecciated limestone w/num, voids tred clay	5,0	1.9	38%	15%	Kirschberg Evaporite			
-180-				5,0	3,1	62%	25%	145-190 (45')			
-185				5,0	2,0	40%	1770				
-198-	•		: 	5,0	2.8	578	3270				
					1						
	COM	I IPL	ETION DEPTH: <u>190.0'</u> DATE: <u>7/25/2014</u>		DEPI	TH TO	WATI DA	ER: None TE: 7/28/2014			

*

6	A)	CAPITOL AGGR LOG OF BC	EGA	ATE IG N	S, L'	ГD. В-3	Г	
TYPI	EOF	BOE	RING: Continuous Core	LC	CATI	ON:	Bar	ne Property Medina la	enty
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE DRILLED	CORE	CORE RECOVERY %	RQD		UNIT DRY WT.
-5-			Mixed hard tan dense limestone and moderately hard 14. tan Chalky limestone, w/solution featbres and leached pockets	4.0	3.8	95%	4220		
-10-	the .		10,0	5.0	3,8	7670	17%		
			Hard It. Tan duse / mestore	5,0	4.9	9870	682		
-20-			- 5" clay seam@ 19,2" - 1+,9 tay + 9 tay be 10 20"	5,0	5,0	ivolo	69%		
-2.5-				5,0	5,0	100%	8370		
-30-			Mixed hard and moderately hard	5.0	5.0	100%	7778		
-35-			It. tan and white grainy porous limestone	5.0	5,0	100%	61%		
-40-			j. J	5.0	5.0	100%	70%	-	1111
-45				5.0	5.0	100%	90%		
-50			-soft@ 49,0'-49,4'	5,0	5.0	100%	64%		
	:		(continued)						-
	COM	PLE	TION DEPTH:		DEPT	HTO	WATE	R:	
L		-	MALL:				DA'	TE:	

1

•••

G	2		CAPITOL AGGR	EG	ATE	S, L	TD.		····
15	5	/	LOG OF BC	RIN	IGN	IO.	B-3	5	
TYPE	OF	BOR	EING: CONTINUOUS COTE	L	DCAT	ION:	Bai	ne Property, Medina la	nty
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE	CORE	CORE RECOVERY %	Summer R		UNIT DRY WT.
-55	•		Mixed hard and moderately hard It. tan and white grainy porous limistone	5.0	3.0	100%	82%		
-60-	plis .			5.0	5,0	100%	71%		-
-45-				5,0	5.0	10070	71%		
-70-				5.0	5,0	100%	79%		-
-75-			75,0'	5,0	5.0	100%	80%		
-80-			Moderately hardwhite grainy rorous limestone w/oecasional solution teatures	5.0	<u>\$</u> ,0	100%	33%		
-85-			-2"brown day seam @81.b 82.3 Moderately hand + hard /t. gray dense limestore, brecciated w/	5.0	5.0	100%	6670		
-90-			solution teatures	5.0	5.0	10070	7670		
-95-			Hard gray dense limestone Wisolution features 29 "chert@92.1	5.0	4,9	983	59%		
-1002-			Hard gray slightly vuggy limestone 3.5" chert@98.3"	5.0	4.5	90%	8%		
			(continued)						-
•	COM	PLE	TION DEPTH:		DEPT	HTO	WATE	R:	_
L	•		DATE:	-			DA'	ТЕ:	

1

•••



••••

TYPE OF BORING: Continuous Core LOCATION: Bane Property, Medinal Hard SOIL DESCRIPTION Bugging BIANNS SOIL DESCRIPTION Bugging Hard Hard light gray dense limestone 4.5 Soft white chalk @.Y'-2.4' 4.5 Soft white chalk @.Y'-2.4' 5.0' Soft white cha	UNIT DRY WT.
$\frac{1}{12} \frac{1}{12} \frac$	UNIT DRY WT.
Hard light gray dense limestone 2 soft white chalk @ 1.7'-2.4' -soft white chalk @ 4.5'-4.7' -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10- -10-	····
-10 -10 -15 -w/red alay below 14.5' 5.0' 2.0' 40% 9% 5.0' 3.5' 70% 18%	
-w/red clay below 14.5' 5.0' 2.0' 40% 9% 5.0' 3.5' 70% 18%	-
5.0' 3.5' 70% 18%	-
-20	
- fossiliferous below 22'w/solution 5.0' 5.0' 100% 90% features	
-30- -30-	
-35-	
-w 7'' chert @ 38.7' 5.0' 4.3' 86% 75%	
-9, ray & vuggy w/ tan moderately hard, -45 -9, ray & vuggy w/ tan moderately hard, Harded pockets @ 44, 2'-45.6	
-50 4.8' 9672 50%	-
COMPLETION DEPTH.	-
DATE: DATE:	

.

:

6	A)	CAPITOL AGGR LOG OF BO	EG/	TE: G N	S, L'	ГD. В-6	; ;	
TYPE	OF	BOR	ung: Continuous Core	LO	CATI	ON:	Ban	e Property, Medina la	nty
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE	CORE	CORE RECOVERY %	Summon Red		UNIT DRY WT.
-55			Hard light gray dense limestone	5,0'	5.0	10070	672		···· -
	t∦⇒.		Hand light arm, dense fossili from	5,0'	4,5	90%	53%		-
			limestone w/ highly broken zone @ 62-65,5	5.0	2.7	542	8%		-
45				5.0'	1.6'	3270	970		
26			74.2	5,0'	4.7'	94%	54%		-
			Moderately hardlight tan dense chalky Timestone	5.0	:4.0	80%	19%		
20			84.7	5.0	3.8	76%	24%		
-90-			Hard 17. gray dense limestone - 6"chert@ 86.9"	5.0'	2.4	48%	2020		
45			Moderately hard light tan chalky limestore w/tan cloyey laminatio to 95	5,0'	4.0'	80%	17%		
-1/19			Hard It gray tossiliterous limestone	5.0	4.1	823	2270		
100			(continued)						-
	сом	PLE	TION DEPTH: DATE:		DEPT	HTO	WATE	R:	_
here and the second		-			-		DA		

:

6	A)	CAPITOL AGGR	EGA	TE	S, L	ΓD.	4	
TYPE	COF	BOF	ING: Continuous Core	LO	CATI	0. 4 0N:	Bai	ne Property Medinala	inte
DEPTH FT.	SYMBOLS	SAMPLES	SOIL DESCRIPTION	CORE	CORE RECOVERED	CORE RECOVERY %	Juneon Rad		UNIT DRY WT.
-105			Hard 17, gray fossiliferous limestone Wopenvertical Fracture @ 100-101 102.5 Moderately hard tan chalky Timestone w locc. hard light gray	5.0	5.0	10070	72%		
-110-	ţ₿Ş.		Noden tely hand light tan araillaceous	5.0'	4.8	100%	41%		
			limestone w/tan clayey lanii nations	5.0'	5.0	1000	8670		-
-120			Hard light gray dense limestone W/moderately hard white Chalk seams	5.0'	5,0'	105%	7470		
-125			Hard lightgray dense + slightly	5.0'	4.3	86%	54%		-
-136			porous limestone wloccasionel' inclined open frecture	5.0	.¥7'	9470	43%		
126			131.0 Itard white slightly Vuggy limistone 133,0 Hard light gray brecciated limestone	5,0	48	96)0	77%		
-1)3			- gray w/ opensolution features and calcite below 137.0"	5,0'	5.0'	100%	90%		
-140				5,0	4.9	98%	84%		
-158			Mixed hardgray Vuggy limes tone and moderately hard gray porous limes tone, leached + sugary texture Would dense crystalline 15. seams	5.0	2,4	4870	277	· · · · · · · · · · · · · · · · · · ·	
	COM	PLI	(Continued)		DEPT	THTO	WATI	TD.	-
			DATE:				DA	TE:	

.....

. .

. :

TYPE OF BORING: Continuous Core LOCATION: Bane Property, Medi: Lit Soil Description Handore Budden Soil Description Soil Budden Budden Soil Budden Soil Budden Budden Soil Description Soil Budden Budden Soil Description Soil Budden Budden Soil Budden Soil Budden Bud	Na County TAN AND TANA MAL								
Li Soil Description Boord Boord ELEVATION: Boord -155 Havd tan dense limestore -162 5.0'	UNIT DRY WT.								
-155: Hard tan devise limestore -162 -162									
-162 Hard tan devise limestone 5.0' 4.7' 9470 722	-								
	1 -								
5.0 0,4 8% 0%	-								
Hard light gray brecciated and healed vuggy limestone 5.0' 4.0' 80% 32%									
-175 5,0' Z.3' 46% 18%									
-180- -180-									
-185	-								
-w/dense limestone seame 5.0 4.1' 822014%	-								
-195	· –								
5.0 2.3 4670 770									
COMPLETION DEPTH: 200.0' DEPTH TO WATER: 1/on 2	COMPLETION DEPTH: ZOD.D' DEPTH TO WATER: 1/one								

:

ATTACHMENT G

Well Registration and Construction Logs

STATE OF TEXAS WELL REPORT for Tracking #121567								
Owner:	BANE LOREN D	Owner Well #:	No Data					
Address:	12623 HUNTER'S CHASE SAN ANTONIO , TX 78230	Grid #:	68-33-1					
Well Location	n: 1580 CR 265 MICO , TX 78056	Latitude:	29° 28' 19 " N					
Well County:	Medina	Longitude:	098° 58' 50" W					
Elevation:	No Data	GPS Brand Used:	No Data					
Type of Work	<: New Well	Proposed Use:	Domestic					
Drilling Date:	Started: 10/17/2003 Completed: 10/22/2003							
Diameter of Hole:	Diameter: 8 in From Surface To 420 ft							
Drilling Method:	Air Rotary							
Borehole Completion:	No Data							
Annular Seal Data: 1st Interval: From 0 ft to 226 ft with 12 BENTONILE (#sacks and material) 2nd Interval: From (No Data) ft to (No Data) ft with 4 (#sacks and material) 3rd Interval: No Data Method Used: BENTONILE POLYMOR GROUT PUMPED THROUGH Cemented By: TREMMIE PIPE BY DRILLER Distance to Septic Field or other Concentrated Contamination: No Data Distance to Property Line: No Data Method of Verification: OWNER Approved by Variance: No Data								
Surface Completion:	Surface Slab Installed							
Water Level:	Static level: 210 ft. below land surface o Artesian flow: No Data	on (No Data)						
Packers:	3 220 MOLDED RUBBER							
Plugging Info:	Casing or Cement/Bentonite left in well: No I	Data						
Type Of Pump:	Submersible Depth to pump bowl: 340 ft							
Well Tests:	Jetted \ Estimated Yield: 20 GPM with (No Data) ft drawdo	wn after (No Data) h	ours					

Attention Owner: Confidentiality Privilege Notice on reverse side of owner's copy.	P.O. Box	as Department of Water Well Driller/F 12157 Austin, Texas 787 Toll free (I Email address: <u>water.</u> WELL	License and ump Installer Pro 11 (512)463-786 00)803-9202 well@license.s REPORT	i Regula Ogram 10 FAX (512) tate.tx.us	tion 2 463-8616 2	This form must be or and filed with the de and owner within 6 upon completion of	partment partment) days he well.
I OWNED	AS STREET, AND			4.45.6.5			
BANE, LOREN D	Address 12623	HUNTER'S CHAS	e SAN 1	ANTON	10 State 7 Y	Zip 78	230
ZI WELLE LOCATION	Contra Plan y at Raining			-10.00578()#"S			
County MEDINA	Physical Add	CR265	City MIC	0	State T-1	Zip 784	256
3) Type of Work Wew Well Reconditionin Replacement Deepening	Lat. 4) Propose Industrial Rig Supp	d Use (check) U Ma Inigation U Injection In If Public	Long. mitor Enviro stion Public : Supply well, went	nmental Soil Supply 🛄 e plans submi	Grid # Boring Domestic De-watering Testwe Ited? Yes No	<u>68 33 4</u> 5)	<u>~/</u> 11
6) Drilling Date Started <u>10 / 17 / 03</u> Completed <u>(0 / 22 / 03</u>	Dia.(in)	From (ft) To C 2	(ft) U Ai -20 U Ai	ir Rotary ir Hammer [aod (check) U Driv Mud Rotary D Box Cable Tool U Jeta	ren red X	
0-80 HARO 80-260 (1	FLINT .	+ Cinestow	8) Bo	rehole Co Under-rean avel Packed	mpletion Open red Gravel Packed rive the interval from	Hole C Straigh	t Wall to bot t
320-340	While LT man	Hand Limster	2. Dia. (in.)	New Or Used	Steel, Plastic, etc. Perf., Slotted, etc Screen Mfg., if comme	Setting (fl) reisil From To	Gage Casing Screen
340- 420	PARTIN	Rotorus	4/2	N	Schto PUL Slotted	0 420 370-420	
(Use reverse side of V	Vell Owner's copy,	If necessary)	9) Ci Cen	ementing I centing from addised Que	Bata 2 26 ft to ft to the ft to	_fl. #of sacks used fl. #of sacks used	12 Be, 4 Pe,
13) Plugged Gwell plug Casing left in well: Cement/Beaton	ged within 48 l	IOUIS	Ceme Distar Metho	nting By <u>The</u> ice to septic s of of verificat	ystem field or other concestion of above distance	they Or ile	mft.,
			10) S 	urface Co cified Surfac cified Surfac	mpletion e Slab Installed e Skeve Installed		
14) Type Pump Turbine Jet Other	Submersible	Cylinder	DAp	ess Adapter (proved Altern	Jsed ative Procedure Used		
Depth to pump bowls, cylinder, jet etc 15) Water Test Typetest Pump Bailer J. Yield: 20 Krom with ft d	etted D Estimat	ed brs.	11) Static Artesi	Vater Levi level_212 an Flow	elft. below Date gpm. Date	<u> </u>	
16) Water Quality Did you knowingly penetrate a strata v YES T NO If yes, did you subm Type of water Was a chemical analysis made Type	which contain und nit a REPORT OF Depth of S	esirable constituents. UNDESIRABLE WATER Stata	12) P 3	'ackers Moldee	Type (lubba	Depth 2220	2
Company or individual's Name ((type or print)	Pipe Creek	pala W	ell in	Lic. No.	24500	1
Address <u>to Box</u> Sinsture Rander Poble	6333 vt	s 11 / 14 / 93	City - p	<u>e Cireo</u> UP M	K State 7)	Zip76	063 L
BERGER BERGER BERGER WERE RECEIPTED AND AND AND AND AND AND AND AND AND AN	Contraction of the second second		allerin State in the second	A CARLES AND A		Contraction of Some series	

GRUNDFOS



2 and 3 HP Submersible Pump Protection Application

Dear Grundfos Dealer:

To make certain your customer's Grundfos submersible pump is fully covered under the supplemental *Performance PLUS* protection program, READ AND FOLLOW THESE INSTRUCTIONS CAREFULLY. Please type or print clearly.

Complete this policy application in full. Then sign and date the application.

Your customer should keep the top white copy of the application. Keep the yellow copy for your records and send the two remaining copies (pink and goldenrod) to your Grundfos distributor.

Should the pump fail due to any of the conditions specified under *Performance PLUS* prior to the expiration of the 5-year term of the policy, your customer should contact you. You then pull the pump and determine if the cause of failure is within the terms of the *Performance PLUS* policy. If it is, Grundfos will repair or replace the pump at no charge through your distributor. *Performance PLUS* does not cover the labor charges involved in pulling and reinstalling the pump.

The *Performance PLUS* protection policy covers all Grundfos domestic submersible pumps 2 and 3 HP. It is limited to the original owner of the pump and is non-transferable.

1 14 Scal 80

No. B 17773 Customer Name LOREN D. BANE Address 12623 HUNTORS CHASE City <u>S. A.</u> State <u>TEX</u> zip <u>78230</u> Grundfos Dealer Name CAZASELER Address 603 2046 City Honde State Top Zip 78861 Pump Model: 16530-24 Serial Number 032595 Phase & Volts: Horsepower: 230 V 36 Date Installed: Application: Stockwell 1 6/81 Pumping Water Level: Pump Setting: 260 320 Well Inside Diameter: Depth: 346 5.0 Flow (GPM): Lightning Arrestor Type: DE AR gat open 2! Brand of Pump Replaced: New Installation or Jouble Replacement: Press Tank wx 302 I certify that a proper lightning arrestor has been installed with this Grundfos submersible, and that the unit has been installed in accordance with the Grundfos Installation and Operating Instructions. Dealer Signature: Date:

Policy

SP-SL-015

NEW PRINTED IN U.S.A.

1/26/85

CUSTOMER COPY

Aboveground Storage Tank Facility Plan Application

Texas Commission on Environmental Quality

For Permanent Storage on The Edwards Aquifer Recharge and Transition Zones And Relating to 30 TAC §213.5(e), Effective June 1, 1999

To ensure that the application is administratively complete, confirm that all fields in the form are complete, verify that all requested information is provided, consistently reference the same site and contact person in all forms in the application, and ensure forms are signed by the appropriate party.

Note: Including all the information requested in the form and attachments contributes to more streamlined technical reviews.

Signature

To the best of my knowledge, the responses to this form accurately reflect all information requested concerning the proposed regulated activities and methods to protect the Edwards Aquifer. This **Aboveground Storage Tank Facility Plan Application** is hereby submitted for TCEQ review and Executive Director approval. The application was prepared by:

Print Name of Customer/Agent: J. Brandon Klenzendorf, P.E.

Date: 6.14.2023

Signature of Customer/Agent:

Brandon Kh

Regulated Entity Name: Capitol Aggregates Rio Medina Operation

Aboveground Storage Tank (AST) Facility Information

1. Tanks and substance stored:

Table 1 - Tank and Substance Storage

AST Number	Size (Gallons)	Substance to be Stored	Tank Material	
1	10,000	Diesel fuel	Steel	
2	600	Grease Lube	Steel	
3	250	Hydraulic Oil	Steel	
4	250	Gear Oil	Steel	



AST Number	Size (Gallons)	Substance to be Stored	Tank Material
	AST 5 - 250 AST 6 -250 AST 7 - 500 AST 7 - 500 AST 8 - 500 AST 9 - 500 AST 10 - 5,000 AST 11 (unregulated contents; no associated plan fee) - 450 AST 12 (unregulated contents; no associated plan fee) - 250	AST 5 - Motor Oil AST 6 - Transmission Oil AST 7 - Used Oil AST 8 - Gasoline AST 9 - Future Undecided Petroleum Product AST 10 - Future Undecided Petroleum Product AST 11 - Diesel Exhaust Fluid Additive AST 12 - Antifreeze	
5	(18) 55 gallon drums (no associated plan fee) - 990 gallons total	Drums 1 through 18 - Used Oil, Hydrualic Fluid, Drive Train Fluid	Steel

Total x 1.5 = regulated tanks AST 1 through AST 10 and 18 55-gal drums plus displacement volume of unregulated tanks AST 11 and AST 12 = 29,335 Gallons

2. The AST will be placed within a containment structure that is sized to capture one and one-half (1 1/2) times the storage capacity of the system. For facilities with more than one tank system, the containment structure is sized to capture one and one-half (1 1/2) times the cumulative storage capacity of all systems.

Attachment A - Alternative Methods of Secondary Containment. Alternative methods for providing secondary containment are proposed. Specifications that show equivalent protection for the Edwards Aquifer are attached.

3. Inside dimensions and capacity of containment structure(s):

Length (L) (Ft.)	Width (W) (Ft.)	Height (H) (Ft.)	L x W x H = (Ft3)	Gallons	
38.5	35.2	3	4,066	30,413	

Table 2 - Secondary Containment

Length (L) (Ft.)	Width (W) (Ft.)	Height (H) (Ft.)	L x W x H = (Ft3)	Gallons

Total: <u>30,413</u> Gallons

4. All piping, hoses, and dispensers will be located inside the containment structure.

Some of the piping to dispensers or equipment will extend outside the containment structure.

🔀 The piping will be aboveground

The piping will be underground

- 5. The containment area must be constructed of and in a material impervious to the substance(s) being stored. The proposed containment structure will be constructed of <u>concrete dike and concrete flooring sufficiently impervious to petroleum products plus a</u> <u>roof system</u>.
- 6. Attachment B Scaled Drawing(s) of Containment Structure. A scaled drawing of the containment structure that shows the following is attached:

Interior dimensions (length, width, depth and wall and floor thickness).

- Internal drainage to a point convenient for the collection of any spillage.
- \boxtimes Tanks clearly labeled.
- Piping clearly labeled.

Dispenser clearly labeled.

Site Plan Requirements

Items 7 - 18 must be included on the Site Plan.

7. The Site Plan must have a minimum scale of 1'' = 400'.

Site Plan Scale: 1" = <u>100</u>'.

- 8. 100-year floodplain boundaries:
 - Some part(s) of the project site is located within the 100-year floodplain. The floodplain is shown and labeled.
 - No part of the project site is located within the 100-year floodplain.
 - The 100-year floodplain boundaries are based on the following specific (including date of material) sources(s): Floodplain data obtained from Federal Emergency Management Agency (FEMA) Flood Insurance Rate Map (FIRM) for Medina County, Texas and unincorporated areas, map number 48325C0375D, 15 May 2020.
- 9. The layout of the development is shown with existing and finished contours at appropriate, but not greater than ten-foot contour intervals. Show lots, recreation centers, buildings, roads, etc.

\boxtimes	The layout of the development is shown with existing contours.	Finished topographic
	contours will not differ from the existing topographic configuration	ion and are not shown.

10. All known wells (oil, water, unplugged, capped and/or abandoned, test holes, etc.):

There are <u>8</u> (#) wells present on the project site and the locations are shown and
labeled. (Check all of the following that apply):
\bigtriangledown The wells are not in use and have been preperly abandoned

 \square The wells are not in use and have been properly abandoned.

The wells are not in use and will be properly abandoned.

The wells are in use and comply with 16 TAC § 76.

There are no wells or test holes of any kind known to exist on the project site.

- 11. Geologic or manmade features which are on the site:
 - All sensitive geologic or manmade features identified in the Geologic Assessment are shown and labeled.
 - No sensitive geologic or manmade features were identified in the Geologic Assessment.

Attachment C - Exception to the Geologic Assessment. A request and justification for an exception to a portion of the Geologic Assessment is attached.

- 12. The drainage patterns and approximate slopes anticipated after major grading activities.
- 13. Areas of soil disturbance and areas which will not be disturbed.
- 14. 🖂 Locations of major structural and nonstructural controls. These are the temporary and permanent best management practices.
- 15. \square Locations where soil stabilization practices are expected to occur.
- 16. \square Surface waters (including wetlands).

🗌 N/A

17. \square Locations where stormwater discharges to surface water or sensitive features.

There will be no discharges to surface water or sensitive features.

18. \square Legal boundaries of the site are shown.

Best Management Practices

19. Any spills must be directed to a point convenient for collection and recovery. Spills from storage tank facilities must be removed from the controlled drainage area for disposal within 24 hours of the spill.

In the event of a spill, any spillage will be removed from the containment structure within 24 hours of the spill and disposed of properly.

In the event of a spill, any spillage will be drained from the containment structure through a drain and valve within 24 hours of the spill and disposed of properly. The drain and valve system are shown in detail on the scaled drawing.

20. All stormwater accumulating inside the containment structure will be disposed of through an authorized waste disposal contractor.

 \boxtimes Containment area will be covered by a roof.

Containment area will not be covered by a roof.

A description of the alternate method of stormwater disposal is submitted for the executive director's review and approval and is attached.

- 21. Attachment D Spill and Overfill Control. A site-specific description of the methods to be used at the facility for spill and overfill control is attached.
- 22. Attachment E Response Actions to Spills. A site-specific description of the planned response actions to spills that will take place at the facility is attached.

Administrative Information

- 23. A Water Pollution Abatement Plan (WPAP) is required for construction of any associated commercial, industrial or residential project located on the Recharge Zone.
 - The WPAP application for this project was approved by letter dated <u>22 June 2016 for</u> <u>original application; and 24 August 2018 for the WPAP modification</u>. A copy of the approval letter is attached at the end of this application.
 - The WPAP application for this project was submitted to the TCEQ on _____, but has not been approved.
 - A WPAP application is required for an associated project, but it has not been submitted.

There will be no building or structure associated with this project. In the event a building or structure is needed in the future, the required WPAP will be submitted to the TCEQ.

- The proposed AST is located on the Transition Zone and a WPAP is not required. Information requested in 30 TAC 213.5 subsection (b) (4)(B) and (C) and (5) is provided with this application. (Forms TCEQ-0600 Permanent Stormwater Section and TCEQ-0602 Temporary Stormwater Section or Stormwater Pollution Prevention Plan/SW3P).
- 24. This facility is subject to the requirements for the reporting and cleanup of surface spills and overfills pursuant to 30 TAC 334 Subchapter D relating to Release Reporting and Corrective Action.
- 25. Submit one (1) original and one (1) copy of the application, plus additional copies as needed for each affected incorporated city, groundwater conservation district, and county in which the project will be located. The TCEQ will distribute the additional

copies to these jurisdictions. The copies must be submitted to the appropriate regional office.

26. Any modification of this AST Facility Plan application will require executive director approval, prior to construction, and may require submission of a revised application, with appropriate fees.
Attachment A – Alternative Methods of Secondary Containment

Not applicable. The ASTs will be placed within a containment structure that is sized to capture at least one and one-half $(1 \frac{1}{2})$ times the cumulative storage capacity of all systems.

Attachment B – Scaled Drawing(s) of Containment Structure

Scaled Drawings of the new containment structure at the Tank Storage Area associated with the new storage tanks (AST 1 through AST 12) are provided below. The Scaled Drawings include the tank locations, tank dimensions, associated piping, overfill alarm, and fuel dispensers clearly labeled. The new tanks that will be housed in the new containment system at Tank Storage Area Project Site, plus planned future tanks, are described below:

- AST 1 is a 10,000-gallon diesel fuel steel permanent tank;
- AST 2 is a 600-gallon grease lubricant steel permanent tank;
- AST 3 is a 250-gallon hydraulic oil steel permanent tank;
- AST 4 is a 250-gallon gear oil steel permeant tank;
- AST 5 is a 250-gallon motor oil steel permanent tank;
- AST 6 is a 250-gallon transmission oil steel permanent tank;
- AST 7 is a 500-gallon used oil steel permanent tank;
- AST 8 is a 500-gallon gasoline steel permanent tank;
- AST 9 is a planned 500-gallon future steel permanent tank with unknown contents at this time but will contain petroleum product;
- AST 10 is a planned future steel permanent tank with unknown storage volume or contents at this time but will contain petroleum product with a volume not to exceed 5,000 gallons;
- AST 11 is a 450-gallon diesel exhaust fluid additive steel permanent tank; the contents of this tank are not regulated under the AST Plan requirements and therefore are not included in the associated plan fee; the displacement volume for this tank is accounted for in the available containment structure volume;
- AST 12 is a 250-gallon antifreeze steel permanent tank; the contents of this tank are not regulated under the AST Plan requirements and therefore are not included in the associated plan fee; the displacement volume for this tank is accounted for in the available containment structure volume; and
- Eighteen (18) 55-gallon portable steel drums containing various petroleum products; the drums are not subject to associated plan fees but because the drums contain regulated contents, the volume is accounted for in required containment volume calculation at one and one-half times the total storage volume.

The Tank Storage Area consists of operation of the proposed ASTs within an impervious concrete dike secondary containment structure, designed to have an available storage volume sufficient to contain one and one-half times the cumulative storage capacity of the regulated tanks and drums plus the displacement volume of unregulated tanks stored within the containment structure. This AST Plan includes regulated tanks and drums containing a cumulative storage volume of up to 19,090 gallons (note: the capacity of AST 10 is unknown at this time but will not exceed 5,000 gallons as proposed in this plan). This total storage volume of the regulated tanks AST 11 and AST 12 that are stored in the Tank Storage Area containment structure equals 29,335 gallons (19,090-gal times 1.5 plus 450-gal plus 250-gal = 29,335 gal).

The secondary containment structure at Tank Storage Area Project Site includes a concrete slab sufficiently impervious to petroleum products (inner dimensions approximately 38.5 feet by 35.2 ft), concrete dike (approximately 3 feet in height), and a roof system. The secondary containment structure provides a total containment volume of 30,413 gallons which exceeds the required containment volume of 29,335 gallons.

In addition to the Scaled Drawings of the Tank Storage Area, the Site Plan shows the Tank Storage Area and surrounding areas, including the 100-year floodplain boundary. The Tank Storage Area is not located within the 100-year floodplain. Wells and geologic or manmade features identified in the approved Geologic Assessment are shown and labeled on the Site Plan. A total of two wells (the other six borings are not in use and have been properly abandoned) are present on the Rio Medina Operation Site, and the nearest well (S-13-MB) is greater than 300 feet to the east of the Tank Storage Area Project Site. The nearest geologic or manmade feature (S-10-MB) is not characterized as a sensitive geologic feature and is located approximately 100 feet to the west of the Tank Storage Area Project Site.







	PROPERTY BOUNDARY		MANMADE POND
	EXISTING GROUND ELEVATION CONTOUR (FT-MSL)		VEGETATED BUFFER
			EARTHEN BERM
(EXISTING FENCE		QUARRY LIMITS
2	FLOWLINE	>>>	ROADSIDE DITCH
3	WATER WELL		
D	GEOLOGIC FEATURE		
	SOIL BORING		
	100-YEAR FLOODPLAIN (NOTE 1)		

		7	8
ΞΙ	NERAL CO	DNSTRUCTION NOTES:	
	WRITTEI LATER T THE DAT REGULA THE COI	N CONSTRUCTION NOTIFICATION MUST BE GIVEN TO TH HAN 48 HOURS PRIOR TO COMMENCEMENT OF THE REC TE ON WHICH THE REGULATED ACTIVITY WILL COMMENC TED ACTIVITY, AND THE NAME OF THE PRIME CONTRAC NTACT PERSON.	E APPROPRIATE TCEQ REGIONAL OFFICE NO GULATED ACTIVITY. INFORMATION MUST INCLUDE E, THE NAME OF THE APPROVED PLAN FOR THE TOR AND THE NAME AND TELEPHONE NUMBER OF
	ALL CON WITH CO INDICAT ACTIVITI APPROV	NTRACTORS CONDUCTING REGULATED ACTIVITIES ASSO DMPLETE COPIES OF THE APPROVED WATER POLLUTION ING THE SPECIFIC CONDITIONS OF ITS APPROVAL. DUR IES, THE CONTRACTORS ARE REQUIRED TO KEEP ON-SI (AL LETTER.	CIATED WITH THIS PROJECT MUST BE PROVIDED I ABATEMENT PLAN AND THE TCEQ LETTER ING THE COURSE OF THESE REGULATED TE COPIES OF THE APPROVED PLAN AND
	IF ANY S SENSITI IMMEDIA ACTIVITI THE MET POTENT	ENSITIVE FEATURE IS DISCOVERED DURING CONSTRUCT VE FEATURE MUST BE SUSPENDED IMMEDIATELY. THE A ATELY NOTIFIED OF ANY SENSITIVE FEATURES ENCOUNT IES NEAR THE SENSITIVE FEATURE MAY NOT PROCEED THODS PROPOSED TO PROTECT THE SENSITIVE FEATUR IALLY ADVERSE IMPACTS TO WATER QUALITY.	TION, ALL REGULATED ACTIVITIES NEAR THE APPROPRIATE TCEQ REGIONAL OFFICE MUST BE TERED DURING CONSTRUCTION. THE REGULATED UNTIL THE TCEQ HAS REVIEWED AND APPROVED RE AND THE EDWARDS AQUIFER FROM ANY
	NO TEMI BE INST OTHER S	PORARY ABOVEGROUND HYDROCARBON AND HAZARDO ALLED WITHIN 150 FEET OF A DOMESTIC, INDUSTRIAL, IR SENSITIVE FEATURE.	OUS SUBSTANCE STORAGE TANK SYSTEM SHALL RIGATION, OR PUBLIC WATER SUPPLY WELL, OR
	PRIOR T MEASUF MANUFA TEMPOF DURING	O COMMENCEMENT OF CONSTRUCTION, ALL TEMPORA RES MUST BE PROPERLY SELECTED, INSTALLED, AND MA ACTURERS SPECIFICATIONS AND GOOD ENGINEERING P RARY STORM WATER SECTION OF THE APPROVED EDWA CONSTRUCTION. IF INSPECTIONS INDICATE A CONTROL	RY EROSION AND SEDIMENTATION (E&S) CONTROL AINTAINED IN ACCORDANCE WITH THE RACTICES. CONTROLS SPECIFIED IN THE RDS AQUIFER PROTECTION PLAN ARE REQUIRED L HAS BEEN USED INAPPROPRIATELY, OR

WHEN THE SEDIMENT OCCUPIES 50% OF THE BASIN VOLUME. 8. LITTER, CONSTRUCTION DEBRIS, AND CONSTRUCTION CHEMICALS EXPOSED TO STORMWATER SHALL BE PREVENTED FROM BECOMING A POLLUTANT SOURCE FOR STORMWATER DISCHARGES (E.G., SCREENING OUTFALLS, PICKED UP DAILY).

INCORRECTLY, THE APPLICANT MUST REPLACE OR MODIFY THE CONTROL FOR SITE SITUATIONS. THE CONTROLS

AT A FREQUENCY SUFFICIENT TO MINIMIZE OFFSITE IMPACTS TO WATER QUALITY (E.G., FUGITIVE SEDIMENT IN

DESIGN CAPACITY HAS BEEN REDUCED BY 50%. A PERMANENT STAKE MUST BE PROVIDED THAT CAN INDICATE

MUST REMAIN IN PLACE UNTIL DISTURBED AREAS ARE REVEGETATED AND THE AREAS HAVE BECOME

STREET BEING WASHED INTO SURFACE STREAMS OR SENSITIVE FEATURES BY THE NEXT RAIN).

9. ALL SPOILS (EXCAVATED MATERIAL) GENERATED FROM THE PROJECT SITE MUST BE STORED ON-SITE WITH PROPER E&S CONTROLS. FOR STORAGE OR DISPOSAL OF SPOILS AT ANOTHER SITE ON THE EDWARDS AQUIFER RECHARGE ZONE, THE OWNER OF THE SITE MUST RECEIVE APPROVAL OF A WATER POLLUTION ABATEMENT PLAN FOR THE PLACEMENT OF FILL MATERIAL OR MASS GRADING PRIOR TO THE PLACEMENT OF SPOILS AT THE OTHER SITE.

10. STABILIZATION MEASURES SHALL BE INITIATED AS SOON AS PRACTICABLE IN PORTIONS OF THE SITE WHERE CONSTRUCTION ACTIVITIES HAVE TEMPORARILY OR PERMANENTLY CEASED, BUT IN NO CASE MORE THAN 14 DAYS AFTER THE CONSTRUCTION ACTIVITY IN THAT PORTION OF THE SITE HAS TEMPORARILY OR PERMANENTLY CEASED. WHERE THE INITIATION OF STABILIZATION MEASURES BY THE 14TH DAY AFTER CONSTRUCTION ACTIVITY TEMPORARY OR PERMANENTLY CEASE IS PRECLUDED BY WEATHER CONDITIONS, STABILIZATION MEASURES SHALL BE INITIATED AS SOON AS PRACTICABLE. WHERE CONSTRUCTION ACTIVITY ON A PORTION OF THE SITE IS TEMPORARILY CEASED, AND EARTH DISTURBING ACTIVITIES WILL BE RESUMED WITHIN 21 DAYS, TEMPORARY STABILIZATION MEASURES DO NOT HAVE TO BE INITIATED ON THAT PORTION OF SITE. IN AREAS EXPERIENCING DROUGHTS WHERE THE INITIATION OF STABILIZATION MEASURES BY THE 14TH DAY AFTER CONSTRUCTION ACTIVITY HAS TEMPORARILY OR PERMANENTLY CEASED IS PRECLUDED BY SEASONAL ARID CONDITIONS, STABILIZATION MEASURES SHALL BE INITIATED AS SOON AS PRACTICABLE.

11. THE FOLLOWING RECORDS SHALL BE MAINTAINED AND MADE AVAILABLE TO THE TCEQ UPON REQUEST: THE DATES WHEN MAJOR GRADING ACTIVITIES OCCUR; THE DATES WHEN CONSTRUCTION ACTIVITIES TEMPORARILY OR PERMANENTLY CEASE ON A PORTION OF THE SITE: AND THE DATES WHEN STABILIZATION MEASURES ARE INITIATED.

12. THE HOLDER OF ANY APPROVED EDWARD AQUIFER PROTECTION PLAN MUST NOTIFY THE APPROPRIATE REGIONAL OFFICE IN WRITING AND OBTAIN APPROVAL FROM THE EXECUTIVE DIRECTOR PRIOR TO INITIATING ANY OF THE FOLLOWING:

- 12.A. ANY PHYSICAL OR OPERATIONAL MODIFICATION OF ANY WATER POLLUTION ABATEMENT STRUCTURE(S), INCLUDING BUT NOT LIMITED TO PONDS, DAMS, BERMS, SEWAGE TREATMENT PLANTS, AND DIVERSIONARY STRUCTURES;
- 12.B. ANY CHANGE IN THE NATURE OR CHARACTER OF THE REGULATED ACTIVITY FROM THAT WHICH WAS ORIGINALLY APPROVED OR A CHANGE WHICH WOULD SIGNIFICANTLY IMPACT THE ABILITY OF THE PLAN TO PREVENT POLLUTION OF THE EDWARDS AQUIFER;

12.C. ANY DEVELOPMENT OF LAND PREVIOUSLY IDENTIFIED AS UNDEVELOPED IN THE ORIGINAL WATER POLLUTION ABATEMENT PLAN.

SAN ANTONIO REGIONAL OFFICE 14250 JUDSON ROAD SAN ANTONIO, TEXAS 78233-4480 PHONE: (210) 490-3096 FAX: (210) 545-4329

REV	DATE	

TITLE:

PERMANENTLY STABILIZED.

DESCRIPTION

APP DRN GEOSYNTEC CONSULTANTS, INC. 8217 SHOAL CREEK BLVD, SUITE 200

consultants

Geosyntec[▷]

TEXAS ENG. FIRM REGISTRATION NO. 1182 AUSTIN, TEXAS 78757 PHONE: 512.451.4003

SITE PLAN

PROJECT: CAPITOL AGGREGATES RIO MEDINA OPERATION					
SITE: MEDINA COUNTY, TX					
THIS DRAWING MAY NOT BE ISSUED	STEOF TEN	DESIGN BY:	OAB	DATE:	JUNE 2023
CONSTRUCTION, UNLESS SEALED.		DRAWN BY:	MDN	PROJECT NO.:	TXW9681
Brandon Kleyender	J. BRANDON KLENZENDORF	CHECKED BY:	JBK	FORM:	
SIGNATURE		REVIEWED BY:	JBK	DRAWING NO .:	
6/14/2023 DATE	Contraction of the second seco	APPROVED BY:	JBK	1	OF1
	7			8	

Attachment C – Exception to the Geologic Assessment

Not applicable. All geologic or manmade features identified in the Geologic Assessment (included within this AST Plan application) are shown and labeled in the vicinity of the Tank Storage Area Project Site. Additional geologic or manmade features were identified on-site but are not shown on the Site Plan due to their distance from the Tank Storage Area Project Site. None of the identified features located near the Tank Storage Area Project Site were classified as sensitive. Therefore, an Exception to the Geologic Assessment is not required. Additional geologic or manmade features at the Rio Medina Operation Site are identified in the approved Geologic Assessment.

Attachment D – Spill and Overfill Control

The objective of this section is to describe measures to prevent or reduce the discharge of pollutants to drainage systems or water courses from leaks, spills, and overfills by reducing their probability, training employees, and using BMPs to mitigate releases.

Employee Education

Measures must be implemented to prevent or reduce the discharge of pollutants to drainage systems or watercourses from leaks, spills, and overfills. Employee training programs must include the following, at a minimum:

- awareness that different materials pollute in different amounts and the definition of a "significant spill" for each material used;
- awareness of the appropriate response for "significant" and "insignificant" spills and when a spill must be reported to the TCEQ;
- potential dangers to humans and the environment from spills and leaks;
- hold regular meetings to discuss and reinforce appropriate disposal procedures (incorporate into routine safety meetings);
- establish a continuing education program to indoctrinate new employees; and
- have representatives oversee and enforce proper spill prevention and control measures.

General Measures

The following general measures for spill response actions shall be implemented at the Site:

- To the extent that the work can be accomplished safely, contain and immediately clean up spills of oil, petroleum products, substances listed under 40 CFR parts 110, 117 and 302, and sanitary and septic wastes.
- Store hazardous materials and wastes in covered containers and protect from vandalism.
- Place a stockpile of spill cleanup materials where it will be readily accessible.
- Train employees in spill prevention and cleanup.
- Designate responsible individuals to oversee and enforce control measures.
- Cover spills and protect them from stormwater run-on during rainfall to the extent that it does not compromise cleanup activities.
- Do not bury or wash spills with water.
- Store and dispose of used clean up materials, contaminated materials, and recovered spill material that is no longer suitable for the intended purpose in conformance with the provisions in applicable BMPs.
- Do not allow water used for cleaning and decontamination to enter storm drains or watercourses. Collect and dispose of contaminated water in accordance with applicable regulations.
- Contain water overflow or minor water spillage and do not allow it to discharge into drainage facilities or watercourses.
- Place Safety Data Sheets (SDSs), as well as proper storage, cleanup, and spill reporting instructions for hazardous materials stored or used on the project site in an open, conspicuous, and accessible location.
- Keep waste storage areas clean, well-organized, and equipped with ample cleanup supplies as appropriate for the materials being stored. Perimeter controls, containment structures, covers, and liners should be repaired or replaced as needed to maintain proper function.

Refilling Operations

This procedure provides guidelines to be used in the safe and environmentally responsible refueling and maintenance of vehicles and equipment with petroleum products stored on-site, refilling of the diesel storage tank, or any other reasonably sized transfer of fuel at the Site. This procedure encompasses those activities involving the transfer of reasonable quantities of fuel at the Site in a manner such that risks to health, safety, and the environment are minimized as much as is practicable.

• Operating personnel must be present at all times when large quantity diesel/fuel transfer is taking place.

- Observe tank capacity and contents and ensure that the volumes being added to the bulk tank are not excessive and that the correct material is being added.
- Make certain that a Safety Data Sheet (SDS) is available.
- Ensure that all pipes and hoses are properly connected and can be observed during any transfer of fuel. Both ends of hoses should be visible and should not be used in unlighted areas at night.
- Capitol Aggregates must approve all locations where refueling or transfer is taking place.
- Portable containment and spill control devices (such as buckets, drip pans, etc.) should be available, should a leak occur, and ready for use.
- Vehicle parking brake should be set prior to any refueling activity.
- Should leaks or spills occur, any transfer must immediately be stopped and the cause of the leak or spill investigated and resolved.
- All pipes and hoses must be voided of contents prior to disassembly and removal from coupling devices.
- Vehicle engines should be shut off during refueling or transfer.

Attachment E – Response Actions to Spills

The objective of this section is to describe the planned response actions to spills that could take place at the facility.

Cleanup

A spill containment kit and cleanup materials shall be stored where they are readily accessible. The Site Spill Coordinator shall be designated to oversee and enforce control measures and maintain and restock the spill containment kit, as needed. Clean up leaks and spills immediately. Use a rag for small spills on paved surfaces, a damp mop for general cleanup, and absorbent materials for larger spills. If the spilled material is hazardous, then the used cleanup materials are also hazardous and must be disposed of as hazardous waste. Never hose down or bury dry material spills. Clean up as much of the material as possible and dispose of properly.

Minor Spills

Minor spills typically involve small quantities of oil, gasoline, paint, etc. which can be controlled by the first responder at the discovery of the spill. Use absorbent materials on small spills rather than washing the area with water or burying the spill. Absorbent materials should be promptly removed and disposed of properly. Follow the practice below for a minor spill:

- Contain the spread of the spill.
- Recover spill materials
- Clean the contaminated area and properly dispose of contaminated materials.

Semi-Significant Spills

Semi-significant spills that are below reportable quantities, are not hazardous, and can be controlled by the first responder along with the aid of other personnel such as laborers and the foreman, etc. This response may require the cessation of all other activities in order to contain the spill. Semi-significant spills should be cleaned up immediately using the following practices:

- Contain the spread of the spill.
- Notify the spill cleanup coordinate or project foreman immediately.
- If the spill occurs on paved or impermeable surfaces, clean up using dry methods (absorbent materials, cat litter and/or rags provided in the spill containment kit). Contain the spill by encircling it with absorbent materials to prevent the spread of the material.
- If the spill occurs in dirt areas, the spill can be contained by constructing an earthen berm; spilled material can be excavated and properly dispose of.
- If the spill occurs during rain, cover spill with tarps or other material to prevent contaminating runoff.

Significant or Hazardous Spills

For significant or hazardous spills, reportable quantities include the following limits:

- For spills of hazardous substances, the federal requirements in 40 CFR 302.4 apply.
- For petroleum products or used oil, the reportable quantity for a spill onto land is 25 gallons, and the reportable quantity directly into water is enough to create a sheen on the water.

In the event of a significant or hazardous spill in excess of reportable quantities, Site personnel shall:

• Notify the TCEQ San Antonio Regional Office by telephone as soon as possible and within 24 hours at (210) 490-3096 between 8 AM and 5 PM. After hours, contact the State of Texas Spill-Reporting Hotline and the State Emergency Response Commission (SERC) at 1-800-832-8224. All emergency phone numbers must be posted at the Site.

- For spills of federal reportable quantities, in conformance with the requirements in 40 CFR parts 110, 119, and 302, notify the National Response Center at 1-800-424-8802. Notifying the National Response Center does not constitute notice to TCEQ.
- Notification should first be made by telephone and followed up with a written report. Spills reported to the National Response Center can be reported via an online reporting tool.
- The services of a spills contractor or a Haz-Mat team should be obtained immediately. Construction personnel should not attempt to clean up a spill until the appropriate and qualified staff have arrived at the job site.
- Other agencies which may need to be notified include, but are not limited to, City Police Department, County Sheriff Office, Fire Departments, etc.

In the event of a reportable spill, the following Emergency Response Agencies can be contacted for assistance. Always inform your supervisor of a reportable spill immediately. Follow company policy when responding to an emergency.

TCEQ Region 13 office (Includes Medina County) State Emergency Response Center (Spill Reporting)	(210) 490-3096 (800) 832-8224
TCEQ Main Office	(512) 239-3700
National Response Center	(800) 424-8802
Castroville Fire Department	(830) 931-4751 911
Castroville Police Department	(830) 931-2222 911
Medina County Sherriff's Office	(830) 741-6150
USEPA, Region VI	(214) 665-2200 (800) 887-6063
Medina County Emergency Management Coordinator	(330) 722-9240
Medical Clinic of Hondo	(830) 426-7444

Vehicle or Equipment Fueling and Maintenance

Fuel or hazardous substances will be stored on-site in the ASTs described in this application. Temporary ASTs with a capacity of 250 gallons or more will not be required during project activities. In the event that fuel trucks are used for on-site vehicle fueling, best management practices will be implemented during fueling and maintenance at the site include the following:

- Inspect on-site vehicles and equipment daily for leaks and conduct repairs immediately.
- Check incoming vehicles and equipment (including delivery trucks, and employee and subcontractor vehicles) for leaking oil and fluids. Do not allow leaking vehicles or equipment on-site.
- Always use secondary containment, such as a drip pan or drop cloth, to catch spills or leaks when removing or changing fluids. Drip pans are to be used where leaks or spills of fuel can occur and where making and breaking hose connections.
- Promptly transfer used fluids to the proper waste or recycling drums; do not leave full drip pans or other open containers lying around.
- Use absorbent materials on small spills rather than hosing down or burying the spill. Remove the absorbent materials promptly and dispose of properly.
- Fueling hoses should have check valves to prevent hose drainage after filling.
- Spills and vehicle leaks are to be cleaned up with absorbent materials for dry cleanup methods. The use of water for such cleanups is to be minimized.
- "Topping off" of fuel tanks is discouraged.
- Personnel are not to leave fueling equipment unattended during fueling operations.
- Oil filters disposed of in trashcans or dumpsters can leak oil and pollute stormwater. Place the oil filter in a funnel over a waste oil-recycling drum to drain excess oil before disposal. Oil filters can also be recycled. Ask the oil supplier or recycler about recycling oil filters.
- Store cracked batteries in a non-leaking secondary container. Do this with all cracked batteries even if you think all the acid has drained out. If you drop a battery, treat it as if it is cracked. Put it into the containment area until it is verified as not leaking.
- Equipment such as valves, pumps, flanges, and connections shall be checked regularly for leaks and be repaired as needed.

In the event of a leak from the ASTs and associated piping, hoses, and dispensers described in this application into the secondary containment structures, spillage will be removed from the containment structure within 24 hours of the spill and disposed of properly. All stormwater accumulating inside the containment structure will be disposed of through an authorized waste disposal contractor.

Administrative Information

The original WPAP approval letter (dated 22 June 2016) and WPAP modification approval letter (dated 24 August 2018) are attached.

Bryan W. Shaw, Ph.D., P.E., *Chairman* Toby Baker, *Commissioner* Jon Niermann, *Commissioner* Richard A. Hyde, P.E., *Executive Director*



TEXAS COMMISSION ON ENVIRONMENTAL QUALITY

Protecting Texas by Reducing and Preventing Pollution

June 22, 2016

Mr. Paul Detterline Capitol Aggregates, Inc. 11551 Nacogdoches Road San Antonio, Texas 78217

Re: Edwards Aquifer, Medina County

NAME OF PROJECT: Capitol Aggregates Rio Medina Operation; Located at 1576 County Road 265; Hondo, Texas

PLAN TYPE: Request for the Approval of a Water Pollution Abatement Plan (WPAP); 30 Texas Administrative Code (TAC) Chapter 213 Edwards Aquifer

Regulated Entity No. RN108959032; Additional ID No. 13000073

Dear Mr. Detterline:

The Texas Commission on Environmental Quality (TCEQ) has completed its review of the WPAP application for the above-referenced project submitted to the San Antonio Regional Office by Geosyntec Consultants on behalf of Capitol Aggregates, Inc. on January 22, 2016. Final review of the WPAP was completed after additional material was received on March 18, 2016, April 18, 2016, and June 6, 2016. As presented to the TCEQ, the Temporary and Permanent Best Management Practices (BMPs) were selected and construction plans were prepared by a Texas Licensed Professional Engineer to be in general compliance with the requirements of 30 TAC Chapter 213. These planning materials were sealed, signed and dated by a Texas Licensed Professional Engineer. Therefore, based on the engineer's concurrence of compliance, the planning materials for construction of the proposed project and pollution abatement measures are hereby approved subject to applicable state rules and the conditions in this letter. The applicant or a person affected may file with the chief clerk a motion for reconsideration of the executive director's final action on this Edwards Aquifer Protection Plan. A motion for reconsideration must be filed no later than 23 days after the date of this approval letter. *This approval expires two (2) years from the date of this letter unless, prior to the expiration date, more than 10 percent of the construction has commenced on the project or an extension of time has been requested.*

PROJECT DESCRIPTION

The proposed limestone quarry project will have an area of approximately 503 acres. The proposed quarry pit will disturb approximately 356 acres. The proposed activities for the site include quarrying to an elevation no deeper than 872.6 feet above mean sea level (a.m.s.l.). The proposed impervious cover is 24.5 (4.9 percent) acres consisting of a scale house, office, gravel parking areas, fueling and maintenance area, gravel haul roads, and rock crushing equipment. No on-site sewage facility is proposed at this time. Project wastewater (domestic) will be collected in portable toilets and disposed of by a TCEQ registered waste disposal service. Trash generated onsite will be disposed of in a dumpster and handled by a

TCEQ Region 13 • 14250 Judson Rd. • San Antonio, Texas 78233-4480 • 210-490-3096 • Fax 210-545-4329

licensed waste service. A tributary to Elm Creek runs along the north portion of the proposed quarry and a stream crossing will be utilized to access the northwest corner of the property.

PERMANENT POLLUTION ABATEMENT MEASURES

To prevent the pollution of stormwater runoff originating onsite or upgradient of the site and potentially flowing across and off the site, the specific controls described below will be utilized.

An extended detention basin in series with a grassy swale and three (3) natural vegetative filter strips designed using the TCEQ technical guidance document, <u>Complying with the Edwards Aquifer Rules:</u> <u>Technical Guidance on Best Management Practices (2005)</u>, will be utilized to treat stormwater runoff. The required total suspended solids (TSS) treatment for this project is 18,659 pounds of TSS generated from the 24.50 acres of impervious cover. The approved measures meet the required 80 percent removal of the increased load in TSS caused by the project.

The extended detention basin in series with the grassy swale will treat 16,755 pounds of TSS generated from 22.0 acres of impervious cover. The basin capture volume is 190,032 cubic feet and the grassy swale is minimum length 150 feet, maximum bottom width of 16 feet, side slopes of 3:1 (H:V), slope 1.65 percent, and 80 percent vegetated cover.

The three natural VFS will treat 1,904 pounds of TSS generated from 2.5 acres of impervious cover. The 50 foot undisturbed natural vegetative strips shall not exceed slopes of 10 percent and shall have no flow concentrating areas on the strip. The contributing area will be limited to 72 feet through the use of earthen berms on the upgradient side of the impervious cover.

A minimum 100 foot natural buffer will be maintained along the perimeter of the property and the tributary to Elm Creek to reduce soil erosion and runoff velocities. Expansion of the quarry will occur in 10 acre increments. Expansion in phases will allow vegetation to remain in place and limit the amount of soil that is disturbed.

Earthen berms (safety berm) composed of compacted soil and/or overburden will be constructed during clearing and grubbing activities. Portions of these earthen berms will be permanent near the property boundary and adjacent to the floodplain to divert runoff from upgradient areas.

Process water will be generated from the rock crushing equipment and will be treated onsite for reuse. Initial rock crushing activity will consist of a portable rock crusher, thickener, a sedimentation pond, and steel storage tanks to manage the process water generated. Process water will be sent to the thickener, in which solids are separated through the use of a flocculating agent. Solids will be stored in a lined sedimentation pond and the clean water will be pumped to the steel storage tanks for reuse. The sediment will be dried and used as fill material onsite.

As the quarry expands, the rock crushing activity will be relocated into the quarry pit and will be replaced with a permanent rock crushing plant, high rate thickener, lined sedimentation pond, and lined fresh water pond. The ponds will both be sized to contain the 10-year, 24-hour event with 1 foot of freeboard. The freshwater pond will be approximately 6.5 feet deep and the sedimentation pond will be approximately 4.0 feet deep. The ponds will be constructed with a bottom elevation greater than the minimum quarry pit elevation of 872.6 ft-msl. Both ponds will have 3:1 horizontal to vertical side slopes and will be implemented with a vegetated earthen berm built up around the perimeters. A liner will be provided for both ponds meeting clay liner specifications or geotextile fabric specifications.

Stockpiles will be located outside the quarry pit during initial development of the site. As quarry operations continue and the quarry pit expands, support activities may be relocated within the quarry pit.

A containment area for 150% of the mobile generator fuel volume or a double walled mobile generator tank will be provided.

A stream crossing will be located across a tributary to Elm Creek which runs along the north portion of the proposed quarry. Natural vegetated buffers will be maintained adjacent to the floodplain. The stream

crossing will be paved and include culverts that can convey the 2 year storm without overtopping the road. There is no proposed mining within the floodplain of the tributary to Elm Creek.

GEOLOGY

According to the geologic assessment included with the application, the site is located on the lower Devil Rivers Formation. Ten (10) geologic features and five (5) manmade features were evaluated by the project geologist. No features were rated as sensitive. Features located within the tributary to Elm Creek S-2-O, S-3-O, S-4-O, S-5-O, S-6-O, S-7-O, and S-8-MB will be protected by the 100 foot natural buffer proposed for the tributary to Elm Creek. No regulated activities (such as construction or soil disturbing activities) will take place within the natural buffers. The San Antonio Regional Office site assessment conducted on May 13, 2016 revealed that the site was generally as described in the application.

SPECIAL CONDITIONS

- I. The BMPs and measures proposed in the application and/or described in this approval letter must be operational prior to any soil disturbing activities within a BMP's drainage area.
- II. An Aboveground Storage Tank (AST) facility plan will be submitted and approved prior to the installation of ASTs for storing fuel and/or other hydrocarbons or hazardous substances, as required, as the quarry operations progress.
- III. This letter does not authorize construction of the future ready mix concrete plant or the hot mix asphalt plant.
- The existing conditions of the tributary to Elm Creek are included with the application. The IV. assessment includes photographic (taken May 13, 2016 and June 5, 2016) and narrative documentation that will enable future comparisons for the purposes of determining impact from sediment accumulation. The plan holder must install erosion and sediment controls that have been designed to retain sediment on-site to the extent practicable with consideration for local topography and rainfall. Discharges that would cause or contribute to a violation of water quality standards, or would fail to protect and maintain existing designated areas of receiving waters are not allowed. Routine inspections must be performed following rain events to determine if the tributary to Elm Creek has accumulated any sediment from the quarrying activity. Accumulations of sediment must be removed before the next rain event and may require coordination with other governmental authorities. Records of inspection, maintenance, and repairs of the crossing's control measures must include the date of the inspection, date of regular maintenance, date(s) of discovery of areas in need of sediment removal, and date(s) that the control measure(s) were returned to full function. Those records must be maintained on site and be available for review by TCEQ.
- V. Intentional discharges of sediment laden water from regulated activities are not allowed. If dewatering becomes necessary, appropriate measures must be taken.
- VI. Pursuant to 30 TAC §213.4(h)(3) and as stated in the Edwards Aquifer protection plan, this protection plan approval or extension will expire and no extension will be granted if more than 50% of the total construction has not been completed within 10 years from the initial approval of the plan. A new Edwards Aquifer protection plan must be submitted to the TCEQ with the appropriate fees for review and approval by the executive director prior to commencing or continuing any construction or regulated activities beyond 10 years. The Applicant must submit a status report for the project containing information regarding the percentage of the total project construction cannot be demonstrated to be at least 50% complete, the Applicant must submit a new Edwards Aquifer protection plan to the TCEQ for review and

approval before continuing any construction or regulated activities beyond 10 years from the date of initial approval of the plan.

STANDARD CONDITIONS

- 1. Pursuant to Chapter 7 Subchapter C of the Texas Water Code, any violations of the requirements in 30 TAC Chapter 213 may result in administrative penalties.
- 2. The holder of the approved Edwards Aquifer protection plan must comply with all provisions of 30 TAC Chapter 213 and all best management practices and measures contained in the approved plan. Additional and separate approvals, permits, registrations and/or authorizations from other TCEQ Programs (i.e., Stormwater, Water Rights, UIC) can be required depending on the specifics of the plan.
- 3. In addition to the rules of the Commission, the applicant may also be required to comply with state and local ordinances and regulations providing for the protection of water quality.

Prior to Commencement of Construction:

- 4. Within 60 days of receiving written approval of an Edwards Aquifer Protection Plan, the applicant must submit to the San Antonio Regional Office, proof of recordation of notice in the county deed records, with the volume and page number(s) of the county deed records of the county in which the property is located. A description of the property boundaries shall be included in the deed recordation in the county deed records. A suggested form (Deed Recordation Affidavit, TCEQ-0625) that you may use to deed record the approved WPAP is enclosed.
- 5. All contractors conducting regulated activities at the referenced project location shall be provided a copy of this notice of approval. At least one complete copy of the approved WPAP and this notice of approval shall be maintained at the project location until all regulated activities are completed.
- 6. Modification to the activities described in the referenced WPAP application following the date of approval may require the submittal of a plan to modify this approval, including the payment of appropriate fees and all information necessary for its review and approval prior to initiating construction of the modifications.
- 7. The applicant must provide written notification of intent to commence construction, replacement, or rehabilitation of the referenced project. Notification must be submitted to the San Antonio Regional Office no later than 48 hours prior to commencement of the regulated activity. Written notification must include the date on which the regulated activity will commence, the name of the approved plan and program ID number for the regulated activity, and the name of the prime contractor with the name and telephone number of the contact person. The executive director will use the notification to determine if the approved plan is eligible for an extension.
- 8. Temporary erosion and sedimentation (E&S) controls, i.e., silt fences, rock berms, stabilized construction entrances, or other controls described in the approved WPAP, must be installed prior to construction and maintained during construction. Temporary E&S controls may be removed when vegetation is established and the construction area is stabilized. If a water quality pond is proposed, it shall be used as a sedimentation basin during construction. The TCEQ may monitor stormwater discharges from the site to evaluate the adequacy of temporary E&S control measures. Additional controls may be necessary if excessive solids are being discharged from the site.
- 9. All borings with depths greater than or equal to 20 feet must be plugged with non-shrink grout from the bottom of the hole to within three (3) feet of the surface. The remainder of the hole must be backfilled with cuttings from the boring. All borings less than 20 feet must be backfilled with cuttings from the boring. All borings with a plugged within four (4) days of completion of the drilling operation. Voids may be filled with gravel.

During Construction:

- 10. During the course of regulated activities related to this project, the applicant or agent shall comply with all applicable provisions of 30 TAC Chapter 213, Edwards Aquifer. The applicant shall remain responsible for the provisions and conditions of this approval until such responsibility is legally transferred to another person or entity.
- 11. This approval does not authorize the installation of temporary aboveground storage tanks on this project. If the contractor desires to install a temporary aboveground storage tank for use during construction, an application to modify this approval must be submitted and approved prior to installation. The application must include information related to tank location and spill containment. Refer to Standard Condition No. 6, above.
- 12. If any sensitive feature (caves, solution cavities, sink holes, etc.) is discovered during construction, all regulated activities near the feature must be suspended immediately. The applicant or his agent must immediately notify the San Antonio Regional Office of the discovery of the feature. Regulated activities near the feature may not proceed until the executive director has reviewed and approved the methods proposed to protect the feature and the aquifer from potentially adverse impacts to water quality. The plan must be sealed, signed, and dated by a Texas Licensed Professional Engineer.
- 13. Two wells and six borings exists on site. All water wells, including injection, dewatering, and monitoring wells must be in compliance with the requirements of the Texas Department of Licensing and Regulation under Title 16 TAC Chapter 76 (relating to Water Well Drillers and Pump Installers) and all other locally applicable rules, as appropriate.
- 14. If sediment escapes the construction site, the sediment must be removed at a frequency sufficient to minimize offsite impacts to water quality (e.g., fugitive sediment in street being washed into surface streams or sensitive features by the next rain). Sediment must be removed from sediment traps or sedimentation ponds not later than when design capacity has been reduced by 50 percent. Litter, construction debris, and construction chemicals shall be prevented from becoming stormwater discharge pollutants.
- 15. Intentional discharges of sediment laden water are not allowed. If dewatering becomes necessary, the discharge will be filtered through appropriately selected best management practices. These may include vegetated filter strips, sediment traps, rock berms, silt fence rings, etc.
- 16. The following records shall be maintained and made available to the executive director upon request: the dates when major grading activities occur, the dates when construction activities temporarily or permanently cease on a portion of the site, and the dates when stabilization measures are initiated.
- 17. Stabilization measures shall be initiated as soon as practicable in portions of the site where construction activities have temporarily or permanently ceased, and construction activities will not resume within 21 days. When the initiation of stabilization measures by the 14th day is precluded by weather conditions, stabilization measures shall be initiated as soon as practicable.

After Completion of Construction:

- 18. A Texas Licensed Professional Engineer must certify in writing that the permanent BMPs or measures were constructed as designed. The certification letter must be submitted to the San Antonio Regional Office within 30 days of site completion.
- 19. The applicant shall be responsible for maintaining the permanent BMPs after construction until such time as the maintenance obligation is either assumed in writing by another entity having ownership or control of the property (such as without limitation, an owner's association, a new property owner or lessee, a district, or municipality) or the ownership of the property is transferred to the entity. The regulated entity shall then be responsible for maintenance until another entity assumes such obligations in writing or ownership is transferred. A copy of the transfer of responsibility must be filed with the executive director through San Antonio Regional Office within 30 days of the transfer. A copy of the transfer form (TCEQ-10263) is enclosed.

- 20. Upon legal transfer of this property, the new owner(s) is required to comply with all terms of the approved Edwards Aquifer protection plan. If the new owner intends to commence any new regulated activity on the site, a new Edwards Aquifer protection plan that specifically addresses the new activity must be submitted to the executive director. Approval of the plan for the new regulated activity by the executive director is required prior to commencement of the new regulated activity.
- 21. An Edwards Aquifer protection plan approval or extension will expire and no extension will be granted if more than 50 percent of the total construction has not been completed within ten years from the initial approval of a plan. A new Edwards Aquifer protection plan must be submitted to the San Antonio Regional Office with the appropriate fees for review and approval by the executive director prior to commencing any additional regulated activities.
- 22. At project locations where construction is initiated and abandoned, or not completed, the site shall be returned to a condition such that the aquifer is protected from potential contamination.

This action is taken under authority delegated by the Executive Director of the Texas Commission on Environmental Quality. If you have any questions or require additional information, please contact Ms. Lillian Butler of the Edwards Aquifer Protection Program of the San Antonio Regional Office at (210) 403-4026.

Sincerely,

Lynn Bumguardner, Water Section Manager San Antonio Region Texas Commission on Environmental Quality

LB/LB/eg

Enclosure: Deed Recordation Affidavit, Form TCEQ-0625

cc: Mr. Brandon Klenzendorf, P.E., Geosyntec Consultants, Inc. The Honorable Chris Schuchart, Medina County Mr. David Caldwell, Medina County UWD Mr. Roland Ruiz, Edwards Aquifer Authority TCEQ Central Records, Building F, MC 212 Bryan W. Shaw, Ph.D., P.E., *Chairman* Jon Niermann, *Commissioner* Emily Lindley, *Commissioner* Toby Baker, *Executive Director*



TEXAS COMMISSION ON ENVIRONMENTAL QUALITY

Protecting Texas by Reducing and Preventing Pollution

August 24, 2018

Mr. Aaron Savell Century Asphalt, Ltd. P.O. BOX 187 Houston, Texas 77011

Re: Edwards Aquifer, Medina County

NAME OF PROJECT: Capitol Aggregates Rio Medina Operation; Located at 1576 County Road 265; Hondo, Texas

PLAN TYPE: Request for the Modification of the Approved Water Pollution Abatement Plan (WPAP); 30 Texas Administrative Code (TAC) Chapter 213 Edwards Aquifer

Regulated Entity No. RN108959032; Additional ID No. 13000715

Dear Mr. Savell:

The Texas Commission on Environmental Quality (TCEQ) has completed its review of the WPAP Modification application for the above-referenced project submitted to the San Antonio Regional Office by Geosyntec Consultants on behalf of Century Asphalt, Ltd. on June 28, 2018. Final review of the WPAP Modification was completed after additional material was received on August 22, 2018. As presented to the TCEQ, the Temporary and Permanent Best Management Practices (BMPs) were selected and construction plans were prepared by a Texas Licensed Professional Engineer to be in general compliance with the requirements of 30 TAC Chapter 213. These planning materials were sealed, signed and dated by a Texas Licensed Professional Engineer. Therefore, based on the engineer's concurrence of compliance, the planning materials for construction of the proposed project and pollution abatement measures are hereby approved subject to applicable state rules and the conditions in this letter. The applicant or a person affected may file with the chief clerk a motion for reconsideration of the executive director's final action on this Edwards Aquifer Protection Plan. A motion for reconsideration must be filed no later than 23 days after the date of this approval letter. This approval expires two (2) years from the date of this letter unless, prior to the expiration date, more than 10 percent of the construction has commenced on the project or an extension of time has been requested.

BACKGROUND

Capital Aggregates Rio Medina Operation WPAP (13000073) was approved June 22, 2016, for a limestone quarry project with an area of approximately 503 acres. The proposed quarry pit will disturb approximately 356 acres. The proposed activities for the site include quarrying to an elevation no deeper than 872.6 feet above mean sea level (a.m.s.l.). The proposed impervious cover is 24.5 (4.9-percent) acres consisting of a scale house, office, gravel parking areas, fueling and maintenance area, gravel haul roads, and rock crushing equipment. The approved permanent BMP(s) include an extended detention basin in series with a grassy swale and three (3) natural vegetative filter strips.

TCEQ Region 13 • 14250 Judson Rd. • San Antonio, Texas 78233-4480 • 210-490-3096 • Fax 210-545-4329

Mr. Aaron Savell August 24, 2018 Page 2

PROJECT DESCRIPTION

The proposed modification includes the installation of a hot mix asphalt plant located within 10.57 acres of the overall 503-acre site. The designated area will be exclusively for the hot mix asphalt plant and stormwater runoff from the plant area will be treated by a site specific permanent BMP, in lieu of treatment by Capital Aggregates Rio Medina Operations, as previously approved. In addition, a section of a natural vegetative filter strip previously approved to treat a gravel haul road north of the project site will be removed. This section will provide an access entrance to the hot mix asphalt plant and stormwater treatment for this access drive will be provided by the proposed new PBMP. Vegetated earthen berms will be utilized to route stormwater runoff towards the PBMP. The new impervious cover within the 10.57-acre site will be 9.0 acres (85.15-percent). The overall 503-acre site will result in 33.5 acres of impervious cover. No on-site sewage facility is proposed at the hot mix asphalt plant. Project wastewater (domestic) will be collected in portable toilets and disposed of by a TCEQ registered waste disposal service.

PERMANENT POLLUTION ABATEMENT MEASURES

To prevent the pollution of stormwater runoff originating on-site or upgradient of the site and potentially flowing across and off the site after construction, a wet basin, designed using the TCEQ technical guidance document, <u>Complying with the Edwards Aquifer Rules:</u> <u>Technical Guidance on Best</u> <u>Management Practices (2005)</u>, will be utilized and constructed to treat stormwater runoff. The required total suspended solids (TSS) treatment for this project is 7,197 pounds of TSS generated from the 9.45 acres of impervious cover, which includes access drive. The approved measures meet the required 80 percent removal of the increased load in TSS caused by the project.

The wet basin is an online facility that is configured with a 2-foot depth sediment forebay and 4-foot depth main pool and includes 1.0-foot freeboard emergency spillway. The permanent pool is designed with a water quality volume of 50,498 cubic feet (41,649 cubic feet required) and a designed capacity at water quality elevation of 76,357 cubic feet. The basin will be lined with a 6- inch compacted topsoil overlying a 12-inch thick compacted clay liner, side slopes are constructed at 3:1 (H:V) for grass stabilization and the outflow structure is provided to drain the water quality volume in a minimum of 24 hours. A separate drain pipe with a manual valve is not included because the basin cannot completely drain by gravity; therefore, a pump will be available onsite to completely drain the basin for maintenance purposes as needed.

<u>GEOLOGY</u>

According to the geologic assessment included with the application, the site is located on the lower Devil Rivers Formation. Ten (10) geologic features and five (5) manmade features were evaluated by the project geologist within the previously approved 503-acre area. However, no features were identified within the 10.57-acre project area by the project geologist. The San Antonio Regional Office site assessment conducted on July 19, 2018 revealed that the site was generally as described in the application.

SPECIAL CONDITIONS

I. This modification is subject to all Special and Standard Conditions listed in the WPAP approval letter dated June 22, 2016.

- II. An Aboveground Storage Tank (AST) plan must be submitted for the installation and use of fuel and oil containment required within the Hot Asphalt Plant project area. Installation and use of an AST or AST system is not authorized by this letter.
- III. All permanent pollution abatement measures shall be inspected, and all measures be fully operational prior to first use of the newly constructed facilities.
- IV. All sediment and/or media removed from the wet basin during maintenance activities shall be properly disposed of according to 30 TAC 330 or 30 TAC 335, as applicable.

STANDARD CONDITIONS

- 1. Pursuant to Chapter 7 Subchapter C of the Texas Water Code, any violations of the requirements in 30 TAC Chapter 213 may result in administrative penalties.
- 2. The holder of the approved Edwards Aquifer protection plan must comply with all provisions of 30 TAC Chapter 213 and all best management practices and measures contained in the approved plan. Additional and separate approvals, permits, registrations and/or authorizations from other TCEQ Programs (i.e., Stormwater, Water Rights, UIC) can be required depending on the specifics of the plan.
- 3. In addition to the rules of the Commission, the applicant may also be required to comply with state and local ordinances and regulations providing for the protection of water quality.

Prior to Commencement of Construction:

- 4. Within 60 days of receiving written approval of an Edwards Aquifer Protection Plan, the applicant must submit to the San Antonio Regional Office, proof of recordation of notice in the county deed records, with the volume and page number(s) of the county deed records of the county in which the property is located. A description of the property boundaries shall be included in the deed recordation in the county deed records. A suggested form (Deed Recordation Affidavit, TCEQ-0625) that you may use to deed record the approved WPAP is enclosed.
- 5. All contractors conducting regulated activities at the referenced project location shall be provided a copy of this notice of approval. At least one complete copy of the approved WPAP and this notice of approval shall be maintained at the project location until all regulated activities are completed.
- 6. Modification to the activities described in the referenced WPAP application following the date of approval may require the submittal of a plan to modify this approval, including the payment of appropriate fees and all information necessary for its review and approval prior to initiating construction of the modifications.
- 7. The applicant must provide written notification of intent to commence construction, replacement, or rehabilitation of the referenced project. Notification must be submitted to the San Antonio Regional Office no later than 48 hours prior to commencement of the regulated activity. Written notification must include the date on which the regulated activity will commence, the name of the approved plan and program ID number for the regulated activity, and the name of the prime contractor with the name and telephone number of the contact person. The executive director will use the notification to determine if the approved plan is eligible for an extension.
- 8. Temporary erosion and sedimentation (E&S) controls, i.e., silt fences, rock berms, stabilized construction entrances, or other controls described in the approved WPAP, must be installed prior to construction and maintained during construction. Temporary E&S controls may be removed when vegetation is established and the construction area is stabilized. If a water quality pond is proposed, it shall be used as a sedimentation basin during construction. The TCEQ may monitor stormwater discharges from the site to evaluate the adequacy of temporary E&S control measures. Additional controls may be necessary if excessive solids are being discharged from the site.

Mr. Aaron Savell August 24, 2018 Page 4

9. All borings with depths greater than or equal to 20 feet must be plugged with non-shrink grout from the bottom of the hole to within three (3) feet of the surface. The remainder of the hole must be backfilled with cuttings from the boring. All borings less than 20 feet must be backfilled with cuttings from the boring. All borings must be backfilled or plugged within four (4) days of completion of the drilling operation. Voids may be filled with gravel.

During Construction:

- 10. During the course of regulated activities related to this project, the applicant or agent shall comply with all applicable provisions of 30 TAC Chapter 213, Edwards Aquifer. The applicant shall remain responsible for the provisions and conditions of this approval until such responsibility is legally transferred to another person or entity.
- 11. This approval does not authorize the installation of temporary aboveground storage tanks on this project. If the contractor desires to install a temporary aboveground storage tank for use during construction, an application to modify this approval must be submitted and approved prior to installation. The application must include information related to tank location and spill containment. Refer to Standard Condition No. 6, above.
- 12. If any sensitive feature (caves, solution cavities, sink holes, etc.) is discovered during construction, all regulated activities near the feature must be suspended immediately. The applicant or his agent must immediately notify the San Antonio Regional Office of the discovery of the feature. Regulated activities near the feature may not proceed until the executive director has reviewed and approved the methods proposed to protect the feature and the aquifer from potentially adverse impacts to water quality. The plan must be sealed, signed, and dated by a Texas Licensed Professional Engineer.
- 13. Two wells and six bore holes exist on site. All water wells, including injection, dewatering, and monitoring wells must be in compliance with the requirements of the Texas Department of Licensing and Regulation under Title 16 TAC Chapter 76 (relating to Water Well Drillers and Pump Installers) and all other locally applicable rules, as appropriate.
- 14. If sediment escapes the construction site, the sediment must be removed at a frequency sufficient to minimize offsite impacts to water quality (e.g., fugitive sediment in street being washed into surface streams or sensitive features by the next rain). Sediment must be removed from sediment traps or sedimentation ponds not later than when design capacity has been reduced by 50 percent. Litter, construction debris, and construction chemicals shall be prevented from becoming stormwater discharge pollutants.
- 15. Intentional discharges of sediment laden water are not allowed. If dewatering becomes necessary, the discharge will be filtered through appropriately selected best management practices. These may include vegetated filter strips, sediment traps, rock berms, silt fence rings, etc.
- 16. The following records shall be maintained and made available to the executive director upon request: the dates when major grading activities occur, the dates when construction activities temporarily or permanently cease on a portion of the site, and the dates when stabilization measures are initiated.
- 17. Stabilization measures shall be initiated as soon as practicable in portions of the site where construction activities have temporarily or permanently ceased, and construction activities will not resume within 21 days. When the initiation of stabilization measures by the 14th day is precluded by weather conditions, stabilization measures shall be initiated as soon as practicable.

After Completion of Construction:

18. A Texas Licensed Professional Engineer must certify in writing that the permanent BMPs or measures were constructed as designed. The certification letter must be submitted to the San Antonio Regional Office within 30 days of site completion.

Mr. Aaron Savell August 24, 2018 Page 5

- 19. The applicant shall be responsible for maintaining the permanent BMPs after construction until such time as the maintenance obligation is either assumed in writing by another entity having ownership or control of the property (such as without limitation, an owner's association, a new property owner or lessee, a district, or municipality) or the ownership of the property is transferred to the entity. The regulated entity shall then be responsible for maintenance until another entity assumes such obligations in writing or ownership is transferred. A copy of the transfer of responsibility must be filed with the executive director through San Antonio Regional Office within 30 days of the transfer. A copy of the transfer form (TCEQ-10263) is enclosed.
- 20. Upon legal transfer of this property, the new owner(s) is required to comply with all terms of the approved Edwards Aquifer protection plan. If the new owner intends to commence any new regulated activity on the site, a new Edwards Aquifer protection plan that specifically addresses the new activity must be submitted to the executive director. Approval of the plan for the new regulated activity by the executive director is required prior to commencement of the new regulated activity.
- 21. An Edwards Aquifer protection plan approval or extension will expire and no extension will be granted if more than 50 percent of the total construction has not been completed within ten years from the initial approval of a plan. A new Edwards Aquifer protection plan must be submitted to the San Antonio Regional Office with the appropriate fees for review and approval by the executive director prior to commencing any additional regulated activities.
- 22. At project locations where construction is initiated and abandoned, or not completed, the site shall be returned to a condition such that the aquifer is protected from potential contamination.

This action is taken under authority delegated by the Executive Director of the Texas Commission on Environmental Quality. If you have any questions or require additional information, please contact Ms. Lillian Butler of the Edwards Aquifer Protection Program of the San Antonio Regional Office at (210) 403-4026.

Sincerely,

Lynn Bumguardner, Water Section Manager San Antonio Region Office Texas Commission on Environmental Quality

LB/LB/eg

Enclosures: Deed Recordation Affidavit, Form TCEQ-0625 Change in Responsibility for Maintenance of Permanent BMPs, Form TCEQ-10263

cc: Mr. Brandon Klenzendorf, P.E., Geosyntec Consultants, Inc. The Honorable Chris Schuchart, Medina County Mr. David Caldwell, Medina County UWD Mr. Roland Ruiz, Edwards Aquifer Authority

Temporary Stormwater Section

Texas Commission on Environmental Quality

for Regulated Activities on the Edwards Aquifer Recharge Zone and Relating to 30 TAC §213.5(b)(4)(A), (B), (D)(I) and (G); Effective June 1, 1999

To ensure that the application is administratively complete, confirm that all fields in the form are complete, verify that all requested information is provided, consistently reference the same site and contact person in all forms in the application, and ensure forms are signed by the appropriate party.

Note: Including all the information requested in the form and attachments contributes to more streamlined technical reviews.

Signature

To the best of my knowledge, the responses to this form accurately reflect all information requested concerning the proposed regulated activities and methods to protect the Edwards Aquifer. This **Temporary Stormwater Section** is hereby submitted for TCEQ review and executive director approval. The application was prepared by:

Print Name of Customer/Agent: J. Brandon Klenzendorf, P.E.

Date: 6 .14.2023

Signature of Customer/Agent:

y. Brandon Kloh

Regulated Entity Name: Capitol Aggregates Rio Medina Operation

Project Information

* J. BRANDON KLENZENDORF 113007 CENSED S/ONAL ENG

Potential Sources of Contamination

Examples: Fuel storage and use, chemical storage and use, use of asphaltic products, construction vehicles tracking onto public roads, and existing solid waste.

1. Fuels for construction equipment and hazardous substances which will be used during construction:

The following fuels and/or hazardous substances will be stored on the site: <u>Diesel fuel</u>, grease lubricant, hydraulic oil, gear oil, motor oil, transmission oil, gasoline, and used oil.

These fuels and/or hazardous substances will be stored in:

Aboveground storage tanks with a cumulative storage capacity of less than 250 gallons will be stored on the site for less than one (1) year.

TCEQ-0602 (Rev. 02-11-15)

Aboveground storage tanks with a cumulative storage capacity between 250 gallons and 499 gallons will be stored on the site for less than one (1) year.

- Aboveground storage tanks with a cumulative storage capacity of 500 gallons or more will be stored on the site. An Aboveground Storage Tank Facility Plan application must be submitted to the appropriate regional office of the TCEQ prior to moving the tanks onto the project.
- Fuels and hazardous substances will not be stored on the site.
- 2. Attachment A Spill Response Actions. A site specific description of the measures to be taken to contain any spill of hydrocarbons or hazardous substances is attached.
- 3. Temporary aboveground storage tank systems of 250 gallons or more cumulative storage capacity must be located a minimum horizontal distance of 150 feet from any domestic, industrial, irrigation, or public water supply well, or other sensitive feature.
- 4. Attachment B Potential Sources of Contamination. A description of any activities or processes which may be a potential source of contamination affecting surface water quality is attached.

Sequence of Construction

5. Attachment C - Sequence of Major Activities. A description of the sequence of major activities which will disturb soils for major portions of the site (grubbing, excavation, grading, utilities, and infrastructure installation) is attached.

For each activity described, an estimate (in acres) of the total area of the site to be disturbed by each activity is given.

For each activity described, include a description of appropriate temporary control measures and the general timing (or sequence) during the construction process that the measures will be implemented.

6. Name the receiving water(s) at or near the site which will be disturbed or which will receive discharges from disturbed areas of the project: <u>Tributary to Elm Creek</u>

Temporary Best Management Practices (TBMPs)

Erosion control examples: tree protection, interceptor swales, level spreaders, outlet stabilization, blankets or matting, mulch, and sod. Sediment control examples: stabilized construction exit, silt fence, filter dikes, rock berms, buffer strips, sediment traps, and sediment basins. Please refer to the Technical Guidance Manual for guidelines and specifications. All structural BMPs must be shown on the site plan.

7. Attachment D – Temporary Best Management Practices and Measures. TBMPs and measures will prevent pollution of surface water, groundwater, and stormwater. The construction-phase BMPs for erosion and sediment controls have been designed to retain sediment on site to the extent practicable. The following information is attached:

 A description of how BMPs and measures will prevent pollution of surface water, groundwater or stormwater that originates upgradient from the site and flows across the site. A description of how BMPs and measures will prevent pollution of surface water or groundwater that originates on-site or flows off site, including pollution caused by contaminated stormwater runoff from the site. A description of how BMPs and measures will prevent pollutants from entering surface streams, sensitive features, or the aquifer. A description of how, to the maximum extent practicable, BMPs and measures will maintain flow to naturally-occurring sensitive features identified in either the geologic assessment, TCEQ inspections, or during excavation, blasting, or construction.
The temporary sealing of a naturally-occurring sensitive feature which accepts recharge to the Edwards Aquifer as a temporary pollution abatement measure during active construction should be avoided.
 Attachment E - Request to Temporarily Seal a Feature. A request to temporarily seal a feature is attached. The request includes justification as to why no reasonable and practicable alternative exists for each feature. There will be no temporary sealing of naturally-occurring sensitive features on the site.
Attachment F - Structural Practices. A description of the structural practices that will be used to divert flows away from exposed soils, to store flows, or to otherwise limit runoff discharge of pollutants from exposed areas of the site is attached. Placement of structural practices in floodplains has been avoided.
Attachment G - Drainage Area Map. A drainage area map supporting the following requirements is attached:
 For areas that will have more than 10 acres within a common drainage area disturbed at one time, a sediment basin will be provided. For areas that will have more than 10 acres within a common drainage area disturbed at one time, a smaller sediment basin and/or sediment trap(s) will be used. For areas that will have more than 10 acres within a common drainage area disturbed at one time, a sediment basin or other equivalent controls are not attainable, but other TBMPs and measures will be used in combination to protect down slope and side slope boundaries of the construction area. There are no areas greater than 10 acres within a common drainage area that will be used in combination with other reosion and sediment controls within each disturbed at one time. A smaller sediment basin and/or sediment trap(s) will be used in combination with other erosion and sediment controls within each disturbed drainage area

There are no areas greater than 10 acres within a common drainage area that will be disturbed at one time. Erosion and sediment controls other than sediment basins or sediment traps within each disturbed drainage area will be used.

- 11. Attachment H Temporary Sediment Pond(s) Plans and Calculations. Temporary sediment pond or basin construction plans and design calculations for a proposed temporary BMP or measure have been prepared by or under the direct supervision of a Texas Licensed Professional Engineer. All construction plans and design information must be signed, sealed, and dated by the Texas Licensed Professional Engineer. Construction plans for the proposed temporary BMPs and measures are attached.
 - 🛛 N/A
- 12. Attachment I Inspection and Maintenance for BMPs. A plan for the inspection of each temporary BMP(s) and measure(s) and for their timely maintenance, repairs, and, if necessary, retrofit is attached. A description of the documentation procedures, recordkeeping practices, and inspection frequency are included in the plan and are specific to the site and/or BMP.
- 13. All control measures must be properly selected, installed, and maintained in accordance with the manufacturer's specifications and good engineering practices. If periodic inspections by the applicant or the executive director, or other information indicate a control has been used inappropriately, or incorrectly, the applicant must replace or modify the control for site situations.
- 14. If sediment escapes the construction site, off-site accumulations of sediment must be removed at a frequency sufficient to minimize offsite impacts to water quality (e.g., fugitive sediment in street being washed into surface streams or sensitive features by the next rain).
- 15. Sediment must be removed from sediment traps or sedimentation ponds not later than when design capacity has been reduced by 50%. A permanent stake will be provided that can indicate when the sediment occupies 50% of the basin volume.
- 16. 🖂 Litter, construction debris, and construction chemicals exposed to stormwater shall be prevented from becoming a pollutant source for stormwater discharges (e.g., screening outfalls, picked up daily).

Soil Stabilization Practices

Examples: establishment of temporary vegetation, establishment of permanent vegetation, mulching, geotextiles, sod stabilization, vegetative buffer strips, protection of trees, or preservation of mature vegetation.

17. Attachment J - Schedule of Interim and Permanent Soil Stabilization Practices. A schedule of the interim and permanent soil stabilization practices for the site is attached.

- 18. Records must be kept at the site of the dates when major grading activities occur, the dates when construction activities temporarily or permanently cease on a portion of the site, and the dates when stabilization measures are initiated.
- 19. Stabilization practices must be initiated as soon as practicable where construction activities have temporarily or permanently ceased.

Administrative Information

- 20. \square All structural controls will be inspected and maintained according to the submitted and approved operation and maintenance plan for the project.
- 21. If any geologic or manmade features, such as caves, faults, sinkholes, etc., are discovered, all regulated activities near the feature will be immediately suspended. The appropriate TCEQ Regional Office shall be immediately notified. Regulated activities must cease and not continue until the TCEQ has reviewed and approved the methods proposed to protect the aquifer from any adverse impacts.
- 22. Silt fences, diversion berms, and other temporary erosion and sediment controls will be constructed and maintained as appropriate to prevent pollutants from entering sensitive features discovered during construction.

Temporary Stormwater Section Form (TCEQ-0602) Supplemental Information

The information presented in the attachments to form TCEQ-0602 apply specifically to the Tank Storage Area associated with this AST Plan application. Temporary stormwater control measures identified in the original approved WPAP, which are located outside of the tank storage area, are still applicable and will not be modified.

Attachment A – Spill Response Actions

The objective of this section is to describe measures to prevent or reduce the discharge of pollutants to drainage systems or water courses from leaks and spills by reducing the chance for spills, stopping the source of spills, containing and cleaning up spills, properly disposing of spill materials, and training employees.

Employee Education

Measures must be implemented to prevent or reduce the discharge of pollutants to drainage systems or watercourses from leaks and spills by reducing the chance for spills, stopping the source of spills, containing and cleaning up spills, properly disposing of spill materials, and training employees. Employee training programs must include the following, at a minimum:

- awareness that different materials pollute in different amounts and the definition of a "significant spill" for each material used;
- awareness of the appropriate response for "significant" and "insignificant" spills and when a spill must be reported to the TCEQ;
- potential dangers to humans and the environment from spills and leaks;
- hold regular meetings to discuss and reinforce appropriate disposal procedures (incorporate into routine safety meetings);
- establish a continuing education program to indoctrinate new employees; and
- have representatives oversee and enforce proper spill prevention and control measures.

General Measures

The following general measures for spill response actions shall be implemented at the Site:

- To the extent that the work can be accomplished safely, contain and immediately clean up spills of oil, petroleum products, substances listed under 40 CFR parts 110, 117 and 302, and sanitary and septic wastes.
- Store hazardous materials and wastes in covered containers and protect from vandalism.

- Place a stockpile of spill cleanup materials where it will be readily accessible.
- Train employees in spill prevention and cleanup.
- Designate responsible individuals to oversee and enforce control measures.
- Cover spills and protect them from stormwater run-on during rainfall to the extent that it does not compromise cleanup activities.
- Do not bury or wash spills with water.
- Store and dispose of used clean up materials, contaminated materials, and recovered spill material that is no longer suitable for the intended purpose in conformance with the provisions in applicable BMPs.
- Do not allow water used for cleaning and decontamination to enter storm drains or watercourses. Collect and dispose of contaminated water in accordance with applicable regulations.
- Contain water overflow or minor water spillage and do not allow it to discharge into drainage facilities or watercourses.
- Place Safety Data Sheets (SDSs), as well as proper storage, cleanup, and spill reporting instructions for hazardous materials stored or used on the project site in an open, conspicuous, and accessible location.
- Keep waste storage areas clean, well-organized, and equipped with ample cleanup supplies as appropriate for the materials being stored. Perimeter controls, containment structures, covers, and liners should be repaired or replaced as needed to maintain proper function.

Cleanup

A spill containment kit and cleanup materials shall be stored where they are readily accessible. The Site's Spill Coordinator shall be designated to oversee and enforce control measures and maintain and restock the spill containment kit, as needed. Clean up leaks and spills immediately. Use a rag for small spills on paved surfaces, a damp mop for general cleanup, and absorbent materials for larger spills. If the spilled material is hazardous, then the used cleanup materials are also hazardous and must be disposed of as hazardous waste. Never hose down or bury dry material spills. Clean up as much of the material as possible and dispose of properly.

Minor Spills

Minor spills typically involve small quantities of oil, gasoline, paint, etc. which can be controlled by the first responder at the discovery of the spill. Use absorbent materials on small spills rather than washing the area with water or burying the spill. Absorbent materials should be promptly removed and disposed of properly. Follow the practice below for a minor spill:

- Contain the spread of the spill.
- Recover spill materials
- Clean the contaminated area and properly dispose of contaminated materials.

Semi-Significant Spills

Semi-significant spills that are below reportable quantities, are not hazardous, and can be controlled by the first responder along with the aid of other personnel such as laborers and the foreman, etc. This response may require the cessation of all other activities in order to contain the spill. Semi-significant spills should be cleaned up immediately using the following practices:

- Contain the spread of the spill.
- Notify the Spill Coordinator or project foreman immediately.
- If the spill occurs on paved or impermeable surfaces, clean up using dry methods (absorbent materials, cat litter and/or rags provided in the spill containment kit). Contain the spill by encircling it with absorbent materials to prevent the spread of the material.
- If the spill occurs in dirt areas, the spill can be contained by constructing an earthen berm; spilled material can be excavated and properly dispose of.
- If the spill occurs during rain, cover spill with tarps or other material to prevent contaminating runoff.

Significant or Hazardous Spills

For significant or hazardous spills, reportable quantities include the following limits:

- For spills of hazardous substances, the federal requirements in 40 CFR 302.4 apply.
- For petroleum products or used oil, the reportable quantity for a spill onto land is 25 gallons, and the reportable quantity directly into water is enough to create a sheen on the water.

In the event of a significant or hazardous spill in excess of reportable quantities, Site personnel shall:

• Notify the TCEQ San Antonio Regional Office by telephone as soon as possible, and within 24 hours, at (210) 490-3096 between 8 AM and 5 PM. After hours, contact the State of Texas Spill-Reporting Hotline and the State Emergency Response Commission (SERC) at 1-800-832-8224. All emergency phone numbers must be posted at the Site.

- For spills of federal reportable quantities, in conformance with the requirements in 40 CFR parts 110, 119, and 302, notify the National Response Center at 1-800-424-8802. Notifying the National Response Center does not constitute notice to TCEQ.
- Notification should first be made by telephone and followed up with a written report. Spills reported to the National Response Center can be reported via an online reporting tool.
- The services of a spills contractor or a Haz-Mat team should be obtained immediately. Construction personnel should not attempt to clean up a spill until the appropriate and qualified staff have arrived at the job site.
- Other agencies which may need to be notified include, but are not limited to, City Police Department, County Sheriff Office, Fire Departments, etc.

In the event of a reportable spill, the following Emergency Response Agencies can be contacted for assistance. Always inform your supervisor of a reportable spill immediately. Follow company policy when responding to an emergency.

(800) 832-8224
(512) 239-3700
(800) 424-8802
(830) 931-4751 911
(830) 931-2222 911
(830) 741-6150
(214) 665-2200 (800) 887-6063
(330) 722-9240
(830) 426-7444

Vehicle or Equipment Fueling and Maintenance

Fuel or hazardous substances will be stored on-site in the ASTs described in this application during project activities. Temporary ASTs with a capacity of 250 gallons or more will not be required during project activities. In the event that fuel trucks are used for on-site vehicle fueling, BMPs will be implemented during fueling and maintenance at the Site include the following:

- Inspect on-site vehicles and equipment daily for leaks and conduct repairs immediately.
- Check incoming vehicles and equipment (including delivery trucks, and employee and subcontractor vehicles) for leaking oil and fluids. Do not allow leaking vehicles or equipment on-site.
- Always use secondary containment, such as a drip pan or drop cloth, to catch spills or leaks when removing or changing fluids. Drip pans are to be used where leaks or spills of fuel can occur and where making and breaking hose connections.
- Promptly transfer used fluids to the proper waste or recycling drums; do not leave full drip pans or other open containers unattended for longer than necessary.
- Use absorbent materials on small spills rather than hosing down or burying the spill. Remove the absorbent materials promptly and dispose of properly.
- Fueling hoses should have check valves to prevent hose drainage after filling.
- Spills and vehicle leaks are to be cleaned up with absorbent materials for dry cleanup methods. The use of water for such cleanups is to be minimized.
- "Topping off" of fuel tanks is discouraged.
- Personnel are not to leave fueling equipment unattended during fueling operations.
- Oil filters disposed of in trash cans or dumpsters can leak oil and pollute stormwater. Place the oil filter in a funnel over a waste oil-recycling drum to drain excess oil before disposal. Oil filters can also be recycled. Ask the oil supplier or recycler about recycling oil filters.
- Store cracked batteries in a non-leaking secondary container. Do this with all cracked batteries even if you think all the acid has drained out. If you drop a battery, treat it as if it is cracked. Put it into the containment area until it is verified as not leaking.
- Equipment such as valves, pumps, flanges, and connections shall be checked regularly for leaks and be repaired as needed.

Portable Toilet Maintenance

Portable toilets will be used at the Site and will be handled and maintained as follows during construction. Portable toilets must be empty before transporting them. Toilets shall be securely fastened to transport trucks; hand trucks, dollies, or power tailgates shall be used whenever possible. Portable toilets shall be located at least 20 feet from the nearest storm drain inlet or sensitive feature buffer area (note that no sensitive features were identified near the tank storage

area in the Geologic Assessment). Toilets shall be located on a level ground surface, with clear access, and secured with a stake driven into the ground to prevent tipping by accident, weather, or vandalism.

During construction of the Tank Storage Area, portable toilets shall be inspected frequently for leaks (daily during the work week, or on a weekly basis and after rainfall events) and serviced/sanitized at time intervals that will maintain sanitary conditions (typically weekly or on an as-needed basis). A licensed waste collector should service all toilets. Suppliers should carry bleach for disinfection, in the event of a spill or leak, and properly store (or cover)/handle chemical materials. Employees shall be trained on these BMPs, prohibitions of stormwater discharge, and wastewater-discharge requirements.

Attachment B – Potential Sources of Contamination

Potential sources of contamination at the project site include soil, concrete material, fueling operations, material loading areas, trash and debris, spills or leaks, and other waste materials generated on-site. The regulated activities associated with this AST Plan will be located immediately adjacent to the Tank Storage Area.

Attachment C – Sequence of Major Activities

Due to scheduling and worker availability challenges, supply chain concerns, and material availability, Capitol Aggregates may initiate construction activities associated with the concrete pad and structure, roofing system, and electrical systems prior to approval from TCEQ on the AST application. A TCEQ Investigator recently conducted a Compliance Investigation at the Site and identified a portable 500-gallon fuel tank as an alleged violation. This AST Plan application includes this 500-gallon fuel tank and additional future tank installations. Capitol Aggregates will not initiate additional tank installation or transfer of tank contents until approval is received. The project consists of constructing Tank Storage Area, which includes installation of a concrete pad (approximately 38.5 feet by 35.2 ft), concrete dike (approximately 3 feet in height), roof system, and ASTs 1 through ASTs 12 northeast of the quarry pit, near the as part of the existing fueling and maintenance area. During construction, BMPs will be utilized in accordance with the approved WPAP, and no additional soil disturbance was added as part of this project. The major activities set to occur during the construction activities are described below. Activities for construction of Tank Storage Area are presented in the order or sequence in which they should be completed.

- 1) Clear vegetation in Tank Storage Area in preparation of installing concrete pad for the secondary containment structure; approximately 0.05 acres of area will be utilized.
- 2) Maintain any natural vegetated buffers that are existing down-gradient from cleared tank storage area.
- 3) Install concrete pad, concrete dike, and roof system.
- 4) Install gravel around concrete pad for permanent stabilization of adjacent areas.
- 5) Install AST 1 through AST 12 and associated piping systems.

Attachment D – Temporary Best Management Practices and Measures

Overview

The project consists of constructing the Tank Storage Area, which includes installation of a concrete pad (approximately 38.5 feet by 35.2 ft), concrete dike (approximately 3 feet in height), roof system, and ASTs 1 through ASTs 12 northeast of the quarry pit, as part of the existing fueling and maintenance area. During construction, BMPs will be utilized in accordance with the approved WPAP, and no additional soil disturbance will be added. Stormwater runoff from upgradient, undisturbed areas will be routed around the tank storage area. Only a minimal area, approximately 0.05 acres, will be utilized to install the concrete pad and ASTs associated with the Tank Storage Area. The disturbed area will be stabilized immediately with the concrete pad. Since the Tank Storage Area is located near the existing fueling and maintenance area, which is already designated as a disturbed area, only limited, temporary BMPs during construction were required, as described below. Additionally, the area will drain to the extended detention basin in series with the grassy swale approved in the WPAP.

Erosion Control BMPs

The purpose of soil stabilization is to provide a ground cover that limits the rainfall impact energy, provides a limited amount of water storage through rainfall interception, and limits sheet flow runoff velocity by increasing surface roughness. In the natural condition, soil is stabilized by native vegetation. Details of the soil stabilization BMPs that will be implemented are listed below.

• Concrete – The Tank Storage Area includes a concrete pad, which is an effective form of permanent stabilization.

Dust Control

The purpose of dust control is to prevent blowing and movement of dust from exposed soil surfaces, reduce on-site and off-site damage, reduce health hazards, and improve traffic safety.

Fugitive dust is emitted during construction (i.e., excavation, vehicle traffic, human activity) and as a result of wind erosion over the exposed earth surfaces. Proper management practices for dust control reduce or prevent wind erosion by protecting and roughening the soil surface and reducing the surface wind velocity.

Dust control is not expected to be an issue for the Tank Storage Area due to the small impacted area and limited duration of disturbed conditions. However, specific BMPs for dust control are listed below:

• Sprinkling – Sprinkling ground surfaces with water until it is moist is an effective dust control method. When water is used for dust control, the water should be applied such that runoff does not occur.

Sediment Control BMPs

The purpose of a surface water diversion structure is to limit the length of slope over which surface water runoff can travel as sheet flow or shallow concentrated flow. The diversion concentrates and laterally conveys surface water in a non-erosive manner. Since Tank Storage Area is located upgradient of the existing approved extended detention basin in series with the grassy swale, the main BMP for sediment control is the extended detention basin and grassy swale permanent BMPs associated with runoff from the fueling and maintenance area. No additional disturbed area was added as part of this project.

Non-Structural BMPs

Non-structural BMPs should be identified and integrated into any stormwater management program. As with any long-term program, effective implementation of these BMPs may require establishing specific criteria and standard procedures for various types of facilities or operations, and personnel training. In many cases, these procedures are simply "common sense" applied to routine activities. The primary objective of these measures is to prevent or reduce the amount of contaminants released to surface waters; however, the pollutant reduction that can be attributed to these measures has not been quantified. This discussion of non-structural BMPs emphasizes practices to achieve source control and pollution containment and prevention. These BMPs can also improve the operation and maintenance of structural stormwater management systems.

Non-structural BMPs such as preserving natural runoff conditions and maintaining natural vegetated areas will be implemented at the Site. Good housekeeping practices such as street sweeping will be utilized in the event that tracking of sediment onto public roadways is observed. Proper solid waste management will be used to control the accumulation of litter. Solid wastes and litter that accumulate on the land are easily transported by runoff. Appropriate placement of waste receptacles should be considered by the contractor. Regularly scheduled
maintenance of these receptacles and signage can encourage their use. Spill response and prevention, including employee training, are non-structural BMPs that will be used to reduce potential pollution.

BMPs to Prevent Pollution that Originates Upgradient from the Site

Stormwater runoff that originates upgradient from the Tank Storage Area is expected to be minimal, consisting mostly of runoff from the adjacent undeveloped areas. Since the Tank Storage Area is located upgradient of the existing approved extended detention basin in series with the grassy swale, the main BMP for sediment control is the extended detention basin and grassy swale permanent BMPs associated with runoff from the fueling and maintenance area.

BMPs to Prevent Pollution that Originates On-Site or Flows Off-Site

Since the Tank Storage Area is located upgradient of the existing approved extended detention basin in series with the grassy swale, the main BMP for sediment control is the extended detention basin and grassy swale permanent BMPs associated with runoff from the fueling and maintenance area.

BMPs to Prevent Pollution from Entering Surface Streams, Sensitive Features, or the Aquifer

Sensitive features are not expected to be a concern near the Tank Storage Area, as none were identified in the Geologic Assessment. The BMPs described above will be used to manage runoff that discharges to off-site surface streams.

BMPs to Maintain Flow to Sensitive Features

Sensitive features are not located within the Tank Storage Area, as identified in the Geologic Assessment. Therefore, BMPs are not required to maintain flow to sensitive features since they are not present. In the event that a sensitive feature is identified during TCEQ inspection or during construction activities, work activities will stop in the vicinity of the feature, TCEQ will be notified, temporary sediment and erosion controls will be implemented and remain in place until the feature closure methods have been completed, and a feature protection plan will be submitted to TCEQ for approval before commencing construction activities in the vicinity of the feature.

Attachment E – Request to Temporarily Seal a Feature

Not applicable. Sensitive geologic or manmade features are not located near the tank storage area; therefore, a Request to Temporarily Seal Features is not required.

Attachment F – Structural Practices

Structural practices used to divert flows away from exposed soils, to store flows, or to limit runoff discharge of pollutants from exposed areas of the Site. Since the Tank Storage Area is located upgradient of the existing approved extended detention basin in series with the grassy swale, the main BMP for sediment control is the extended detention basin and grassy swale permanent BMPs associated with runoff from the fueling and maintenance area. Placement of structural practices in the floodplain has been avoided.

<u>Attachment G – Drainage Area Map</u>

Construction activities associated with the Tank Storage Area installation will not disturb additional areas greater than 10 acres at one time. Therefore, a temporary sediment basin is not required for this project.

Attachment H – Temporary Sediment Pond(s) Plans and Calculations

Not applicable. The disturbed acreage is less than 10 acres; therefore, a temporary sediment pond is not required during construction.

Attachment I – Inspection and Maintenance for BMPs

Structural and non-structural BMPs should be inspected weekly and after each rainfall event of 0.5 inches or greater. Written documentation of these inspections should be kept during the course of construction at the project site. A report summarizing the scope of the inspection, the date of the inspection, and major observations relating to the implementation of BMPs must be documented. Major observations include: the location of discharges of sediment or other pollutants from the Site; locations of BMPs that need to be maintained; locations of BMPs that failed to operate as designed or proved inadequate for a particular location; and locations where additional BMPs are needed. Actions taken as a result of inspections must be described on the inspection report. Reports must identify any incidence of non-compliance. The names and qualifications of personnel making the inspections must be included on each report.

All BMPs must be maintained in effective operating condition. If sediment escapes the Site, offsite accumulations of sediment must be removed at a frequency sufficient to minimize off-site impacts to water quality and prior to the next rain event, if feasible. Litter, construction debris, and construction chemicals exposed to stormwater shall be prevented from becoming a pollutant

TXW9681/5.1_f-0602_Attachments.docx

source for stormwater discharges (e.g., screening outfalls, picked up routinely). Erosion from storms or other damage should be repaired as soon as practical by regrading the area and applying new seed. Any sediment deposited on the roadway shall be swept as necessary and within the same day as discovery.

During construction activities, stormwater will not require authorization to discharge under the TPDES General Permit No. TXR150000 for construction activities since the disturbed area is less than 1.0 acres (approximately 0.05 acres will be disturbed).

If periodic inspections by the contractor or other information indicate a control has been used inappropriately, or incorrectly, the contractor must replace or modify the control for Site situations.

Attachment J – Schedule of Interim and Permanent Soil Stabilization Practices

Due to scheduling and worker availability challenges, supply chain concerns, and material availability, Capitol Aggregates may initiate construction activities associated with the concrete pad and structure, roofing system, and electrical systems prior to approval from TCEQ on the AST application. A TCEQ Investigator recently conducted a Compliance Investigation at the Site and identified a portable 500-gallon fuel tank as an alleged violation. This AST Plan application includes this 500-gallon fuel tank and additional future tank installations. Capitol Aggregates will not initiate additional tank installation or transfer of tank contents until approval is received. A schedule of the interim and permanent soil stabilization practices for the Site must be maintained by the contractor. Records must be kept at the Site of the dates when major grading activities occur, the dates when construction activities temporarily or permanently cease on a portion of the Site, and the dates when stabilization measures are initiated. Stabilization practices must be initiated as soon as practicable where construction activities have temporarily or permanently ceased.

Interim Stabilization

Temporary stabilization measures must be initiated as soon as practicable in portions of the Site where construction activities have temporarily ceased, but no more than 14 days after the construction activities in that portion of the Site have ceased. In areas where temporary stabilization measures are infeasible, the contractor may alternatively utilize temporary perimeter controls. The contractor must document the reason why stabilization measures are not feasible and must demonstrate that the perimeter controls will retain sediment on-site to the extent practicable.

Permanent Stabilization

Permanent stabilization consists of the concrete pad. Areas shall not remain disturbed for more than 14 days after the permanent completion of construction activities. Natural vegetation shall be preserved to the extent practicable.

Final stabilization is considered to have occurred when all soil disturbing activities have been completed and a uniform (i.e., evenly distributed, without large bare areas) perennial vegetative cover, with a density of at least 70 percent of the native background vegetative cover for the area, has been established, or equivalent permanent stabilization measures (such as the use of impervious cover) have been employed.

The owner is responsible for the maintenance of these stabilization measures until the applicant informs the TCEQ in writing that another organization or individual has assumed control or ownership of the property.

Permanent Stormwater Section

Texas Commission on Environmental Quality

for Regulated Activities on the Edwards Aquifer Recharge Zone and Relating to 30 TAC §213.5(b)(4)(C), (D)(li), (E), and (5), Effective June 1, 1999

To ensure that the application is administratively complete, confirm that all fields in the form are complete, verify that all requested information is provided, consistently reference the same site and contact person in all forms in the application, and ensure forms are signed by the appropriate party.

Note: Including all the information requested in the form and attachments contributes to more streamlined technical reviews.

Signature

To the best of my knowledge, the responses to this form accurately reflect all information requested concerning the proposed regulated activities and methods to protect the Edwards Aquifer. This **Permanent Stormwater Section** is hereby submitted for TCEQ review and executive director approval. The application was prepared by:

Print Name of Customer/Agent: J. Brandon Klenzendorf, P.E.

Date: 6 · 14 · 2023

Signature of Customer/Agent

J. Brandon Kyll

Regulated Entity Name: Capitol Aggregates Rio Medina Operation

Permanent Best Management Practices (BMPs)

Permanent best management practices and measures that will be used during and after construction is completed.

1. Permanent BMPs and measures must be implemented to control the discharge of pollution from regulated activities after the completion of construction.

🛛 N/A

2. These practices and measures have been designed, and will be constructed, operated, and maintained to insure that 80% of the incremental increase in the annual mass loading of total suspended solids (TSS) from the site caused by the regulated activity is removed. These quantities have been calculated in accordance with technical guidance prepared or accepted by the executive director.

The TCEQ Technical Guidance Manual (TGM) was used to design permanent BMPs and measures for this site.

J. BRANDON KLENZENDOR

A technical guidance other than the TCEQ TGM was used to design permanent BMPs and measures for this site. The complete citation for the technical guidance that was used is: _____

🖂 N/A

3. Owners must insure that permanent BMPs and measures are constructed and function as designed. A Texas Licensed Professional Engineer must certify in writing that the permanent BMPs or measures were constructed as designed. The certification letter must be submitted to the appropriate regional office within 30 days of site completion.

N/A

- 4. Where a site is used for low density single-family residential development and has 20 % or less impervious cover, other permanent BMPs are not required. This exemption from permanent BMPs must be recorded in the county deed records, with a notice that if the percent impervious cover increases above 20% or land use changes, the exemption for the whole site as described in the property boundaries required by 30 TAC §213.4(g) (relating to Application Processing and Approval), may no longer apply and the property owner must notify the appropriate regional office of these changes.
 - The site will be used for low density single-family residential development and has 20% or less impervious cover.
 - The site will be used for low density single-family residential development but has more than 20% impervious cover.
 - The site will not be used for low density single-family residential development.
- 5. The executive director may waive the requirement for other permanent BMPs for multifamily residential developments, schools, or small business sites where 20% or less impervious cover is used at the site. This exemption from permanent BMPs must be recorded in the county deed records, with a notice that if the percent impervious cover increases above 20% or land use changes, the exemption for the whole site as described in the property boundaries required by 30 TAC §213.4(g) (relating to Application Processing and Approval), may no longer apply and the property owner must notify the appropriate regional office of these changes.
 - Attachment A 20% or Less Impervious Cover Waiver. The site will be used for multi-family residential developments, schools, or small business sites and has 20% or less impervious cover. A request to waive the requirements for other permanent BMPs and measures is attached.
 - The site will be used for multi-family residential developments, schools, or small business sites but has more than 20% impervious cover.
 - The site will not be used for multi-family residential developments, schools, or small business sites.
- 6. Attachment B BMPs for Upgradient Stormwater.

	 A description of the BMPs and measures that will be used to prevent pollution of surface water, groundwater, or stormwater that originates upgradient from the site and flows across the site is attached. No surface water, groundwater or stormwater originates upgradient from the site and flows across the site, and an explanation is attached. Permanent BMPs or measures are not required to prevent pollution of surface water, groundwater, or stormwater that originates upgradient from the site and flows across the site, and an explanation is attached.
7.	X Attachment C - BMPs for On-site Stormwater.
	 A description of the BMPs and measures that will be used to prevent pollution of surface water or groundwater that originates on-site or flows off the site, including pollution caused by contaminated stormwater runoff from the site is attached. Permanent BMPs or measures are not required to prevent pollution of surface water or groundwater that originates on-site or flows off the site, including pollution caused by contaminated stormwater runoff.
8.	Attachment D - BMPs for Surface Streams. A description of the BMPs and measures that prevent pollutants from entering surface streams, sensitive features, or the aquifer is attached. Each feature identified in the Geologic Assessment as sensitive has been addressed.
	N/A
9.	The applicant understands that to the extent practicable, BMPs and measures must maintain flow to naturally occurring sensitive features identified in either the geologic assessment, executive director review, or during excavation, blasting, or construction.
	 The permanent sealing of or diversion of flow from a naturally-occurring sensitive feature that accepts recharge to the Edwards Aquifer as a permanent pollution abatement measure has not been proposed. Attachment E - Request to Seal Features. A request to seal a naturally-occurring sensitive feature, that includes, for each feature, a justification as to why no reasonable and practicable alternative exists, is attached.
10.	Attachment F - Construction Plans. All construction plans and design calculations for the proposed permanent BMP(s) and measures have been prepared by or under the direct supervision of a Texas Licensed Professional Engineer, and are signed, sealed, and dated. The plans are attached and, if applicable include:
	 Design calculations (TSS removal calculations) TCEQ construction notes All geologic features All proposed structural BMP(s) plans and specifications
	□ N/A

11.	Attachment G - Inspection, Maintenance, Repair and Retrofit Plan. A plan for the inspection, maintenance, repairs, and, if necessary, retrofit of the permanent BMPs and measures is attached. The plan includes all of the following:
	 Prepared and certified by the engineer designing the permanent BMPs and measures Signed by the owner or responsible party Procedures for documenting inspections, maintenance, repairs, and, if necessary
	retrofit A discussion of record keeping procedures
\square	N/A
12.	Attachment H - Pilot-Scale Field Testing Plan . Pilot studies for BMPs that are not recognized by the Executive Director require prior approval from the TCEQ. A plan for pilot-scale field testing is attached.
\square	N/A
13.	Attachment I -Measures for Minimizing Surface Stream Contamination. A description of the measures that will be used to avoid or minimize surface stream contamination and changes in the way in which water enters a stream as a result of the construction and development is attached. The measures address increased stream flashing, the creation of stronger flows and in-stream velocities, and other in-stream effects caused by the regulated activity, which increase erosion that results in water quality degradation.

 \square N/A

Responsibility for Maintenance of Permanent BMP(s)

Responsibility for maintenance of best management practices and measures after construction is complete.

14. The applicant is responsible for maintaining the permanent BMPs after construction until such time as the maintenance obligation is either assumed in writing by another entity having ownership or control of the property (such as without limitation, an owner's association, a new property owner or lessee, a district, or municipality) or the ownership of the property is transferred to the entity. Such entity shall then be responsible for maintenance until another entity assumes such obligations in writing or ownership is transferred.

N/A

15. A copy of the transfer of responsibility must be filed with the executive director at the appropriate regional office within 30 days of the transfer if the site is for use as a multiple single-family residential development, a multi-family residential development, or a non-residential development such as commercial, industrial, institutional, schools, and other sites where regulated activities occur.

🖂 N/A

Permanent Stormwater Section Form (TCEQ-0600) Supplemental Information

The information presented in the attachments to form TCEQ-0600 applies specifically to the aboveground storage tank facility associated with this AST Plan (Tank Storage Area) at the Capitol Aggregates Rio Medina Operation. Permanent pollution abatement measures identified in the original approved WPAP consist of vegetated buffers, earthen berms, rock berms, an extended detention basin, and a grassy swale. The approved WPAP outlines three ways in which stormwater is managed on site: 1) captured in the quarry pit and allowed to evaporate; 2) routed via earthen berms and discharged through rock berms to the site perimeter or through natural vegetated buffers to the tributary to Elm Creek; or 3) routed to an on-site permanent extended detention basin and grassy swale for the initial support activities area, while those activities are located outside the quarry pit, and discharged off-site.

Capitol Aggregates is submitting an AST Plan to address the new tanks at the Tank Storage Area (AST 1 through AST 12). The project consists of constructing Tank Storage Area, which includes the installation of a concrete pad (approximately 38.5 feet by 35.2 ft), concrete dike (approximately 3 feet in height), roof system, and ASTs 1 through 12 near the existing fueling and maintenance area on-site. The main BMP during construction consists of those included in the approved WPAP, and no additional soil disturbance was added as part of this project. Due to scheduling and worker availability challenges, supply chain concerns, and material availability, Capitol Aggregates may initiate construction activities associated with the concrete pad and structure, roofing system, and electrical systems prior to approval from TCEQ on the AST application. A TCEQ Investigator recently conducted a Compliance Investigation at the Site and identified a portable 500-gallon fuel tank as an alleged violation. This AST Plan application includes this 500-gallon fuel tank and additional future tank installations. Capitol Aggregates will not initiate additional tank installation or transfer of tank contents until approval is received.

Attachment A – 20% or Less Impervious Cover Waiver

Not applicable. The site associated with this AST Plan application will not be used for multifamily residential developments, schools, or small business sites.

Attachment B – BMPs for Upgradient Stormwater

No surface water, groundwater, or stormwater originates upgradient from the site and flows across the site. The Tank Storage Area Project Site will be located on a concrete slab with a concrete dike that will be elevated so that any potential upgradient stormwater runoff will flow

around the Tank Storage Area. Stormwater runoff that originates upgradient from the Tank Storage Area is expected to be minimal. Additionally, the containment system at the Tank Storage Area will include a covered roofing system to prevent precipitation from entering the containment structure. Due to the limited upgradient area draining to the Tank Storage Area and coverage from the roof, permanent BMPs are not necessary for upgradient stormwater.

Attachment C – BMPs for On-Site Stormwater

Additional permanent BMPs are not required to treat runoff from the Tank Storage Area because the impervious cover associated with the area has been addressed by the approved WPAP. The Tank Storage Area is in the fueling and maintenance area and, as described in the approved WPAP, stormwater runoff that originates in this area is routed to the approved existing extended detention basin and grassy swale (existing permanent BMPs). Therefore, additional permanent BMPs from those described in the approved WPAP for on-site stormwater will not be necessary due to the construction of the Tank Storage Area.

Attachment D – BMPs for Surface Streams

Not applicable. Permanent BMPs are not required to prevent pollutants from entering surface streams, sensitive features, or the aquifer because the Tank Storage Area Project Site is covered under the approved WPAP. Sensitive features are not located within 800 feet of the Tank Storage Area.

Attachment E – Request to Seal a Features

Not applicable. Sensitive geologic or manmade features are not located near the Tank Storage Area Project Site; therefore, a Request to Seal Features is not required.

Attachment F – Construction Plans

The Construction Plans for the project are submitted as the Scaled Drawings and Site Plan with the Aboveground Storage Tank (AST) Plan application (form TCEQ-0575). All geologic features and structural BMPs in the vicinity of the Tank Storage Area Project Site and TCEQ construction notes are shown on the Site Plan. Design calculations (TSS removal calculations) are not required for this AST Plan application because the approved WPAP allows for the extended detention basin and grassy swale to address runoff from the Tank Storage Area.

Attachment G – Inspection, Maintenance, Repair and Retrofit Plan

Not applicable. No additional permanent BMPs are proposed. Updates to the Inspection, Maintenance, Repair and Retrofit Plan in the approved WPAP are not required.

<u>Attachment H – Pilot-Scale Field Testing Plan</u>

Not applicable. BMPs that are not recognized by the TCEQ Executive Director are not proposed for the project. Therefore, a Pilot-Scale Field Testing Plan is not required.

Attachment I – Measures for Minimizing Surface Stream Contamination

Not applicable. The Tank Storage Area Project Site will be located at the existing fueling and maintenance area. The impervious cover associated with the Tank Storage Area has been addressed by the approved WPAP. Therefore, additional measures as part of this AST Plan application for minimizing surface stream contamination are not required. The little additional runoff that will be generated from the Tank Storage Area will be attenuated and buffered by the extended detention basin and grassy swale, so stream flashing, stronger flows, and increases in in-stream velocities are not expected to occur as a result of this project.

Agent Authorization Form For Required Signature Edwards Aquifer Protection Program Relating to 30 TAC Chapter 213 Effective June 1, 1999

I William Scotty Gerbes,	
Print Name	
Vice President - Aggregates, Title - Owner/President/Other	
of <u>Capitol Aggregates, Inc.</u> Corporation/Partnership/Entity Name	,
have authorized <u>J. Brandon Klenzendorf</u> Print Name of Agent/Engineer	
of <u>Geosyntec Consultants, Inc.</u> Print Name of Firm	_

to represent and act on the behalf of the above named Corporation, Partnership, or Entity for the purpose of preparing and submitting this plan application to the Texas Commission on Environmental Quality (TCEQ) for the review and approval consideration of regulated activities.

I also understand that:

- 1. The applicant is responsible for compliance with 30 Texas Administrative Code Chapter 213 and any condition of the TCEQ's approval letter. The TCEQ is authorized to assess administrative penalties of up to \$10,000 per day per violation.
- 2. For those submitting an application who are not the property owner, but who have the right to control and possess the property, additional authorization is required from the owner.
- 3. Application fees are due and payable at the time the application is submitted. The application fee must be sent to the TCEQ cashier or to the appropriate regional office. The application will not be considered until the correct fee is received by the commission.
- 4. A notarized copy of the Agent Authorization Form must be provided for the person preparing the application, and this form must accompany the completed application.
- 5. No person shall commence any regulated activity on the Edwards Aquifer Recharge Zone, Contributing Zone or Transition Zone until the appropriate application for the activity has been filed with and approved by the Executive Director.

Ð Applicar

<u>5-9-23</u> Date

THE STATE OF Texas § County of Bexar §

BEFORE ME, the undersigned authority, on this day personally appeared $\underline{W. Scotty}$ Gerbesknown to me to be the person whose name is subscribed to the foregoing instrument, and acknowledged to me that (s)he executed same for the purpose and consideration therein expressed.

GIVEN under my hand and seal of office on this $\frac{940}{202}$ day of $\frac{May}{202}$



Notary Typed or Printed Name of

MY COMMISSION EXPIRES: 7-26-2023

Application Fee Form

Name of Proposed Regulated Entity: Capitol Aggregates		Texas Commission on Environmental Quality							
	Name of Proposed Regulated Entity: Capitol Aggregates Rio Medina Operation								
Regulated Entity Location: 1576 County Road 265, Hondo, Texas 78861									
Name of Customer: <u>Capitol Aggregates Inc.</u>									
Contact Person: Andrew Frye Phor	ne: <u>210-871-7214</u>								
Customer Reference Number (if issued):CN <u>604033142</u>									
Regulated Entity Reference Number (if issued):RN <u>108959032</u>									
Austin Regional Office (3373)									
Havs Travis	Wi	illiamson							
San Antonio Regional Office (3362)									
		alda							
Bexar Xiedina		alue							
Comal Kinney									
Application fees must be paid by check, certified check, o	or money order, payab	le to the Texas							
Commission on Environmental Quality. Your canceled of	check will serve as you	r receipt. This							
form must be submitted with your fee payment. This p	ayment is being submi	tted to:							
Austin Regional Office	an Antonio Regional O	office							
🔀 Mailed to: TCEQ - Cashier 🛛 🗌 C	Overnight Delivery to: 1	CEQ - Cashier							
Revenues Section 1	2100 Park 35 Circle	34							
Mail Code 214 E	Building A, 3rd Floor								
P.O. Box 13088	Austin, TX 78753								
Austin, TX 78711-3088 (512)239-0357	Austin, TX 78711-3088 (512)239-0357							
Site Location (Check All That Apply):									
Site Location (Check All That Apply):									
Site Location (Check All That Apply):	Transi	tion Zone							
Site Location (Check All That Apply): Recharge Zone Contributing Zone	Transi	tion Zone Fee Due							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone	Transi	tion Zone Fee Due							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling	Transi Size 0 Acres	tion Zone Fee Due \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone	Transi Size 0 Acres	tion Zone Fee Due \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks	Transi Size 0 Acres 0 Acres	tion Zone Fee Due \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone	Transi Size 0 Acres 0 Acres	tion Zone Fee Due \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone Plan: Non-residential	Transi Size 0 Acres 0 Acres 0 Acres	tion Zone Fee Due \$ 0 \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone Plan: Non-residential Sewage Collection System	Transi Size 0 Acres 0 Acres 0 Acres 0 L.F.	tion Zone Fee Due \$ 0 \$ 0 \$ 0 \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone Plan: Non-residential Sewage Collection System Lift Stations without sewer lines	Transi Size 0 Acres 0 Acres 0 Acres 0 Acres 0 L.F. 0 Acres	tion Zone Fee Due \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone Plan: Non-residential Sewage Collection System Lift Stations without sewer lines Underground or Aboveground Storage Tank Facility	Transi Size O Acres O Acres O Acres O Acres O L.F. O Acres 10 regulated Tanks	tion Zone Fee Due \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone Plan: Non-residential Sewage Collection System Lift Stations without sewer lines Underground or Aboveground Storage Tank Facility Piping System(s)(only)	Transi Size O Acres O Acres O Acres O Acres O L.F. O Acres 10 regulated Tanks O Each	tion Zone Fee Due \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone Plan: Non-residential Sewage Collection System Lift Stations without sewer lines Underground or Aboveground Storage Tank Facility Piping System(s)(only) Exception	Transi Size O Acres O Acres O Acres O Acres O L.F. O Acres 10 regulated Tanks O Each O Each	tion Zone Fee Due \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0							
Site Location (Check All That Apply): Recharge Zone Contributing Zone Type of Plan Water Pollution Abatement Plan, Contributing Zone Plan: One Single Family Residential Dwelling Water Pollution Abatement Plan, Contributing Zone Plan: Multiple Single Family Residential and Parks Water Pollution Abatement Plan, Contributing Zone Plan: Non-residential Sewage Collection System Lift Stations without sewer lines Underground or Aboveground Storage Tank Facility Piping System(s)(only) Exception Extension of Time	Transi Size O Acres O Acres O Acres O Acres O L.F. O Acres 10 regulated Tanks O Each O Each O Each	tion Zone Fee Due \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0 \$ 0							

TCEQ-0574 (Rev. 02-24-15)

1 of 2

Application Fee Schedule

Texas Commission on Environmental Quality

Edwards Aquifer Protection Program 30 TAC Chapter 213 (effective 05/01/2008)

Water Pollution Abatement Plans and Modifications

Contributing Zone Plans and Modifications

	Project Area in	
Project	Acres	Fee
One Single Family Residential Dwelling	< 5	\$650
Multiple Single Family Residential and Parks	< 5	\$1,500
	5 < 10	\$3,000
	10 < 40	\$4,000
	40 < 100	\$6,500
	100 < 500	\$8,000
	≥ 500	\$10,000
Non-residential (Commercial, industrial, institutional,	< 1	\$3,000
multi-family residential, schools, and other sites	1 < 5	\$4,000
where regulated activities will occur)	5 < 10	\$5,000
	10 < 40	\$6,500
	40 < 100	\$8,000
	≥ 100	\$10,000

Organized Sewage Collection Systems and Modifications

Project	Cost per Linear Foot	Minimum Fee- Maximum Fee
Sewage Collection Systems	\$0.50	\$650 - \$6,500

Underground and Aboveground Storage Tank System Facility Plans and Modifications

Project	Cost per Tank or Piping System	Minimum Fee- Maximum Fee	
,	1 3 7		
Underground and Aboveground Storage Tank Facility	\$650	\$650 - \$6,500	

Exception Requests

Project	Fee
Exception Request	\$500

Extension of Time Requests

Project	Fee
Extension of Time Request	\$150



TCEQ Core Data Form

For detailed instructions on completing this form, please read the Core Data Form Instructions or call 512-239-5175.

SECTION I: General Information

1. Reason for Submission (If other is checked please describe in space provided.)								
New Permit, Registration or Authorization (Core Data Form should be submitted with the program application.)								
Renewal (Core Data Form should be submitted with th	Renewal (Core Data Form should be submitted with the renewal form) Other Modification to existing AST Facility Plan							
2. Customer Reference Number (if issued)	Follow this link to search	3. Regulated Entity Reference Number (if issued)						
CN 604033142	<u>Central Registry**</u>	RN 108959032						

SECTION II: Customer Information

4. General Cu	4. General Customer Information 5. Effective Date for Customer Information Updates (mm/dd/yyyy)												
New Customer Update to Customer Information Change in Regulated Entity Ownership													
Change in Le	Change in Legal Name (Verifiable with the Texas Secretary of State or Texas Comptroller of Public Accounts)												
The Custome	r Name sı	ıbmitted her	e may b	oe updated	automatical	ly base	ed oi	n what is c	urrent	and active	with th	he Texas Sec	retary of State
(SOS) or Texa	(SOS) or Texas Comptroller of Public Accounts (CPA).												
6. Customer	Legal Nam	ne (If an individ	dual, prii	nt last name j	first: eg: Doe, J	lohn)			<u>If new</u>	v Customer,	enter pro	evious Custom	<u>er below:</u>
Capitol Aggrega	ates, Inc.												
7. TX SOS/CP	A Filing N	umber		8. TX State	e Tax ID (11 d	ligits)			9. Fe	deral Tax I	D	10. DUNS I	Number (if
001535417				174272422					(0 dia	:+-)		applicable)	
801525417				1/42/3122	22				(9 algits)		008119422		
									74273	42731225			
11. Type of C	ustomer:	\boxtimes	Corporat	ion				🗌 Individ	dual		Partne	ership: 🗌 Gen	eral 🗌 Limited
Government: [City 🗌 🤇	County 🗌 Fed	eral 🗌	Local 🗌 Sta	te 🗌 Other			Sole P	roprieto	orship	Ot	her:	
12. Number o	of Employ	ees							13. lr	ndepender	ntly Ow	ned and Ope	erated?
0-20	21-100	101-250	251-	500 🗌 50	1 and higher				🛛 Ye	25	🗌 No		
14. Customer	r Role (Pro	posed or Actua	al) – <i>as i</i> i	t relates to th	ne Regulated E	ntity lis	ted o	on this form.	Please	check one oj	f the follo	owing	
Owner		Operator		X	wner & Opera	ator							
Occupation	al Licensee	Respon	sible Par	rty 🗌] VCP/BSA App	olicant				U Other:			
15. Mailing	P.O. Box	33240											
2011101119													
Address:					T				1			1	
	City	San Antonio			State	ТΧ		ZIP	78265	5		ZIP + 4	3240
16. Country M	Vailing In	formation (if	outside	USA)			17. E-Mail Address (if applicable)						
					andrew.frye@capitolaggregates.com								
18. Telephone Number 19. E					19. Extensio	on or C	ode			20. Fax N	umber	(if applicable)	

SECTION III: Regulated Entity Information

21 Concred Regulated F	atitus Indonuso	tion //f (Now Downlat				ation is also assuring a l		
21. General Regulated Entity Information (if New Regulated Entity is selected, a new permit application is also required.)								
🗌 New Regulated Entity 🔄 Update to Regulated Entity Name 🔄 Update to Regulated Entity Information								
The Regulated Entity Name submitted may be updated, in order to meet TCEQ Core Data Standards (removal of organizational endings such as Inc, LP, or LLC).								
22. Regulated Entity Nan	ne (Enter nam	e of the site where the	regulated action	n is taking pl	ace.)			
Capitol Aggregates Rio Medi	ina Operation							
23. Street Address of	1576 CR 265							
the Regulated Entity:								
<u>(No PO Boxes)</u>	City	Hondo	State	ТХ	ZIP	78861	ZIP + 4	
24. County	Medina				•			
		If no Street Ad	ldress is provid	led, fields 2	25-28 are re	equired.		
25. Description to	Project locat	ed on the west side of	County Road 26	5 annroxim	ately 2 5 mile	s north of the intersect	tion of Cour	nty Boad 265 and
Physical Location:	Physical Location: County Road 354.							
26. Nearest City	1					State	N	earest ZIP Code
Rio Medina						ТХ	78	066

Latitude/Longitude are required and may be added/updated to meet TCEQ Core Data Standards. (Geocoding of the Physical Address may be used to supply coordinates where none have been provided or to gain accuracy).

								1	
27. Latitude (N) In Decimal: 29.4		29.48665	55		28. Longitude (W) In Decim		nal:	-98.98795	
Degrees	Minutes		Seconds	Degre	es	Mi			Seconds
29		29 11.95			-98		59		16.61
29. Primary SIC Code 30. Secondary SIC			Code	ode 31. Primary NAICS Code 32. Secondary NAICS				CS Code	
(4 digits)	(4 digits) (5 or 6 digits) (5 or 6 digits)				its)				
1422	2951			212312 324121			324121		
33. What is the Primary E	Business of	f this entity? (Do	o not repeat the SIC c	or NAICS descr	iption.)		1		
Quarry									
P.O. Box 33240									
34. Mailing	. Mailing								
Address:									
	City	San Antonio	State	тх	ZIP	78265		ZIP + 4	3240
35. E-Mail Address: andrew.frye@capitolaggregates.com									
36. Telephone Number 37. Extension or Code 38. Fax Number (if applicable)									
(210) 871-7214		(210) 599-1709							

39. TCEQ Programs and ID Numbers Check all Programs and write in the permits/registration numbers that will be affected by the updates submitted on this form. See the Core Data Form instructions for additional guidance.

Dam Safety	Districts	Edwards Aquifer	Emissions Inventory Air	Industrial Hazardous Waste	
		13000073			
Municipal Solid Waste	New Source Review Air	OSSF	Petroleum Storage Tank	D PWS	
	138162				
Sludge	Storm Water	🔲 Title V Air	Tires	Used Oil	
	TXR05EG16				
Voluntary Cleanup	U Wastewater	Wastewater Agriculture	Water Rights	Other:	

SECTION IV: Preparer Information

40. Name:	Kelsey Krueger, P.G.			41. Title:	Project Geologist	
42. Telephone Number		43. Ext./Code	44. Fax Number	45. E-Mail Address		
(817) 648-9470			() -	kelsey.krueg	er@geosyntec.com	

SECTION V: Authorized Signature

46. By my signature below, I certify, to the best of my knowledge, that the information provided in this form is true and complete, and that I have signature authority to submit this form on behalf of the entity specified in Section II, Field 6 and/or as required for the updates to the ID numbers identified in field 39.

Company:	Geosyntec Consultants, Inc.	Job Title:	Principal E	Principal Engineer		
Name (In Print):	J. Brandon Klenzendorf, P.E.				(512) 354- 3281	
Signature:	Brandon Kleyenda			Date:	6/7/2023	